UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

(Mark One)

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2016

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ______ to _____

Commission File Number 001-36569

LANTHEUS HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State of incorporation)

331 Treble Cove Road, North Billerica, MA (Address of principal executive offices)

> (978) 671-8001 (Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \boxtimes No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer		Accelerated filer	
Non-accelerated filer	☑ (Do not check if a smaller reporting company)	Smaller reporting company	
Indicate by check	mark whether the registrant is a shell company (as defined by Rule 12b-2 of the Act) Ye	s 🗆 No 🖾	
The registrant had	32,724,407 of common stock, \$0.01 par value per share, issued and outstanding as of Ma	y 3, 2016.	

35-2318913 (IRS Employer Identification No.)

> 01862 (Zip Code)

TABLE OF CONTENTS

PART I. FINANCIAL INFORMATION

P	a	g	e

Item 1.	Financial Statements (Unaudited)	1
	Condensed Consolidated Statements of Operations for the Three Months Ended March 31, 2016 and 2015	1
	Condensed Consolidated Statements of Comprehensive Income for the Three Months Ended March 31, 2016 and 2015	2
	Condensed Consolidated Balance Sheets as of March 31, 2016 and December 31, 2015	3
	Condensed Consolidated Statements of Cash Flows for the Three Months Ended March 31, 2016 and 2015	4
	Notes to Unaudited Condensed Consolidated Financial Statements	5
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	18
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	30
Item 4.	Controls and Procedures	31
	PART II. OTHER INFORMATION	
Item 1.	Legal Proceedings	32
Item 1A.	Risk Factors	32
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	33
Item 4	Mine Safety Disclosures	33
Item 6.	Exhibits	34
Signatures		35
Exhibit Ind	<u>ex</u>	36

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements (Unaudited)

Lantheus Holdings, Inc. and subsidiaries

Condensed Consolidated Statements of Operations

(unaudited, in thousands, except share data)

		For the Three Months Ended March 31,	
	2016	2015	
Revenues	\$ 76,474	\$ 74,823	
Cost of goods sold	42,773	39,054	
Gross profit	33,701	35,769	
Operating expenses			
Sales and marketing expenses	9,307	9,072	
General and administrative expenses	9,513	9,123	
Research and development expenses	3,036	6,196	
Total operating expenses	21,856	24,391	
Gain on sale of assets	5,828		
Operating income	17,673	11,378	
Interest expense, net	(7,018)	(10,623)	
Other income (expense), net	58	(383)	
Income before income taxes	10,713	372	
Provision (benefit) for income taxes	390	(3)	
Net income	\$ 10,323	\$ 375	
Net income per common share:			
Basic and diluted	\$ 0.34	\$ 0.02	
Common shares:			
Basic	30,368,240	18,080,944	
Diluted	30,372,691	18,404,393	

See notes to unaudited condensed consolidated financial statements.

Condensed Consolidated Statements of Comprehensive Income

(unaudited, in thousands)

		For the Three Months Ended March 31,	
	2016	2015	
Net income	\$ 10,323	\$ 375	
Foreign currency translation	340	(358)	
Total comprehensive income	<u>\$ 10,663</u>	\$ 17	

See notes to unaudited condensed consolidated financial statements.

Condensed Consolidated Balance Sheets

(unaudited, in thousands, except share data)

	March 31, 2016	December 31, 2015
Assets		
Current assets		
Cash and cash equivalents	\$ 38,880	\$ 28,596
Accounts receivable, net of allowance of \$1,051 and \$881	39,896	37,293
Inventory	15,541	15,622
Other current assets	5,721	3,851
Assets held for sale		4,644
Total current assets	100,038	90,006
Property, plant and equipment, net	85,324	86,517
Capitalized software development costs, net	8,615	9,137
Intangibles, net	19,235	20,496
Goodwill	15,714	15,714
Other long-term assets	20,337	20,509
Total assets	\$ 249,263	\$ 242,379
Liabilities and Stockholders' Deficit		
Current liabilities		
Current portion of long-term debt	\$ 3,650	\$ 3,650
Line of credit	—	
Accounts payable	12,984	11,657
Accrued expenses and other liabilities	14,845	18,502
Liabilities held for sale		1,715
Total current liabilities	31,479	35,524
Asset retirement obligation	8,417	8,145
Long-term debt, net	349,349	349,858
Other long-term liabilities	34,237	34,141
Total liabilities	423,482	427,668
Commitments and contingencies (See Note 16)		
Stockholders' deficit		
Preferred stock (\$0.01 par value, 25,000,000 shares authorized; no shares issued and outstanding)	_	_
Common stock (\$0.01 par value, 250,000,000 shares authorized; 30,377,104 and 30,364,501 shares issued and		
outstanding)	303	303
Additional paid-in capital	175,960	175,553
Accumulated deficit	(348,837)	(359,160)
Accumulated other comprehensive loss	(1,645)	(1,985)
Total stockholders' deficit	(174,219)	(185,289)
Total liabilities and stockholders' deficit	\$ 249,263	\$ 242,379

See notes to unaudited condensed consolidated financial statements.

Condensed Consolidated Statements of Cash Flows

(unaudited, in thousands)

		For the Three Months Ended March 31,	
	2016	2015	
Cash flows from operating activities		<u> </u>	
Net income	\$ 10,323	\$ 375	
Adjustments to reconcile net income to cash flow from operating activities	1.500	0.100	
Depreciation, amortization and accretion	4,586	8,120	
Provision for excess and obsolete inventory	497 407	180 277	
Stock-based compensation Gain on sale of assets			
Other	(5,828) 513	853	
Increase (decrease) in cash from operating assets and liabilities	515	833	
Accounts receivable	(2.634)	2,761	
Inventory	(304)	(953)	
Other current assets	(1,042)	(1,021)	
Income taxes	(235)	87	
Accounts payable	2,070	(771)	
Accrued expenses and other liabilities	(4,573)	5,249	
Cash provided by operating activities	3,780	15,157	
Cash flows from investing activities			
Proceeds from sale of assets	9,000	—	
Capital expenditures	(1,652)	(3,498)	
Redemption of certificate of deposit - restricted	74		
Cash provided by (used in) investing activities	7,422	(3,498)	
Cash flows from financing activities			
Payments on long-term debt	(933)	(18)	
Payments for offering costs		(441)	
Other	(11)		
Cash used in financing activities	(944)	(459)	
Effect of foreign exchange rate on cash	26	(196)	
Increase in cash and cash equivalents	10,284	11,004	
Cash and cash equivalents, beginning of period	28,596	19,739	
Cash and cash equivalents, end of period	\$ 38,880	\$ 30,743	
Supplemental disclosure of cash flow information			
Interest paid	\$ 6,422	\$ 59	
Income taxes paid/(refunded), net	\$ 199	\$ (59)	

See notes to unaudited condensed consolidated financial statements.

Notes to Unaudited Condensed Consolidated Financial Statements

Unless the context otherwise requires, references to the "Company" and "Lantheus" refer to Lantheus Holdings, Inc. and its direct and indirect subsidiaries, references to "Holdings" refer to Lantheus Holdings, Inc., and not to any of its subsidiaries, and references to "LMI" refer to Lantheus Medical Imaging, Inc., the direct subsidiary of Holdings. Solely for convenience, we refer to trademarks, service marks and trade names without the TM, SM and ® symbols. Those references are not intended to indicate, in any way, that the Company will not assert, to the fullest extent permitted under applicable law, its rights to its trademarks, service marks and trade names.

1. Business Overview

Overview

Holdings, a Delaware corporation, is the parent company of LMI, also a Delaware corporation.

The Company develops, manufactures and commercializes innovative diagnostic medical imaging agents and products that assist clinicians in the diagnosis and treatment of cardiovascular and other diseases. The Company's commercial products are used by cardiologists, nuclear physicians, radiologists, internal medicine physicians, technologists and sonographers working in a variety of clinical settings. The Company sells its products to radiopharmacies, hospitals, clinics, group practices, integrated delivery networks and group purchasing organizations. The Company sells its products globally and has operations in the United States, Canada, Puerto Rico and Australia and distribution relationships in Europe, Asia Pacific and Latin America.

The Company's portfolio of 10 commercial products is diversified across a range of imaging modalities. The Company's imaging agents and products include the following:

- DEFINITY is the leading ultrasound contrast imaging agent used by cardiologists and sonographers during cardiac ultrasound, or echocardiography, exams based on revenue and usage. DEFINITY is an injectable agent that, in the United States, is indicated for use in patients with suboptimal echocardiograms to assist in the visualization of the left ventricle, the main pumping chamber of the heart. The use of DEFINITY in echocardiography allows physicians to significantly improve their assessment of the function of the left ventricle.
- TechneLite is a self-contained system, or generator, of technetium (Tc99m), a radioisotope with a six hour half-life, used by radiopharmacies to
 prepare various nuclear imaging agents.
- Xenon Xe 133 Gas, or Xenon, is a radiopharmaceutical gas that is inhaled and used to assess pulmonary function and also cerebral blood flow.
- Neurolite is an injectable, technetium-labeled imaging agent used with SPECT technology to identify the area within the brain where blood flow
 has been blocked or reduced due to stroke.
- Cardiolite is an injectable, technetium-labeled imaging agent, also known by its generic name sestamibi, used with Single Photon Emission Computed Tomography, or SPECT, technology in myocardial perfusion imaging, or MPI, procedures that assess blood flow distribution to the heart.

In the United States, the Company sells DEFINITY through its sales team that calls on healthcare providers in the echocardiography space, as well as group purchasing organizations and integrated delivery networks. The Company's radiopharmaceutical products are primarily distributed through commercial radiopharmacies owned or controlled by third parties. In Puerto Rico and Australia, the Company owns three radiopharmacies and sells its own radiopharmaceuticals, as well as others, directly to end users. In Canada, Europe, Asia Pacific and Latin America, the Company utilizes distributor relationships to market, sell and distribute its products.

Basis of Consolidation and Presentation

The financial statements have been prepared in United States dollars, in accordance with accounting principles generally accepted in the United States of America, or U.S. GAAP. The consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries. All intercompany accounts and transactions have been eliminated in consolidation.

In the opinion of the Company's management, the accompanying unaudited condensed consolidated financial statements include all adjustments, consisting of normal recurring accruals, necessary for a fair presentation of the Company's financial statements for interim periods in accordance with U.S. GAAP. Certain information and footnote disclosures normally included in

financial statements prepared in accordance with U.S. GAAP have been condensed or omitted pursuant to the rules and regulations of the Securities and Exchange Commission, or the SEC. The information included in this quarterly report should be read in conjunction with the Company's consolidated financial statements and the accompanying notes for the year ended December 31, 2015 included in the Company's Form 10-K filed with the SEC on March 2, 2016. The Company's accounting policies are described in the "Notes to Consolidated Financial Statements" in the Form 10-K and updated, as necessary, in this quarterly report. There were no changes to the Company's accounting policies since December 31, 2015. The year-end condensed consolidated balance sheet data presented for comparative purposes was derived from audited financial statements, but does not include all disclosures required by U.S. GAAP. The results of operations for the three months ended March 31, 2016 are not necessarily indicative of the operating results for the full year or for any other subsequent interim period.

The Company currently relies on Jubilant HollisterStier, or JHS, as its sole source manufacturer of DEFINITY, Neurolite and evacuation vials for TechneLite. The Company has additional ongoing technology transfer activities at JHS for its Cardiolite product supply, which is currently approved for manufacture by a single manufacturer. While the Company has ongoing technology transfer activities at Pharmalucence for the manufacture and supply of DEFINITY, such activities have been delayed and the Company cannot project when Pharmalucence will be able to manufacture and supply DEFINITY.

Until the Company successfully becomes dual sourced for its principal products, the Company is vulnerable to future supply shortages. Disruption in the financial performance of the Company could also occur if it experiences significant adverse changes in customer mix, broad economic downturns, adverse industry or Company conditions or catastrophic external events. If the Company experiences one or more of these events in the future, it may be required to implement additional expense reductions, such as a delay or elimination of discretionary spending in all functional areas, as well as scaling back select operating and strategic initiatives.

The Company has historically been dependent on key customers and group purchasing organizations for the majority of the sales of its medical imaging products. The Company's ability to maintain and profitably renew those contracts and relationships with those key customers and group purchasing organizations is an important aspect of the Company's strategy.

Borrowing capacity under the \$50.0 million revolving credit facility, or the Revolving Facility, is calculated by reference to a borrowing base consisting of a percentage of certain eligible accounts receivable, inventory and machinery and equipment minus any reserves, or the Borrowing Base. If the Company is not successful in achieving its forecasted operating results, the Company's accounts receivable and inventory could be negatively affected, thus reducing the Borrowing Base and limiting the Company's borrowing capacity. As of March 31, 2016, the aggregate Borrowing Base was approximately \$48.3 million, which was reduced by the \$8.8 million unfunded Standby Letter of Credit and \$0.1 million in accrued interest, resulting in a net Borrowing Base availability of approximately \$39.4 million. The Company's \$365.0 million senior secured term loan facility, or the Term Facility, contains a number of affirmative, negative, reporting and financial covenants, in each case subject to certain exceptions and materiality thresholds. Incremental borrowings under the Revolving Facility may affect the Company's ability to comply with the covenants in the Term Facility, including the financial covenant restricting total net leverage. Accordingly, the Company may be limited in utilizing its net Borrowing Base availability as a source of liquidity.

Based on the Company's current operating plans, the Company believes its existing cash and cash equivalents, results of operations and availability under the Revolving Facility will be sufficient to continue to fund the Company's liquidity requirements for at least the next twelve months.

Use of Estimates

The preparation of condensed consolidated financial statements in conformity with U.S. GAAP requires management to make certain estimates and assumptions that affect the reported amounts of assets and liabilities at the date of the condensed consolidated financial statements, and the reported amounts of revenues and expenses during the reporting period. The more significant estimates reflected in the Company's condensed consolidated financial statements include certain judgments regarding revenue recognition, goodwill, tangible and intangible asset valuation, inventory valuation, asset retirement obligations, income tax liabilities and related indemnification receivable, deferred tax assets and liabilities and accrued expenses. Actual results could materially differ from those estimates or assumptions.

Recent Accounting Standards

During the first quarter of 2016, the Company early adopted ASU No. 2015-17, *Income Taxes (Topic 740): Balance Sheet Classification of Deferred Taxes* on a retrospective basis. This standard requires all deferred tax assets and liabilities, and any related valuation allowances, to be classified as noncurrent on the balance sheet. Adoption of this standard has resulted in the reclassification of \$0.1 million of current deferred tax assets to noncurrent deferred tax assets and \$0.2 million of current deferred tax liabilities to noncurrent deferred tax liabilities on the balance sheet at both March 31, 2016 and December 31, 2015.

In February 2016, the Financial Accounting Standards Board, or the FASB, issued ASU No. 2016-02, *Leases (Topic 842)*, or ASU 2016-02. ASU 2016-02 supersedes the existing guidance for lease accounting, *Leases (Topic 840)*. ASU 2016-02 was issued to increase transparency and comparability among organizations by requiring lessees to recognize all lease transactions (with terms in excess of 12 months) on the balance sheet as a lease liability and a right-of-use asset (as defined). The accounting for lessors remains largely unchanged. ASU 2016-02 retains a distinction between finance leases and operating leases. For leases with a term of twelve months or less, a lessee is permitted to make an accounting policy election by class of underlying asset not to recognize a lease liability and right-of-use asset. ASU 2016-02 is effective for fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. Early application is permitted for all entities. ASU 2016-02 requires a modified retrospective approach for all leases existing at, or entered into after, the date of initial application, with an option to elect to use certain transition relief. The Company is currently evaluating the impact this ASU will have on our financial position, results of operations, cash flows and disclosures.

In May 2014, the FASB issued ASU 2014-09, *Revenue from Contracts with Customers (Topic 606)* or ASU 2014-09. ASU 2014-09 supersedes nearly all existing revenue recognition guidance under U.S. GAAP. The core principle of ASU 2014-09 is to recognize revenues when promised goods or services are transferred to customers in an amount that reflects the consideration that is expected to be received for those goods or services. ASU 2014-09 defines a five step process to achieve this core principle and, in doing so, it is possible more judgment and estimates may be required within the revenue recognition process than required under existing U.S. GAAP including identifying performance obligations in the contract, estimating the amount of variable consideration to include in the transaction price and allocating the transaction price to each separate performance obligation. In August 2015, the FASB issued ASU 2015-14, *Revenue from Contracts with Customers (Topic 606), Deferral of the Effective Date*, which defers the effective date of ASU 2014-09 to annual reporting periods beginning after December 15, 2017 with early adoption permitted as of its original effective date of December 15, 2016. In March 2016, the FASB issued ASU 2016-08, *Revenue from Contracts with Customers (Topic 606), Principal versus Agent Considerations (Reporting Revenue forss versus Net)*, which further clarifies the implementation guidance on principal versus agent considerations. The new guidance requires either a retrospective or a modified retrospective and objetion. The Company is currently evaluating the impact these ASUs will have on our financial position, results of operations, cash flows and disclosures.

In August 2014, the FASB issued ASU 2014-15, *Presentation of Financial Statements-Going Concern (Subtopic 205-4): Disclosure of Uncertainties about an Entity's Ability to Continue as a Going Concern* or ASU 2014-15. ASU 2014-15 to provide guidance on management's responsibility in evaluating whether there is substantial doubt about a company's ability to continue as a going concern and to provide related footnote disclosures. The amendments in ASU 2014-15 are effective for annual reporting periods ending after December 15, 2016. Early adoption is permitted. The Company does not anticipate this ASU will have a material impact to the Company's financial position, results of operations, cash flows and disclosures.

2. Summary of Significant Accounting Policies

Revenue Recognition

The Company recognizes revenue when evidence of an arrangement exists, title has passed, the risks and rewards of ownership have transferred to the customer, the selling price is fixed and determinable, and collectability is reasonably assured. For transactions for which revenue recognition criteria have not yet been met, the respective amounts are recorded as deferred revenue until such point in time the criteria are met and revenue can be recognized. Revenue is recognized net of reserves, which consist of allowances for returns and rebates.

Revenue arrangements with multiple elements are divided into separate units of accounting if certain criteria are met, including whether the delivered element has stand-alone value to the customer. The arrangement's consideration is then allocated to each separate unit of accounting based on the relative selling price of each deliverable. The estimated selling price of each deliverable is determined using the following hierarchy of values: (i) vendor-specific objective evidence of fair value; (ii) third-party evidence of selling price; and (iii) best estimate of selling price. The best estimate of selling price reflects the Company's best estimate of what the selling price would be if the deliverable was regularly sold by the Company on a stand-alone basis. The consideration allocated to each unit of accounting is then recognized as the related goods or services are delivered, limited to the consideration that is not contingent upon future deliverables. Supply or service transactions may involve the charge of a nonrefundable initial fee with subsequent periodic payments for future products or services. The up-front fees, even if nonrefundable, are recognized as revenue as the products and/or services are delivered and performed over the term of the arrangement.

Inventory

Inventory costs associated with product that has not yet received regulatory approval are capitalized if the Company believes there is probable future commercial use of the product and future economic benefits of the asset. If future commercial use of the product is not probable, then inventory costs associated with such product are expensed during the period the costs are incurred. There was no significant product expensed for the three months ended March 31, 2016 and 2015. At March 31, 2016 and December 31, 2015, the Company had no capitalized inventories associated with product that did not have regulatory approval.

Goodwill

Goodwill is not amortized, but is instead tested for impairment at least annually and whenever events or circumstances indicate that it is more likely than not that it may be impaired. The Company has elected to perform the annual test for goodwill impairment as of October 31 of each year. During the quarter ended March 31, 2016, there were no events that triggered an interim impairment test of goodwill.

3. Fair Value of Financial Instruments

The tables below present information about the Company's assets and liabilities that are measured at fair value on a recurring basis as of March 31, 2016 and December 31, 2015, and indicate the fair value hierarchy of the valuation techniques utilized to determine such fair value. In general, fair values determined by Level 1 inputs utilize quoted prices (unadjusted) in active markets for identical assets or liabilities. Fair values determined by Level 2 inputs utilize data points from active markets that are observable, such as quoted prices, interest rates and yield curves. Fair values determined by Level 3 inputs utilize unobservable data points for the asset or liability.

Total fair value at March 31, 2016	Quoted prices in active markets (Level 1)	Significant other observable inputs (Level 2)	Significant unobservable inputs (Level 3)
\$ 2,577	\$ 2,577	\$ —	\$
\$ 2,577	\$ 2,577	<u>\$ </u>	<u>\$ </u>
Total fair value at December 31, 2015	Quoted prices in active markets (Level 1)	Significant other observable inputs (Level 2)	Significant unobservable inputs (Level 3)
\$ 1,586	\$ 1,586	\$ _	\$
74		74	
\$ 1,660	\$ 1,586	\$ 74	\$
	value at March 31, 2016 \$ 2,577 \$ 2,577 Total fair value at December 31, 2015 \$ 1,586 74	Total fair value at active March 31, 2016prices in active markets 2016\$2,577\$\$\$2,577\$\$\$<	Total fair value at 2016prices in active marketsother observable inputs2016(Level 1) (Level 2)(Level 2)\$2,577\$2.577\$2,577\$2.577\$2,577\$2.577\$2,577\$2.577\$2,577\$\$1,586\$ <td< td=""></td<>

At December 31, 2015, the Company had a \$0.1 million certificate of deposit, which was collateral for a long-term lease and was included in other long-term assets on the condensed consolidated balance sheet. In January 2016, the certificate of deposit was redeemed. Certificates of deposit are classified within Level 2 of the fair value hierarchy, as these are not traded on the open market.

At March 31, 2016, the Company had total cash and cash equivalents of \$38.9 million, which included approximately \$2.6 million of money market funds and \$36.3 million of cash on-hand. At December 31, 2015, the Company had total cash and cash equivalents of \$28.6 million, which included approximately \$1.6 million of money market funds and \$27.0 million of cash on-hand.

The estimated fair values of the Company's financial instruments, including its cash and cash equivalents, receivables, accounts payable and accrued expenses approximate the carrying values of these instruments due to their short term nature. The estimated fair value of the Company's Term Facility at both March 31, 2016 and December 31, 2015, approximated the carrying value because the interest rate is subject to change with market interest rates.

4. Income Taxes

The Company provides for income taxes at the end of each interim period based on the estimated effective tax rate for the full fiscal year in addition to discrete events which impact the interim period. The Company's effective tax rate differs from the U.S. statutory rate principally due to the rate impact of uncertain tax positions, valuation allowance changes and state taxes. Cumulative adjustments to the tax provision are recorded in the interim period in which a change in the estimated annual effective rate is determined. The Company's tax provision was \$0.4 million for the three months ended March 31, 2016, compared to a tax benefit of \$3,000 for the three months ended March 31, 2015.

In connection with the Company's acquisition of the medical imaging business from Bristol-Myers Squibb, or BMS, in 2008, the Company obtained a tax indemnification agreement with BMS related to certain tax obligations arising prior to the acquisition of the Company, for which the Company has the primary legal obligation. The tax indemnification receivable is recognized within other noncurrent assets. The changes in the tax indemnification asset are recognized within other income (expense), net in the condensed consolidated statement of operations. In accordance with the Company's accounting policy, the change in the tax liability and penalties and interest associated with these obligations (net of any offsetting federal or state benefit) is recognized within the tax provision. Accordingly, as these reserves change, adjustments are included in the tax provision while the offsetting adjustment is included in other income (expense), net. Assuming that the receivable from BMS continues to be considered recoverable by the Company, there is no net effect on earnings related to these liabilities and no net cash outflows.

5. Assets Held for Sale

During the fourth quarter of 2015, the Company committed to a plan to sell certain assets and liabilities associated with the Company's international business in Canada. This event qualified for held for sale accounting and the Company determined that the fair value of the net assets being sold significantly exceeded the carrying value as of December 31, 2015. The transaction was finalized in the first quarter of 2016.

Effective January 7, 2016, the Canadian subsidiary of the Company entered into an asset purchase agreement, or the Purchase Agreement, pursuant to which it would sell substantially all of the assets of its Canadian radiopharmacies and Gludef manufacturing and distribution business to one of its existing Canadian radiopharmacy customers.

The purchase price for the asset sale was \$9.0 million in cash. The Purchase Agreement contained customary representations, warranties and covenants by each of the parties. Subject to certain limitations, the buyer will be indemnified for damages resulting from breaches or inaccuracies of the Company's representations, warranties and covenants in the Purchase Agreement.

As part of the transaction, the Company and the buyer also entered into a customary transition services agreement and a long-term supply contract under which the Company will supply the buyer with the Company's products on commercial terms and under which the buyer has agreed to certain product purchase commitments.

The Company did not believe the sale of certain net assets in the international business constituted a strategic shift that would have a major effect on its operations or financial results. As a result, this transaction was not classified as discontinued operations in the Company's financial statements and was classified as assets and liabilities held for sale as of December 31, 2015.

The following table summarizes the major classes of assets and liabilities sold as of January 12, 2016 (date of the sale) and December 31, 2015:

(in thousands)	January 12, 2016	December 31, 2015
Current Assets:		
Accounts receivable, net	\$ 2,565	\$ 2,512
Inventory	730	806
Other current assets	14	26
Total current assets	3,309	3,344
Non-Current Assets:		
Property, plant and equipment, net	760	791
Intangibles, net	462	480
Other long-term assets	28	29
Total assets held for sale	\$ 4,559	\$ 4,644
Current Liabilities:		
Accounts payable	\$ 435	\$ 430
Accrued expense and other liabilities	858	1,285
Total liabilities held for sale	\$ 1,293	\$ 1,715

The sale resulted in a pre-tax book gain of \$5.8 million, which was recorded within operating income in the condensed consolidated statement of operations in the quarter ended March 31, 2016.

6. Inventory

The Company includes within current assets the amount of inventory that is estimated to be utilized within twelve months. Inventory that will be utilized after twelve months is classified within other long-term assets.

Inventory, classified in inventory or other long-term assets, consisted of the following:

(in thousands)	March 31, 2016	December 31, 2015
Raw materials	\$ 6,801	\$ 7,506
Work in process	4,273	2,407
Finished goods	4,467	5,709
Inventory	15,541	15,622
Other long-term assets	1,156	1,156
Total	\$ 16,697	\$ 16,778

At both March 31, 2016 and December 31, 2015, inventories reported as other long-term assets included \$1.2 million of raw materials.

7. Property, Plant and Equipment, net

Property, plant and equipment consisted of the following:

	March 31,	December 31,
(in thousands)	2016	2015
Land	\$ 14,950	\$ 14,950
Buildings	69,222	68,941
Machinery, equipment and fixtures	61,927	60,787
Construction in progress	8,242	9,099
Accumulated depreciation	(69,017)	(67,260)
Property, plant and equipment, net	\$ 85,324	\$ 86,517

For the three months ended March 31, 2016 and 2015, depreciation expense related to property, plant and equipment was \$2.0 million and \$5.7 million, respectively.

Fixed assets dedicated to research and development, or R&D, activities, which were impacted by the March 2013 R&D strategic shift, have a carrying value of \$4.0 million as of March 31, 2016. The Company believes these fixed assets will be utilized for either internally funded ongoing R&D activities or R&D activities funded by a strategic partner. If the Company is not successful in finding a strategic partner and there are no alternative uses for these fixed assets, then they could be subject to impairment in the future.

8. Asset Retirement Obligations

The Company considers the legal obligation to remediate its facilities upon a decommissioning of its radioactive related operations as an asset retirement obligation. The operations of the Company have radioactive production facilities at its North Billerica, Massachusetts and San Juan, Puerto Rico sites.

The Company is required to provide the U.S. Nuclear Regulatory Commission and Massachusetts Department of Public Health financial assurance demonstrating the Company's ability to fund the decommissioning of the North Billerica, Massachusetts production facility upon closure, although the Company does not intend to close the facility. The Company has provided this financial assurance in the form of a \$28.2 million surety bond, which itself is currently secured by an \$8.8 million unfunded Standby Letter of Credit provided to the third party issuer of the bond.

The fair value of a liability for asset retirement obligations is recognized in the period in which the liability is incurred. As of March 31, 2016, the liability is measured at the present value of the obligation expected to be incurred, of approximately \$26.7 million, and is adjusted in subsequent periods as accretion expense is recorded. The corresponding asset retirement costs are capitalized as part of the carrying value of the related long-lived assets and depreciated over the asset's useful life.

The following is a reconciliation of the Company's asset retirement obligations for the three months ended March 31, 2016:

(in thousands)	
Balance at January 1, 2016	\$8,145
Net increase due to changes in estimated future cash flows	39
Accretion expense	233
Balance at March 31, 2016	\$8,417

9. Intangibles, net

Intangibles, net consisted of the following:

	March 31, 2016			
(in thousands)	Cost	Accumulated amortization	Net	Amortization Method
Trademarks	\$ 13,540	\$ 7,389	\$ 6,151	Straight-line
Customer relationships	100,948	89,429	11,519	Accelerated
Other patents	42,780	41,215	1,565	Straight-line
	\$157,268	\$ 138,033	\$19,235	
		December	31, 2015	
		Accumulated		Amortization
<u>(in thousands)</u>	Cost	amortization	Net	Method
Trademarks	\$ 13,540	\$ 6,934	\$ 6,606	Straight-line
Customer relationships	100,737	88,564	12,173	Accelerated
Other patents	42 780	41,063	1,717	Straight-line
Other patents	42,780	41,003	1,/1/	Straight-fine

For the three months ended March 31, 2016 and 2015, the Company recorded amortization expense for its intangible assets of \$1.3 million and \$1.5 million, respectively.

Expected future amortization expense related to the intangible assets is as follows:

(in thousands)	
Remainder of 2016	\$ 3,879
2017	3,391
2018	2,687
2019	1,835
2020	1,594
2021 and thereafter	5,849
	\$19,235

10. Accrued Expenses and Other Liabilities

Accrued expenses and other liabilities are comprised of the following:

(in thousands)	March 31, 2016	December 31, 2015
Compensation and benefits	\$ 7,061	\$ 10,525
Freight, distribution and operations	2,724	2,962
Accrued rebates, discounts and chargebacks	2,279	2,085
Accrued professional fees	1,029	1,493
Marketing expense	572	490
Research and development services	438	360
Other	742	587
	\$ 14.845	\$ 18.502

11. Financing Arrangements

Term Facility

On June 30, 2015, the Company entered into a \$365.0 million seven-year Term Facility, which was issued net of a 1.25% discount of \$4.6 million. The Company has a right to request an increase of the Term Facility in an aggregate amount up to \$37.5 million plus additional amounts subject to certain leverage ratios. The net proceeds of the Term Facility, together with the net proceeds of the initial public offering, or IPO, and cash on hand, were used to refinance in full the aggregate principal amount of the \$400.0 million 9.750% Senior Notes, or the Notes, and pay related premiums, interest and expenses.

The term loans under the Term Facility bear interest, with pricing based from time to time at the Company's election at (i) LIBOR plus a spread of 6.00% (with a LIBOR rate floor of 1.00%) or (ii) the Base Rate (as defined in our Term Facility) plus a spread of 5.00%. Interest under term loans based on (i) the LIBOR rate is payable at the end of each interest period (as defined in our Term Facility) and (ii) the Base Rate is payable at the end of each quarter. At March 31, 2016, the Company's interest rate under the Term Facility was 7.00%.

The Company is permitted to voluntarily prepay the Term Facility, in whole or in part, with a premium applicable for the first six months of the Term Facility in connection with a repricing transaction. The Company is required to make quarterly payments, which began on September 30, 2015, in an amount equal to a quarter of a percent (0.25%) per annum of the original principal amount of the Term Facility. The remaining unpaid principal amount of the Term Facility will be payable on the maturity date, or June 30, 2022.

The Term Facility will require the Company to prepay outstanding term loans, subject to certain exceptions, with:

- 100% of the net cash proceeds of all non-ordinary course sales or other dispositions of assets (including as a result of casualty or condemnation, subject to certain exceptions); the Company may reinvest or commit to reinvest certain of those proceeds in assets useful in our business within twelve months;
- 100% of the net cash proceeds from issuances or incurrence of debt, other than proceeds from debt permitted under the Term Facility and Revolving Facility;
- 50% (with two leverage-based stepdowns) of the Company's excess cash flow; and
- 50% of net payments from any Zurich insurance settlement (as defined therein).

The foregoing mandatory prepayments will be applied to the scheduled installments of principal of the Term Facility in direct order of maturity.

The Term Facility is guaranteed by the Company and Lantheus Real Estate, and obligations under the Term Facility are secured by substantially all the property and assets and all interests of the Company, LMI and Lantheus Real Estate.

The Company's minimum payments of principal obligations under the Term Facility are as follows as of March 31, 2016:

(in thousands)	
Remainder of 2016	\$ 2,738
2017	3,650
2018	3,650
2019	3,650
2020	3,650
2021 and thereafter	344,925
Total debt	362,263
Unamortized debt discount	(4,035)
Unamortized debt issuance costs	(5,229)
Total	352,999
Less current portion	(3,650)
Total long-term debt	\$349,349

Term Facility Covenants

The Term Facility contains a number of affirmative, negative, reporting and financial covenants, in each case subject to certain exceptions and materiality thresholds. The Term Facility requires the Company to be in quarterly compliance, measured on a trailing four quarter basis. The financial covenants are displayed in the table below:

Term Facility Financial Covenants

Period	Total Net Leverage Ratio
Q3 2015 to Q1 2016	6.25 to 1.00
Q2 2016 to Q4 2016	6.00 to 1.00
Q1 2017 to Q2 2017	5.50 to 1.00
Thereafter	5.00 to 1.00

The Term Facility contains usual and customary restrictions on the ability of the Company and its subsidiaries to: (i) incur additional indebtedness (ii) create liens; (iii) consolidate, merge, sell or otherwise dispose of all or substantially all of its assets; (iv) sell certain assets; (v) pay dividends on, repurchase or make distributions in respect of capital stock or make other restricted payments; (vi) make certain investments; (vii) repay subordinated indebtedness prior to stated maturity; and (viii) enter into certain transactions with its affiliates.

Revolving Line of Credit

At March 31, 2016, the Company has a Revolving Facility with an aggregate principal amount not to exceed \$50.0 million. The loans under the Revolving Facility bear interest subject to a pricing grid based on average historical excess availability, with pricing based from time to time at the election of LMI at (i) LIBOR plus a spread ranging from 2.00% or (ii) the Reference Rate (as defined in the agreement) plus 1.00%. The Revolving Facility also includes an unused line fee of 0.375% and expires on June 30, 2020.

As of March 31, 2016, the Company has an unfunded Standby Letter of Credit for up to \$8.8 million. The unfunded Standby Letter of Credit requires an annual fee, payable quarterly, which is set at LIBOR plus a spread of 2.00% and expires in February 2017. It automatically renewed for a one year period and will continue to automatically renew for a one year period at each anniversary date, unless the Company elects not to renew in writing within 60 days prior to such expiration.

The Revolving Facility is guaranteed by Holdings and Lantheus Real Estate and is secured by a pledge of substantially all of the assets of each of the loan parties including accounts receivable, inventory and machinery and equipment. Borrowing capacity is determined by reference to a Borrowing Base, which is based on a percentage of certain eligible accounts receivable, inventory and machinery and equipment minus any reserves. As of March 31, 2016, the aggregate Borrowing Base was approximately \$48.3 million, which was reduced by an outstanding \$8.8 million unfunded Standby Letter of Credit and \$0.1 million in accrued interest, resulting in a net Borrowing Base availability of approximately \$39.4 million.

Revolving Line of Credit Covenants

The Revolving Facility contains a number of affirmative, negative, reporting and financial covenants, as well as a financial covenant during trigger periods in the form of a consolidated fixed charge coverage ratio of not less than 1:00:1:00. Upon an event of default, the lender has the right to declare the loans and other obligations outstanding immediately due and payable and all commitments immediately terminated or reduced, and the lender may, after such events of default, require the Company to make deposits with respect to any outstanding letters of credit in an amount equal to 105% of the greatest amount for which such letter of credit may be drawn.

12. Stockholders' Equity

As of March 31, 2016, the authorized capital stock of the Company consisted of 250,000,000 shares of common stock, par value \$0.01 per share, and 25,000,000 shares of preferred stock, par value \$0.01 per share. The common stockholders are entitled to one vote per share and will share equally on a per share basis in any dividend declared by the Board of Directors, subject to any preferential rights of the holders of any outstanding preferred stock.

The following table presents the changes in stockholders' deficit for the three months ended March 31, 2016:

	Common	Stock			Accumulated	
	Shares	Amount	Additional Paid-In Capital	Accumulated Deficit	Other Comprehensive Loss	Total Stockholders' Deficit
Balance at January 1, 2016	30,364,501	\$ 303	\$175,553	\$ (359,160)	\$ (1,985)	\$ (185,289)
Net income	—			10,323		10,323
Other comprehensive income	—	—	—	_	340	340
Vesting of restricted stock awards	12,603	—	—	—	—	—
Stock-based compensation			407			407
Balance at March 31, 2016	30,377,104	\$ 303	\$175,960	<u>\$ (348,837</u>)	\$ (1,645)	\$ (174,219)

13. Stock-Based Compensation

As of June 24, 2015, the Company adopted the 2015 Equity Incentive Plan, or the 2015 Plan.

The Company's employees are eligible to receive awards under the 2015 Plan. The 2015 Plan is administered by the Board of Directors and permits the granting of stock options, stock appreciation rights, or SARs, restricted stock, restricted stock units and dividend equivalent rights ("DERs") to employees, officers, directors and consultants of the Company. The Board of Directors may, at its sole discretion, grant DERs with respect to any award and such DER is treated as a separate award. The number of shares authorized for issuance under the 2015 Plan increased from 2,190,320 to 4,555,277 on April 26, 2016. Option awards under the 2015 Plan are granted with an exercise price equal to the fair value of the Company's common stock at the date of grant. Time based option awards vest based on time, typically four years, and performance based option awards vest based on the performance criteria specified in the grant. All option awards have a ten-year contractual term. The Company recognizes compensation costs for its time based awards on a straight-line basis equal to the vesting period. The compensation cost for performance based awards is recognized on a graded vesting basis, based on the probability of achieving the performance targets over the requisite service period for the entire award. The fair value of each option award is estimated on the date of grant using a Black-Scholes valuation model. Expected volatilities are based on the historic volatility of a selected peer group. Expected dividends represent the dividends expected to be issued at the date of grant. The expected term of options represents the period of time that options granted are expected to be outstanding. The risk-free interest rate assumption is the U.S. Treasury rate at the date of the grant which most closely resembles the expected life of the options.

Stock-based compensation expense for both time based and performance based stock options, restricted stock awards and common stock grants were recognized in the condensed consolidated statements of operations as follows:

		Months ded 2h 31,
(in thousands)	2016	2015
Cost of goods sold	\$ 67	\$ (1)
Sales and marketing	48	37
General and administrative	233	213
Research and development	59	28
Total stock-based compensation expense	\$407	\$277

14. Net Income Per Share

Basic income (loss) per share is computed by dividing net income (loss) by the weighted average number of shares of common stock outstanding during the period. Diluted earnings per common share is computed by dividing net income by the weighted average number of shares of common stock outstanding during the period, plus the potential dilutive effect of other securities if those securities were converted or exercised. During periods in which the Company incurs net losses, both basic and diluted loss per share is calculated by dividing the net loss by the weighted average shares outstanding and potentially dilutive securities are excluded from the calculation because their effect would be antidilutive.

		Three Months Ended March 31,		
(in thousands, except share and per share amounts)	2016	2015		
Net income	\$ 10,323	\$ 375		
Basic weighted average common shares outstanding	30,368,240	18,080,944		
Effect of dilutive restricted stock awards	4,451	_		
Effect of dilutive stock options		323,449		
Diluted weighted average common shares outstanding	30,372,691	18,404,393		
Basic and diluted income per common share	\$ 0.34	\$ 0.02		

The weighted average number of common shares for the three months ended March 31, 2016 and 2015, did not include 2,221,940 and 751,964 options and unvested restricted stock, respectively, because of their antidilutive effect.

15. Other Income (Expense), net

Other income (expense), net consisted of the following:

	Three N End Marci	led
(in thousands)	2016	2015
Foreign currency losses	\$(237)	\$(378)
Tax indemnification income (expense)	296	(4)
Other expense	(1)	(1)
Total other income (expense), net	<u>\$ 58</u>	\$(383)

16. Legal Proceedings and Contingencies

From time to time, the Company is a party to various legal proceedings arising in the ordinary course of business. In addition, the Company has in the past been, and may in the future be, subject to investigations by governmental and regulatory authorities, which expose it to greater risks associated with litigation, regulatory or other proceedings, as a result of which the Company could be required to pay significant fines or penalties. The outcome of litigation, regulatory or other proceedings cannot be predicted with certainty, and some lawsuits, claims, actions or proceedings may be disposed of unfavorably to the Company. In addition, intellectual property disputes often have a risk of injunctive relief which, if imposed against the Company, could materially and adversely affect its financial condition or results of operations. As of March 31, 2016, the Company had no material ongoing litigation in which the Company was a defendant or any material ongoing regulatory or other proceedings and had no knowledge of any investigations by government or regulatory authorities in which the Company is a target that could have a material adverse effect on its current business.

On December 16, 2010, LMI filed suit against one of its insurance carriers seeking to recover business interruption losses associated with the NRU reactor shutdown and the ensuing global Moly supply shortage. The claim is the result of the shutdown of the NRU reactor in Chalk River, Ontario. The NRU reactor was off-line from May 2009 until August 2010. The defendant answered the complaint on January 21, 2011, denying substantially all of the allegations, presenting certain defenses and requesting dismissal of the case with costs and disbursements. Discovery, including international discovery and related motion practice, has been on-going for more than three years. The defendant filed a motion for summary judgment on July 14, 2014. The Company filed a memorandum of law in opposition to defendant's motion for summary judgment on August 25, 2014. The defendant filed a reply memorandum of law in further support of its motion for summary judgment on September 15, 2014. Expert witness discovery was completed on October 31, 2014. On March 25, 2015, the United States District Court for the Southern District of New York granted defendant's motion for summary judgment. On September 4, 2015, the Company filed an appeal of the District Court decision with the United States Court of Appeals for the Second Circuit. On December 4, 2015, the United States Court of Appeals for the Second Circuit heard oral arguments of the Company and the defendant in connection with the Company's appeal. The Company filed a reply brief to the defendant in connection with the Company's appeal. The Company and the defendant in connection with the Company's appeal.



17. Related Party Transactions

Avista, the Company's majority shareholder, provided certain advisory services to the Company pursuant to an advisory services and monitoring agreement. The Company was required to pay an annual fee of \$1.0 million and other reasonable and customary advisory fees, as applicable, paid on a quarterly basis. The initial term of the agreement was seven years. On June 25, 2015, the Company exercised its right to terminate its advisory services and monitoring agreement with Avista. In connection with such termination, the Company has paid Avista Capital Holdings, L.P. an aggregate termination fee of \$6.5 million, which was included in general and administrative expenses in the condensed consolidated statement of operations during the quarter ended June 30, 2015. During the three months ended March 31, 2016, the Company did not incur any costs associated with this agreement as compared to \$0.3 million for the prior year comparative period. At both March 31, 2016 and December 31, 2015, there were no amounts outstanding.

In the first quarter of 2016, the Company entered into a services agreement with INC Research, LLC, or INC, to provide pharmacovigilance services. Avista and certain of its affiliates are principal owners of both INC and the Company. The agreement has a term of three years. During the three months ended March 31, 2016, the Company incurred costs associated with this agreement of approximately \$0.2 million. At March 31, 2016, \$0.2 million was included in accrued expenses and other liabilities.

The Company purchases inventory supplies from VWR Scientific, or VWR. Avista and certain of its affiliates are principal owners of both VWR and the Company. During the three months ended March 31, 2016 and 2015, the Company made purchases of \$103,000 and \$71,000, respectively. At March 31, 2016 and December 31, 2015, \$1,000 and \$10,000, respectively, was included in accounts payable and accrued expenses and other liabilities.

18. Segment Information

The Company reports two operating segments, U.S. and International, based on geographic customer base. The results of these operating segments are regularly reviewed by the Company's chief operating decision maker, the President and Chief Executive Officer. The Company's segments derive revenues through the manufacturing, marketing, selling and distribution of medical imaging products, focused primarily on cardiovascular diagnostic imaging. The U.S. segment comprises 84.9% and 81.1% of consolidated revenues for the three months ended March 31, 2016 and 2015, respectively and 93.2% and 92.0% of consolidated assets at March 31, 2016 and December 31, 2015, respectively. All goodwill has been allocated to the U.S. operating segment.

Selected information for each business segment are as follows (in thousands):

		Three Months Ended March 31,	
	2016	2015	
Revenues			
U.S.	\$70,770	\$ 65,788	
International	11,541	14,156	
Total revenue, including inter-segment	82,311	79,944	
Less inter-segment revenue	(5,837)	(5,121)	
	\$76,474	\$ 74,823	
Revenues from external customers			
U.S.	\$64,933	\$ 60,667	
International	11,541	14,156	
	\$76,474	\$ 74,823	
Operating income			
U.S.	\$13,403	\$ 12,679	
International	4,135	(1,474)	
Total operating income, including inter-segment	17,538	11,205	
Inter-segment operating income	135	173	
Operating income	17,673	11,378	
Interest expense, net	(7,018)	(10,623)	
Other income (expense), net	58	(383)	
Income before income taxes	\$10,713	\$ 372	
Total operating income, including inter-segment Inter-segment operating income Operating income Interest expense, net Other income (expense), net	17,538 135 17,673 (7,018) 58	11,205 173 11,378 (10,623) (383)	

In the table below, the Company has revised its previous presentation of U.S. and International total assets as of December 31, 2015, to correctly reflect the allocation of total assets between U.S. and International segments. This revision resulted in a decrease to total International assets of \$12.7 million with a corresponding increase to total U.S. assets in the same amount. This revision had no impact on the balance sheet, statement of operations or statement of cash flows for the year ended December 31, 2015.

	March 31, 2016	December 31, 2015	
Total assets			
U.S.	\$232,217	\$ 222,926	
International	17,046	19,453	
	\$249,263	\$ 242,379	

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Cautionary Note Regarding Forward-Looking Statements

Some of the statements contained in this quarterly report are forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. These forward-looking statements, including, in particular, statements about our plans, strategies, prospects and industry estimates are subject to risks and uncertainties. These statements identify prospective information and include words such as "anticipates," "intends," "plans," "seeks," "believes," "estimates," "expects," "should," "could," "predicts," "targets, "hopes" and similar expressions. Examples of forward-looking statements include, but are not limited to, statements we make regarding: (i) our outlook and "targets," expectations including, without limitation, in connection with continued market expansion and penetration for our commercial products, particularly DEFINITY in the face of increased competition; (ii) our outlook and expectations in connection with future performance of Xenon in the face of increased competition; (iii) our outlook and expectations related to products manufactured at JHS and Pharmalucence and global isotope supply; (iv) our outlook and expectations related to our intention to seek to engage strategic partners to assist in developing and potentially commercializing development candidates; and (v) our liquidity, including our belief that our existing cash, cash equivalents, anticipated revenues and availability under our revolving credit facility, or Revolving Facility, are sufficient to fund our existing operating expenses, capital expenditures and liquidity requirements for at least the next twelve months. Forward-looking statements are based on our current expectations and assumptions regarding our business, the economy and other future conditions. Because forward-looking statements relate to the future, they are subject to inherent uncertainties, risks and changes in circumstances that are difficult to predict. Our actual results may differ materially from those contemplated by the forward-looking statements. They are neither statements of historical fact nor guarantees or assurances of future performance. The matters referred to in the forward-looking statements contained in this prospectus may not in fact occur. We caution you therefore against relying on any of these forward-looking statements. Important factors that could cause actual results to differ materially from those in the forward-looking statements include regional, national or global political, economic, business, competitive, market and regulatory conditions and the following:

- our ability to continue to increase segment penetration for DEFINITY in suboptimal echocardiograms and the increased segment competition from other echocardiography contrast agents, including Optison from GE Healthcare and Lumason from Bracco Diagnostics Inc., or Bracco;
- risks associated with revenues and unit volumes for Xenon in pulmonary studies with increased segment competition resulting from Mallinckrodt's recent re-launch of their Xenon product;
- our dependence on key customers and group purchasing organization arrangements for our medical imaging products, and our ability to
 maintain and profitably renew our contracts and relationships with those key customers and group purchasing organizations;
- our dependence upon third parties for the manufacture and supply of a substantial portion of our products, including for DEFINITY at JHS;
- risks associated with the technology transfer programs to secure production of our products at alternate contract manufacturer sites, including for DEFINITY at Pharmalucence where activities have been significantly delayed;
- risks associated with the manufacturing and distribution of our products and the regulatory requirements related thereto;
- the instability of the global Molybdenum-99, or Moly, supply;
- the dependence of certain of our customers upon third party healthcare payors and the uncertainty of third party coverage and reimbursement rates;
- uncertainties regarding the impact of U.S. healthcare reform on our business, including related reimbursements for our current and potential future products;
- our being subject to extensive government regulation and our potential inability to comply with those regulations;
- · potential liability associated with our marketing and sales practices;
- the occurrence of any side effects with our products;
- our exposure to potential product liability claims and environmental liability;
- risks associated with our lead agent in development, flurpiridaz F 18, including our ability to:
 - attract strategic partners to successfully complete the Phase 3 clinical program and possibly commercialize the agent;
 - obtain Food and Drug Administration, or FDA, approval; and
 - gain post-approval market acceptance and adequate reimbursement;

- risks associated with being able to negotiate in a timely manner relationships with potential strategic partners to advance our other development
 programs on acceptable terms, or at all;
- the extensive costs, time and uncertainty associated with new product development, including further product development relying on external development partners, all against an evolving diagnostic landscape;
- our inability to protect our intellectual property and the risk of claims that we have infringed on the intellectual property of others;
- risks associated with prevailing economic conditions and financial, business and other factors beyond our control;
- risks associated with our international operations;
- our inability to adequately protect our facilities, equipment and technology infrastructure;
- our inability to hire or retain skilled employees and key personnel;
- risks related to our outstanding indebtedness and our ability to satisfy those obligations;
- · costs and other risks associated with the Sarbanes-Oxley Act and the Dodd-Frank Act; and
- risks related to the ownership of our common stock.

Factors that could cause or contribute to such differences include, but are not limited to, those that are discussed in other documents we file with the Securities and Exchange Commission. Any forward-looking statement made by us in this quarterly report speaks only as of the date on which it is made. Factors or events that could cause our actual results to differ may emerge from time to time, and it is not possible for us to predict all of them. We undertake no obligation to publicly update any forward-looking statement, whether as a result of new information, future developments or otherwise, except as may be required by law.

The following discussion and analysis of our financial condition and results of operations should be read together with the condensed consolidated financial statements and the related notes included in Item 1 of this Quarterly Report on Form 10-Q as well as the other factors described in "Risk Factors" in our Annual Report on Form 10-K for the fiscal year ended December 31, 2015.

Overview

We are a global leader in the development, manufacture and commercialization of innovative diagnostic medical imaging agents and products that assist clinicians in the diagnosis and treatment of cardiovascular and other diseases. Our agents are routinely used to diagnose coronary artery disease, congestive heart failure, stroke, peripheral vascular disease and other diseases. Clinicians use our imaging agents and products across a range of imaging modalities, including nuclear imaging, echocardiography and magnetic resonance imaging, or MRI. We believe that the resulting improved diagnostic information enables healthcare providers to better detect and characterize, or rule out, disease, potentially achieving improved patient outcomes, reducing patient risk and limiting overall costs for payers and the entire healthcare system.

Our commercial products are used by cardiologists, nuclear physicians, radiologists, internal medicine physicians, sonographers and technologists working in a variety of clinical settings. We sell our products to hospitals, clinics, group practices, integrated delivery networks, group purchasing organizations and radiopharmacies.

We sell our products globally and have operations in the United States, Puerto Rico, Canada and Australia and third-party distribution relationships in Europe, Asia Pacific and Latin America.

Our Products

Our principal products include the following:

DEFINITY is an ultrasound contrast agent used in ultrasound exams of the heart, also known as echocardiography exams. DEFINITY contains perflutren-containing lipid microspheres and is indicated in the United States for use in patients with suboptimal echocardiograms to assist in imaging the left ventricular chamber and left endocardial border of the heart in ultrasound procedures. We launched DEFINITY in 2001, and its last issued patent in the United States will currently expire in 2021 and in numerous foreign jurisdictions in 2019. We also have an active next generation development program for this agent.

TechneLite is a technetium generator which provides the essential nuclear material used by radiopharmacies to radiolabel Cardiolite, Neurolite and other technetium-based radiopharmaceuticals used in nuclear medicine procedures. TechneLite uses Moly as its main active ingredient.

Xenon is a radiopharmaceutical gas that is inhaled and used to assess pulmonary function and also cerebral blood flow. Xenon is manufactured by a third party and packaged by us.

Sales of our contrast agent, DEFINITY, are made in the United States and Canada through our sales team of approximately 80 employees. In the United States, our nuclear imaging products, including TechneLite, Xenon, Cardiolite and Neurolite, are primarily distributed through commercial radiopharmacies, the majority of which are controlled by or associated with Cardinal, UPPI, GE Healthcare and Triad. A small portion of our nuclear imaging product sales in the United States are made through our direct sales force to hospitals and clinics that maintain their own in-house radiopharmaceutical capabilities. Outside the United States, we own two radiopharmacies in Australia and one in Puerto Rico. On January 12, 2016, we sold our Canadian radiopharmacies to Isologic and entered into a long-term supply agreement with Isologic under which we will supply Isologic with certain of our products on commercial terms, including a material breach by the other party, bankruptcy by either party and certain force majeure events. We also maintain our own direct sales forces in these markets so we can control the importation, marketing, distribution and sale of our imaging agents in these regions. In Europe, Asia Pacific and Latin America, we rely on third party distributors to market, sell and distribute our nuclear imaging and contrast agent products, either on a country-by-country basis or on a multicountry regional basis.

The following table sets forth our revenue derived from our principal products:

		Three Months Ended March 31,		
(dollars in thousands)	2016	%	2015	%
DEFINITY	\$31,422	41.1	\$25,666	34.3
TechneLite	24,836	32.5	20,860	27.9
Xenon	8,174	10.7	13,194	17.6
Other	12,042	15.7	15,103	20.2
Revenues	<u>\$76,474</u>	100.0	\$74,823	100.0

Key Factors Affecting Our Results

Our business and financial performance have been, and continue to be, affected by the following:

Growth of DEFINITY

We believe the market opportunity for our contrast agent, DEFINITY, remains significant. DEFINITY is currently our fastest growing and highest margin commercial product. We believe that DEFINITY sales will continue to grow and that DEFINITY will constitute a greater share of our overall product mix. As we better educate the physician and healthcare provider community about the benefits and risks of this product, we believe we will experience further penetration of suboptimal echocardiograms.

Prior to the supply issues with Ben Venue Laboratories in 2012, sales of DEFINITY continually increased year-over-year since June 2008, when the boxed warning on DEFINITY was modified. Unit sales of DEFINITY had decreased substantially in late 2007 and early 2008 as a result of an FDA request in October 2007 that we and GE Healthcare, which distributes Optison, a competitor to DEFINITY, add a boxed warning to their products to notify physicians and patients about potentially serious safety concerns or risks posed by the products. However, in May 2008, the FDA boxed warning was modified in response to the substantial advocacy efforts of prescribing physicians. In October 2011, we received FDA approval of further modifications to the DEFINITY label, including: further relaxing the boxed warning; eliminating the sentence in the Indication and Use section "The safety and efficacy of DEFINITY with exercise stress or pharmacologic stress testing have not been established" (previously added in October 2007 in connection with the imposition of the box warning); and including summary data from the post-approval CaRES (Contrast echocardiography Registry for Safety Surveillance) safety registry and the post-approval pulmonary hypertension study. Bracco's ultrasound contrast agent, Lumason, has substantially similar safety labeling as DEFINITY and Optison. The future growth of our DEFINITY sales will be dependent on our ability to continue to increase segment penetration for DEFINITY in suboptimal echocardiograms" of our annual report on Form 10-K for the fiscal year ended market penetration for the appropriate use of DEFINITY in suboptimal echocardiograms" of our annual report on Form 10-K for the fiscal year ended December 31, 2015.

There are three echocardiography contrast agents approved by the FDA for sale in the U.S.—DEFINITY which as of December 2015 had an approximately 78% segment share, Optison, and Lumason, which was approved by the FDA in October 2014. Lumason

is known as SonoVue outside of the U.S. and is already approved for sale in Europe and certain Asian markets, including China, Japan and Korea. While we believe that additional promotion in the U.S. echocardiography segment will help raise awareness around the value that echocardiography contrast brings and potentially increase the overall contrast penetration rate, if Bracco successfully commercializes Lumason in the U.S. without otherwise increasing the overall usage of ultrasound contrast agents, our own growth expectations for DEFINITY revenue, gross profit and gross margin may have to be adjusted.

Competition for Xenon

Xenon gas for lung ventilation diagnosis is our third largest product by revenue. Historically, several companies, including Mallinckrodt, sold packaged Xenon as a pulmonary imaging agent in the U.S., but since 2010 we have been the only supplier of this imaging agent in the U.S. In March 2016, Mallinckrodt received regulatory approval from the FDA to again sell packaged Xenon in the U.S. and has begun to do so. Depending upon the pricing, extent of availability and market penetration of Mallinckrodt's offering, we believe we are at risk for volume loss and price erosion from those customers that are not subject to price or volume commitments with us. In order to increase the predictability of our Xenon business, we have entered into Xenon supply agreements at committed volumes and substantially reduced prices with previously non-contracted customers. These steps should result in more predictable Xenon unit volumes in 2016, but with sales at substantially lower revenue and gross margin contributions as compared to 2015. See "Part II—Item 1A. Risk Factors —We face potential supply and demand challenges for Xenon."

Inventory Supply

Our products consist of contrast imaging agents and radiopharmaceuticals (including technetium generators). We obtain a substantial portion of our imaging agents from third party suppliers. JHS is currently our sole source manufacturer of DEFINITY, Neurolite and evacuation vials, an ancillary component for our TechneLite generators, and we have ongoing technology transfer activities at JHS for our Cardiolite product supply. In the meantime, our Cardiolite product supply is approved for manufacture by a single manufacturer. Until JHS is approved by certain foreign regulatory authorities to manufacture certain of our products, we will face continued limitations on where we can sell those products outside of the United States.

In addition to JHS, we are also currently working to secure additional alternative suppliers for our key products as part of our ongoing supply chain diversification strategy. On November 12, 2013, we entered into a Manufacturing and Supply Agreement with Pharmalucence to manufacture and supply DEFINITY. However, these activities have been delayed and we cannot project when Pharmalucence will be able to manufacture and supply DEFINITY. See "Part I – Item IA. Risk Factors – Our dependence upon third parties for the manufacture and supply of a substantial portion of our products could prevent us from delivering our products to our customers in the required quantities, within the required timeframes, or at all, which could result in order cancellations and decreased revenues" of our annual report on Form 10-K for the fiscal year ended December 31, 2015.

Radiopharmaceuticals are decaying radioisotopes with half-lives ranging from a few hours to several days. These products cannot be kept in inventory because of their limited useful lives and are subject to just-in-time manufacturing, processing and distribution, which takes place at our North Billerica, Massachusetts facility.

Global Isotope Supply

Currently, our largest supplier of Moly and our only supplier of Xenon is Nordion, which relies on the NRU reactor in Chalk River, Ontario. For Moly and Xenon, we have supply agreements with Nordion that expire on October 31, 2016, and for Moly, supply agreements with NTP of South Africa, ANSTO of Australia, and IRE of Belgium, each running through December 31, 2017. The Canadian government required the NRU reactor to shut down for at least four weeks at least once a year for inspection and maintenance. The 2015 shutdown period ran from April 13, 2015 until May 13, 2015, and we were able to source all of our standing order customer demand for Moly during this time period from our other suppliers. However, because Xenon is a by-product of the Moly production process and is currently captured only by Nordion, during this shutdown period, we were not able to supply all of our standing order customer demand for Xenon during the outage. Because the month-long NRU shutdown was fully anticipated in our 2015 budgeting process, the shutdown did not have a material adverse effect on our 2015 results of operations, financial condition and cash flows.

We believe we are well-positioned with our current supply partners to have a secure supply of Moly, including low-enriched uranium, or LEU, Moly, when the NRU reactor transitions in October 2016 from providing regular supply of medical isotopes to providing only emergency back-up supply of HEU based medical isotopes through March 2018. ANSTO has under construction, in cooperation with NTP, a new Moly processing facility that ANSTO believes will expand its production capacity by approximately 2.5 times, with expanded commercial production planned to start in the latter part of 2016. In addition, IRE recently received approval from its regulator to expand its production capability by up to 50% of its former capacity. The new ANSTO and IRE production capacity is expected to replace the NRU's current routine production. In January 2015, we announced entering into a new strategic

agreement with IRE for the future supply of Xenon. Under the terms of the agreement, IRE will provide bulk Xenon to us for processing and finishing once development work has been completed and all necessary regulatory approvals have been obtained. We currently estimate commercial production will occur in 2016. If we are not able to begin providing commercial quantities of Xenon prior to the NRU reactor's supply transition in 2016, there may be a period of time during which we are not able to offer Xenon in our portfolio of commercial products. See "Part II—Item 1A. Risk Factors—We face potential supply and demand challenges for Xenon."

Demand for TechneLite

Since the global Moly supply shortage in 2009 to 2010, we have experienced reduced demand for TechneLite generators from pre-shortage levels even though volume has increased in absolute terms from levels during the shortage following the return of our normal Moly supply in August 2010. However, we do not know if overall industry demand for technetium will ever return to pre-shortage levels. See "Part I – Item 1A. Risk Factors—The Moly supply shortage caused by the 2009-10 NRU reactor shutdown has had a negative effect on the demand for some of our products, which will likely continue in the future" of our annual report on Form 10-K for the fiscal year ended December 31, 2015.

Separate from the Moly supply shortage, we believe there has also been a decline in the MPI study market because of industry-wide cost containment initiatives that have resulted in a transition of where imaging procedures are performed, from free-standing imaging centers to the hospital setting. While the total number of patient studies has not returned to pre-shortage levels, the total MPI market was essentially flat for the period 2011 through 2014.

In November 2015, CMS announced the 2016 final Medicare payment rules for hospital outpatient settings. Under the final rules, each technetium dose produced from a generator for a diagnostic procedure in a hospital outpatient setting is reimbursed by Medicare at a higher rate if that technetium dose is produced from a generator containing Moly sourced from at least 95 percent LEU. In January 2013, we began to offer a TechneLite generator which contains Moly sourced from at least 95 percent LEU and which satisfies the requirements for reimbursement under this incentive program. Although demand for LEU generators appears to be growing, we do not know when, or if, this incremental reimbursement for LEU Moly generators will result in a material increase in our generator sales.

Research and Development Expenses

To remain a leader in the marketplace, we have historically made substantial investments in new product development. As a result, the positive contributions of those internally funded R&D programs have been a key factor in our historical results and success. In March 2013, we began to implement a strategic shift in how we fund our important R&D programs. We have reduced our internal R&D resources while at the same time we are seeking to engage strategic partners to assist us in the further development and commercialization of our important agents in development, including flurpiridaz F 18, 18F LMI 1195 and LMI 1174. As a result of this shift, we are seeking strategic partners to assist us with the further two important agents in development, 18F LMI 1195 and LMI 1174, we are also seeking to engage strategic partners to assist us with the ongoing development activities relating to these agents.

Segments

We report our results of operations in two operating segments: United States and International. We generate a greater proportion of our revenue and net income in the United States segment, which consists of all regions of the United States with the exception of Puerto Rico.

Executive Overview

Our results in the three months ended March 31, 2016 reflect the following:

- increased revenues and segment penetration for DEFINITY in the suboptimal echocardiogram segment as a result of our sales efforts and sustained availability of product supply;
- increased revenues for TechneLite, mainly the result of a contract with a significant customer;
- decreased revenues for Xenon, mainly the result of lower selling prices;
- \$5.8 million gain on the sale of our Canadian radiopharmacies;
- lower international revenues as a result of the sale of our Canadian radiopharmacies and unfavorable exchange rates;
- decreased depreciation over the prior year period associated with the scheduled decommissioning of certain long-lived assets in the prior year, and
- decreased interest expense due to the refinancing of long-term debt in connection with the IPO.

Results of Operations

	For the Thr Ended Ma	
(dollars in thousands)	2016	2015
Revenues	\$76,474	\$ 74,823
Cost of goods sold	42,773	39,054
Gross profit	33,701	35,769
Operating expenses		
Sales and marketing expenses	9,307	9,072
General and administrative expenses	9,513	9,123
Research and development expenses	3,036	6,196
Total operating expenses	21,856	24,391
Gain on sale of assets	5,828	
Operating income	17,673	11,378
Interest expense, net	(7,018)	(10,623)
Other income (expense), net	58	(383)
Income before income taxes	10,713	372
Provision (benefit) for income taxes	390	(3)
Net income	<u>\$10,323</u>	\$ 375

Revenues

Revenues are summarized as follows:

		Three Months Ended March 31,	
(dollars in thousands)	2016	2015	
United States			
DEFINITY	\$30,793	\$25,182	
TechneLite	21,733	18,173	
Xenon	8,172	13,186	
Other	4,235	4,126	
Total U.S. revenues	64,933	60,667	
International			
DEFINITY	629	484	
TechneLite	3,103	2,687	
Xenon	2	8	
Other	7,807	10,977	
Total International revenues	11,541	14,156	
Revenues	\$76,474	\$74,823	

Total revenues increased \$1.7 million, or 2.2%, to \$76.5 million in the three months ended March 31, 2016, as compared to \$74.8 million in the three months ended March 31, 2015. U.S. segment revenue increased \$4.3 million, or 7.0%, to \$65.0 million in the three months ended March 31, 2016, as compared to \$60.7 million in the prior year period. The International segment revenues decreased \$2.6 million, or 18.5%, to \$11.5 million in the three months ended March 31, 2016, as compared to \$14.2 million in the prior year period.

The increase in U.S. segment revenues for the three months ended March 31, 2016, as compared to the prior year period is primarily due to a \$5.6 million increase in DEFINITY revenues as a result of higher unit volumes and a \$3.6 million increase in TechneLite revenues as a result of a contract with a significant customer that increased unit volumes. Offsetting these increases was a \$5.0 million decrease in Xenon revenues over the prior year period primarily as a result of a contract with a significant customer that reduced unit pricing in exchange for committed volume purchases.

The decrease in the International segment revenues for the three months ended March 31, 2016, as compared to the prior year period is primarily a result of the decrease in revenues from third party products in Canada because of the sale of the Canadian radiopharmacies and a \$0.8 million unfavorable foreign exchange.

Rebates and Allowances

Estimates for rebates and allowances represent our estimated obligations under contractual arrangements with third parties. Rebate accruals and allowances are recorded in the same period the related revenue is recognized, resulting in a reduction to revenue and the establishment of a liability which is included in accrued expenses. These rebates result from performance-based offers that are primarily based on attaining contractually specified sales volumes and growth, Medicaid rebate programs for certain products, administrative fees of group purchasing organizations, royalties and certain distributor related commissions. The calculation of the accrual for these rebates and allowances is based on an estimate of the third party's buying patterns and the resulting applicable contractual rebate or commission rate(s) to be earned over a contractual period.

An analysis of the amount of, and change in, reserves is summarized as follows:

(dollars in thousands)	Rebates	Allowances	Total
Balance, as of January 1, 2016	\$ 2,303	\$ 38	\$ 2,341
Current provisions relating to revenues in current year	1,644	52	1,696
Adjustments relating to prior years' estimate	(39)		(39)
Payments/credits relating to revenues in current year	(582)	(25)	(607)
Payments/credits relating to revenues in prior years	(1,047)	(49)	(1,096)
Balance, as of March 31, 2016	\$ 2,279	\$ 16	\$ 2,295

Accrued sales rebates were approximately \$2.3 million at both March 31, 2016 and December 31, 2015.

Costs of Goods Sold

Cost of goods sold consists of manufacturing, distribution, intangible asset amortization and other costs related to our commercial products. In addition, it includes the write-off of excess and obsolete inventory.

Cost of goods sold is summarized as follows:

		Three Months Ended March 31,	
(dollars in thousands)	2016	2015	
United States	\$33,210	\$26,862	
International	9,563	12,192	
Total Cost of Goods Sold	\$42,773	\$39,054	

Total cost of goods sold increased \$3.7 million, or 9.5%, to \$42.8 million in the three months ended March 31, 2016, as compared to \$39.1 million in the three months ended March 31, 2015. U.S. segment cost of goods sold increased approximately \$6.3 million, or 23.6%, to \$33.2 million in the three months ended March 31, 2016, as compared to \$26.9 million in the prior year period. For the three months ended March 31, 2016, the International segment cost of goods sold decreased \$2.6 million, or 21.6%, to \$9.6 million, as compared to \$12.2 million in the prior year period.

The increase in the U.S. segment cost of goods sold for the three months ended March 31, 2016 over the prior year period is primarily due to a \$4.7 million increase in TechneLite cost of goods sold and an increase of \$1.3 million in Xenon cost of goods sold due to higher material costs.

The decrease in the International segment cost of goods sold in the three months ended March 31, 2016, as compared to the prior year period, is primarily due to lower manufacturing costs for certain products as a result of the sale of our Canadian radiopharmacies.

Gross Profit

		Three Months Ended March 31,	
(dollars in thousands)	2016	2015	
United States	\$31,723	\$33,805	
International	1,978	1,964	
Total Gross Profit	\$33,701	\$35,769	

Total gross profit decreased \$2.1 million, or 5.8%, to \$33.7 million in the three months ended March 31, 2016, as compared to \$35.8 million in the three months ended March 31, 2015. U.S. segment gross profit decreased \$2.1 million, or 6.2%, to \$31.7 million in the three months ended March 31, 2016, as compared to \$33.8 million in the prior year period. For the three months ended March 31, 2016, the International segment gross profit remained consistent.

The decrease in the U.S. segment gross profit for the three months ended March 31, 2016 over the prior year period is primarily due to a decrease in Xenon gross profit due to lower selling prices and a decrease in TechneLite gross profit due to lower selling prices, which was offset by increased volumes. Offsetting these decreases is an increase in DEFINITY gross profit due to higher unit volumes.

The International segment gross profit for the three months ended March 31, 2016 remained consistent as compared to the prior year period despite the decrease in revenue since it fully offset the decrease in cost of goods sold as a result of the sale of our Canadian radiopharmacies.

Sales and Marketing

	Three 1 Ended M	Months Iarch 31,
(dollars in thousands)	2016	2015
United States	\$8,305	\$8,068
International	1,002	1,004
Total Sales and Marketing	\$9,307	\$9,072

Sales and marketing expenses consist primarily of salaries and other related costs for personnel in field sales, marketing, business development and customer service functions. Other costs in sales and marketing expenses include the development and printing of advertising and promotional material, professional services, market research and sales meetings.

Total sales and marketing expenses increased \$0.2 million, or 2.6%, to \$9.3 million in the three months ended March 31, 2016, as compared to \$9.1 million in the three months ended March 31, 2015. In the U.S. segment, sales and marketing expense increased \$0.2 million, or 2.9%, to \$8.3 million in the three months ended March 31, 2016, as compared to \$8.1 million in the prior year period. In the International segment, sales and marketing expense remained consistent.

The increase in the U.S. segment sales and marketing expenses for the three months ended March 31, 2016 over the prior year period is primarily due to higher salary and benefits expenses mainly driven by higher sales incentive compensation due to the increased DEFINITY revenue.

General and Administrative

	Three	Months
	Ended M	larch 31,
(dollars in thousands)	2016	2015
United States	\$9,154	\$8,740
International	359	383
Total General and Administrative	<u>\$9,513</u>	\$9,123

General and administrative expenses consist of salaries and other related costs for personnel in executive, finance, legal, information technology and human resource functions. Other costs included in general and administrative expenses are professional fees for information technology services, external legal fees, consulting and accounting services as well as bad debt expense, certain facility and insurance costs, including director and officer liability insurance.

Total general and administrative expenses increased \$0.4 million, or 4.3%, to \$9.5 million in the three months ended March 31, 2016, as compared to \$9.1 million in the three months ended March 31, 2015. In the U.S. segment, general and administrative expense increased \$0.4 million, or 4.7%, to \$9.2 million in the three months ended March 31, 2016, as compared to \$8.7 million in the prior year period. In the International segment, general and administrative expense remained consistent.

The increase in the U.S. segment general and administrative expenses for the three months ended March 31, 2016 over the prior year period is primarily due to higher software amortization expense, increased severance and related expenses, increased insurance associated with being an equity public company and an increase in our provision for bad debt. These increases were partially offset by decreases in facility costs and by costs incurred in the prior year period not incurred in the current year period, including a \$0.3 million write-off of deferred initial public offering costs and \$0.3 million Avista sponsor fees.

Research and Development

		Months Aarch 31,
(dollars in thousands)	2016	2015
United States	\$2,867	\$6,015
International	169	181
Total Research and Development	\$3,036	\$6,196

Research and development expenses relate primarily to the development of new products to add to our portfolio and costs related to its medical affairs, medical information and regulatory functions. We do not allocate research and development expenses incurred in the United States to our International segment.

Total research and development expenses decreased \$3.2 million, or 51.0%, to \$3.0 million in the three months ended March 31, 2016, as compared to \$6.2 million in the three months ended March 31, 2015. In the U.S. segment, research and development expense decreased \$3.1 million, or 52.3%, to \$2.9 million in the three months ended March 31, 2016, as compared to \$6.0 million in the prior year period. In the International segment, research and development expense remained.

The decrease in the U.S. segment research and development expenses for the three months ended March 31, 2016 over the prior year period is primarily due to a reduction in depreciation expense as a result of the scheduled decommissioning of certain long-lived assets associated with research and development operations, partially offset by higher pharmacovigilance expenses due to the transition to a new vendor.

Gain on Sale of Assets

Effective January 7, 2016, our Canadian subsidiary entered into an asset purchase agreement, pursuant to which it would sell substantially all of the assets of our Canadian radiopharmacies and Gludef manufacturing and distribution business to one of our existing Canadian radiopharmacy customers. The purchase price for the asset sale was \$9.0 million in cash, which resulted in a pre-tax book gain of \$5.8 million, which was recorded within operating income in the quarter ended March 31, 2016.

Other Expense, Net

		Three Months Ended March 31,	
(dollars in thousands)	2016	2015	
Interest expense	\$(7,024)	\$(10,630)	
Interest income	6	7	
Other income (expense), net	58	(383)	
Total other expense, net	\$(6,960)	\$(11,006)	

Interest Expense

For the three months ended March 31, 2016, compared to the same period in 2015, interest expense decreased by \$3.6 million as a result of the June 2015 refinancing of our long-term debt with a lower interest rate.

Interest Income

For the three months ended March 31, 2016, compared to the same period in 2015, interest income remained consistent.

Other Income (Expense), net

For the three months ended March 31, 2016, compared to the same period in 2015, other income increased by \$0.4 million primarily due to \$0.3 million increase in tax indemnification income and a \$0.1 million decrease in foreign currency losses.

Provision (Benefit) for Income Taxes

	Three M	Months
	Ended M	arch 31,
(dollars in thousands)	2016	2015
Provision (benefit) for income taxes	\$ 390	\$ (3)

For the three months ended March 31, 2016 and 2015, our effective tax rate was 3.6% and 0.8%, respectively. The \$0.4 million increase in the tax provision for the three months ended March 31, 2016, as compared to the same period in 2015, was impacted primarily by changes in uncertain tax positions. Considering our history of losses, we continue to maintain a valuation allowance against substantially all of our net deferred tax assets and therefore our provision (benefit) for income taxes results primarily from taxes due in certain foreign jurisdictions where we generate taxable income, as well as interest and penalties associated with uncertain tax positions offset by reversals of uncertain tax positions as statutes lapse or are settled during the year.

Liquidity and Capital Resources

Cash Flows

The following table provides information regarding our cash flows:

	Three Months Ended March 31,		
(dollars in thousands)	2016	2015	\$ Change
Cash provided by (used in):			
Operating activities	\$3,780	\$15,157	\$(11,377)
Investing activities	\$7,422	\$ (3,498)	\$ 10,920
Financing activities	\$ (944)	\$ (459)	\$ (485)

Net Cash Provided by Operating Activities

Cash provided by operating activities is primarily driven by our earnings and changes in working capital. The \$11.4 million decrease in cash provided by operating activities for the three months ended March 31, 2016 as compared to 2015 was primarily driven by a \$12.1 million increase in cash used for net working capital requirements. The \$12.1 million increase in cash used for net working capital requirements was due to the timing of interest payments on long-term debt which resulted in a \$9.7 million decrease in net working capital as compared to the prior year period. In addition, the increase was further driven by increases in accounts receivable as a result of increases in certain major customer balances. These increases in cash used for net working capital requirements were partially offset by an increase in accounts payable as a result of the timing of payments.

Net Cash Provided by (Used in) Investing Activities

The increase in net cash used in investing activities in the three months ended March 31, 2016 as compared to 2015 primarily reflects the \$9.0 million gross proceeds from the sale of assets and a \$1.8 million decrease in capital expenditures.



Net Cash Used in Financing Activities

Net cash used in financing activities during the three months ended March 31, 2016 was primarily for our quarterly Term Facility payment.

Net cash used in financing activities during the three months ended March 31, 2015 was due to payments of audit and legal fees in connection with the IPO.

Historically, our primary source of cash flows from financing activities is draws against our outstanding Revolving Facility. Going forward, we expect our primary source of cash flows from financing activities to be similar draws against our Revolving Facility. Our primary historical uses of cash in financing activities are principal payments on our term loan and Revolving Facility as well as dividends to our shareholders. See "—External Sources of Liquidity."

External Sources of Liquidity

On June 30, 2015, we completed our initial public offering, entered into a new \$365.0 million seven-year Term Facility and amended and restated our Revolving Facility that has a borrowing capacity of \$50.0 million. The net proceeds of the Term Facility and the initial public offering together with available cash were used to repay in full the aggregate principal amount of the \$400.0 million Notes, and pay related premiums, interest and expenses and pay down \$8.0 million of borrowings under the Revolving Facility.

We have the right to request an increase of the Term Facility in an aggregate amount up to \$37.5 million plus additional amounts subject to certain leverage ratios. The term loans under the Term Facility bear interest, with pricing based from time to time at our election at (i) LIBOR plus a spread of 6.00% (with a LIBOR rate floor of 1.00%) or (ii) the Base Rate (as defined in our Term Facility) plus a spread of 5.00%. Interest under term loans based on (i) the LIBOR rate is payable at the end of each interest period (as defined in our Term Facility) and (ii) the Base Rate is payable at the end of each quarter. At March 31, 2016, our interest rate under the Term Facility was 7.00%. Our Term Facility is guaranteed by the Lantheus Holdings and Lantheus Real Estate, and obligations under the Term Facility are secured by substantially all the property and assets and all interests of Lantheus Holdings, LMI and Lantheus Real Estate.

Our Term Facility contains a number of affirmative, negative, reporting and financial covenants, in each case subject to certain exceptions and materiality thresholds. Incremental borrowings under the Revolving Facility may affect our ability to comply with the covenants in the Term Facility, including the financial covenant restricting total net leverage, accordingly, we may be limited in utilizing our net Borrowing Base availability as a source of liquidity. Our Term Facility requires us to be in quarterly compliance, measured on a trailing four quarter basis. The financial covenants are displayed in the table below:

Term Facility Financial Covenants

Period	Total Net Leverage Ratio
Q3 2015 to Q1 2016	6.25 to 1.00
Q2 2016 to Q4 2016	6.00 to 1.00
Q1 2017 to Q2 2017	5.50 to 1.00
Thereafter	5.00 to 1.00

The Term Facility contains usual and customary restrictions on the ability of us and our subsidiaries to: (i) incur additional indebtedness (ii) create liens; (iii) consolidate, merge, sell or otherwise dispose of all or substantially all of our assets; (iv) sell certain assets; (v) pay dividends on, repurchase or make distributions in respect of capital stock or make other restricted payments; (vi) make certain investments; (vii) repay subordinated indebtedness prior to stated maturity; and (viii) enter into certain transactions with our affiliates.

As of March 31, 2016, we had an unfunded Standby Letter of Credit of \$8.8 million. The unfunded Standby Letter of Credit requires annual fees, payable quarterly, which, subsequent to the amendment, is set at LIBOR plus a spread of 2.00% and expires in February 2017. It automatically renewed for a one year period and will continue to automatically renew for a one year period at each anniversary date, unless we elect not to renew in writing within 60 days prior to such expiration.

Our Revolving Facility is secured by a pledge of substantially all of our assets, including accounts receivable, inventory and machinery and equipment, and is guaranteed by each of Lantheus Holdings and Lantheus Real Estate. Borrowing capacity is determined by reference to a borrowing base, or the Borrowing Base, which is based on (i) a percentage of certain eligible accounts receivable, inventory and machinery and equipment minus (ii) any reserves. As of March 31, 2016, the aggregate Borrowing Base was approximately \$48.3 million, which was reduced by an outstanding \$8.8 million unfunded Standby Letter of Credit and \$0.1 million in accrued interest, resulting in a net borrowing base availability of approximately \$39.4 million.

The loans under our Revolving Facility bear interest with pricing based from time to time at our election at (i) LIBOR plus a spread of 2.00% or (ii) the Reference Rate (as defined in our Revolving Facility) plus a spread of 1.00%. Our Revolving Facility also includes an unused line fee of 0.375% and expires on June 30, 2020.

Our Revolving Facility contains affirmative and negative covenants, as well as restrictions on the ability of LMI, us and our subsidiaries to: (i) incur additional indebtedness or issue preferred stock; (ii) repay subordinated indebtedness prior to its stated maturity; (iii) pay dividends on, repurchase or make distributions in respect of capital stock or make other restricted payments; (iv) make certain investments; (v) sell certain assets; (vi) create liens; (vii) consolidate, merge, sell or otherwise dispose of all or substantially all of our assets; and (viii) enter into certain transactions with our affiliates. Our Revolving Facility also contains customary default provisions as well as cash dominion provisions which allow the lender to sweep our accounts during the period (x) certain specified events of default are continuing under our Revolving Facility or (y) excess availability under our Revolving Facility falls below (i) the greater of \$7.5 million or 15% of the then-current line cap (as defined in the Revolving Facility) for a period of more than five consecutive Business Days or (ii) \$5.0 million. During a covenant trigger period, we are required to comply with a consolidated fixed charge coverage ratio of not less than 1:00: 1:00. The fixed charge coverage ratio is calculated on a consolidated basis for us for a trailing four-fiscal quarter period basis, as (i) EBITDA (as defined in the agreement) minus capital expenditures minus certain restricted payments divided by (ii) interest plus taxes paid or payable in cash plus certain restricted payments paid or payable in cash.

Our ability to fund our future capital needs will be affected by our ability to continue to generate cash from operations and may be affected by our ability to access the capital markets, money markets, or other sources of funding, as well as the capacity and terms of our financing arrangements.

We may from time to time repurchase or otherwise retire our debt and take other steps to reduce our debt or otherwise improve our balance sheet. These actions may include open market repurchases of any notes outstanding, prepayments of our term loans or other retirements or refinancing of outstanding debt, privately negotiated transactions or otherwise. The amount of debt that may be repurchased or otherwise retired, if any, would be decided at the sole discretion of our Board of Directors and will depend on market conditions, trading levels of our debt from time to time, our cash position and other considerations.

Funding Requirements

Our future capital requirements will depend on many factors, including:

- our ability to have product manufactured and released from JHS and other manufacturing sites in a timely manner in the future;
- the pricing environment and the level of product sales of our currently marketed products, particularly DEFINITY and any additional products that we may market in the future;
- revenue mix shifts and associated volume and selling price changes that could result from contractual status changes with key customers and additional competition;
- the costs of further commercialization of our existing products, particularly in international markets, including product marketing, sales and distribution and whether we obtain local partners to help share such commercialization costs;
- · the costs of investing in our facilities, equipment and technology infrastructure;
- the costs and timing of establishing manufacturing and supply arrangements for commercial supplies of our products;
- the extent to which we acquire or invest in products, businesses and technologies;
- the extent to which we choose to establish collaboration, co- promotion, distribution or other similar arrangements for our marketed products;
- the legal costs relating to maintaining, expanding and enforcing our intellectual property portfolio, pursuing insurance or other claims and defending against product liability, regulatory compliance or other claims; and
- the cost of interest on any additional borrowings which we may incur under our financing arrangements.

Until we successfully become dual sourced for our principal products, we are vulnerable to future supply shortages. Disruption in the financial performance could also occur if we experience significant adverse changes in customer mix, broad economic downtums, adverse industry or company conditions or catastrophic external events. If we experience one or more of these events in the future, we may be required to implement additional expense reductions, such as a delay or elimination of discretionary spending in all functional areas, as well as scaling back select operating and strategic initiatives.

If our capital resources become insufficient to meet our future capital requirements, we would need to finance our cash needs through public or private equity offerings, assets securitizations, debt financings, sale-leasebacks or other financing or strategic alternatives, to the extent such transactions are permissible under the covenants of the agreements governing our senior secured credit facilities. Additional equity or debt financing, or other transactions, may not be available on acceptable terms, if at all. If any of these transactions require an amendment or waiver under the covenants in the agreements governing our senior secured credit facilities. Additional equity or debt financing, we will seek to obtain such a waiver to remain in compliance with those covenants. However, we cannot be assured that such an amendment or waiver would be granted, or that additional capital will be available on acceptable terms, if at all.

At March 31, 2016, our only current committed external source of funds is our borrowing availability under our Revolving Facility. We had \$38.9 million of cash and cash equivalents at March 31, 2016. Availability under our Revolving Facility is calculated by reference to the Borrowing Base. If we are not successful in achieving our forecasted results, our accounts receivable and inventory could be negatively affected, reducing the Borrowing Base and limiting our borrowing availability. Our new Term Facility contains a number of affirmative, negative, reporting and financial covenants, in each case subject to certain exceptions and materiality thresholds. Incremental borrowings under the Revolving Facility may affect our ability to comply with the covenants in the Term Facility, including the financial covenant restricting total net leverage. Accordingly, we may be limited in utilizing our net Borrowing Base availability as a source of liquidity.

Based on our current operating plans, we believe that our existing cash and cash equivalents, results of operations and availability under our Revolving Facility will be sufficient to continue to fund our liquidity requirements for at least the next twelve months.

Critical Accounting Policies and Estimates

The discussion and analysis of our financial position and results of operations is based on our condensed consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these condensed consolidated financial statements in accordance with U.S. GAAP requires us to make estimates and judgments that may affect the reported amounts of assets, liabilities, revenues and expenses, and related disclosure of contingent assets and liabilities. On an ongoing basis, we evaluate our estimates and judgments, including those related to revenue recognition and related allowances, inventory, impairments of long-lived assets including intangible assets, impairments of goodwill, income taxes including the valuation allowance for deferred tax assets. Actual results may differ materially from these estimates under different assumptions and conditions. In addition, our reported financial condition and results of operations could vary due to a change in the application of a particular accounting standard.

There have been no material changes to our critical accounting policies or in the underlying accounting assumptions and estimates used in such policies in the three months ended March 31, 2016. For further information, refer to our summary of significant accounting policies and estimates in our annual report on Form 10-K filed for the fiscal year ended December 31, 2015.

Off-Balance Sheet Arrangements

We are required to provide the U.S. Nuclear Regulatory Commission, or NRC, and Massachusetts Department of Public Health financial assurance demonstrating our ability to fund the decommissioning of our North Billerica, Massachusetts production facility upon closure, though we do not intend to close the facility. We have provided this financial assurance in the form of a \$28.2 million surety bond and an \$8.8 million letter of credit.

Since inception, we have not engaged in any other off-balance sheet arrangements, including structured finance, special purpose entities or variable interest entities.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We are exposed to market risk from changes in interest rates and foreign currency exchange rates. We do not hold or issue financial instruments to reduce these risks or for trading purposes.

Interest Rate Risk

As a result of our new Term Facility, we have substantial variable rate debt. Fluctuations in interest rates may affect our business, financial condition, results of operations and cash flows. As of March 31, 2016, we had \$362.3 million in principal outstanding under our Term Facility with a variable interest rate that only varies to the extent LIBOR exceeds one percent.

Furthermore, we are subject to interest rate risk in connection with the Revolving Facility, which is variable rate indebtedness. Interest rate changes could increase the amount of our interest payments and thus negatively impact our future earnings and cash flows. As of March 31, 2016, there was an \$8.8 million unfunded Standby Letter of Credit and \$0.1 million accrued interest, which reduced availability to \$39.4 million on the Revolving Facility. Any increase in the interest rate under the Revolving Facility may have a negative impact on our future earnings to the extent we have outstanding borrowings under the Revolving Facility. The effect of a 100 basis points adverse change in market interest rates, in excess of minimum floors, on our interest expense would be approximately \$0.9 million in the quarter ended March 31, 2016.

Historically, we have not used derivative financial instruments or other financial instruments to hedge such economic exposures.

Foreign Currency Risk

We face exposure to movements in foreign currency exchange rates whenever we, or any of our subsidiaries, enter into transactions with third parties that are denominated in currencies other than ours, or that subsidiary's, functional currency. Intercompany transactions between entities that use different functional currencies also expose us to foreign currency risk.

During three months ended March 31, 2016 and 2015, the net impact of foreign currency changes on transactions was a loss of \$0.2 million and \$0.4 million, respectively. Historically, we have not used derivative financial instruments or other financial instruments to hedge these economic exposures.

A portion of our earnings is generated by our foreign subsidiaries, whose functional currencies are other than the U.S. Dollar. Our earnings could be materially impacted by movements in foreign currency exchange rates upon the translation of the earnings of those subsidiaries into the U.S. Dollar. The Canadian Dollar presents the primary currency risk on our earnings. The cost of goods for our products that are manufactured in the United States and are sold in currencies other than the U.S. Dollar by our foreign subsidiaries are also affected by foreign currency exchange rate movements. Our cost of goods would have increased by \$0.5 million if the U.S. Dollar had been stronger by 10% when compared to the actual rates used during the three months ended March 31, 2016.

If the U.S. Dollar had been uniformly stronger by 10%, compared to the actual average exchange rates, our revenues would have decreased by \$0.7 million and our net income would have decreased by \$1.0 million for the three months ended March 31, 2016.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our principal executive officer and principal financial officer, we conducted an evaluation of our disclosure controls and procedures, as such term is defined under Rule 13a-15(e) or 15d-15(e) promulgated under the Exchange Act. Based on this evaluation, our principal executive officer and our principal financial officer concluded that our disclosure controls and procedures were effective as of the end of the period covered by this report.

Changes in Internal Control Over Financial Reporting

There have been no changes during the quarter ended March 31, 2016 in our internal control over financial reporting (as defined in Rule 13a-15(f) promulgated under the Exchange Act) that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

From time to time, the Company is a party to various legal proceedings arising in the ordinary course of business. In addition, the Company has in the past been, and may in the future be, subject to investigations by governmental and regulatory authorities, which expose it to greater risks associated with litigation, regulatory or other proceedings, as a result of which the Company could be required to pay significant fines or penalties. The outcome of litigation, regulatory or other proceedings cannot be predicted with certainty, and some lawsuits, claims, actions or proceedings may be disposed of unfavorably to the Company. In addition, intellectual property disputes often have a risk of injunctive relief which, if imposed against the Company, could materially and adversely affect its financial condition or results of operations.

On December 16, 2010, LMI filed suit against one of its insurance carriers seeking to recover business interruption losses associated with the NRU reactor shutdown and the ensuing global Moly supply shortage. The claim is the result of the shutdown of the NRU reactor in Chalk River, Ontario. The NRU reactor was off-line from May 2009 until August 2010. The defendant answered the complaint on January 21, 2011, denying substantially all of the allegations, presenting certain defenses and requesting dismissal of the case with costs and disbursements. Discovery, including international discovery and related motion practice, has been on-going for more than three years. The defendant filed a motion for summary judgment on July 14, 2014. The Company filed a memorandum of law in opposition to defendant's motion for summary judgment on August 25, 2014. The defendant filed a reply memorandum of law in three support of its motion for summary judgment on September 15, 2014. Expert witness discovery was completed on October 31, 2014. On March 25, 2015, the United States District Court for the Southern District of New York granted defendant's motion for summary judgment. On September 4, 2015, the Company filed an appeal of the District Court decision with the United States Court of Appeals for the Second Circuit. On December 4, 2015, the United States Court of Appeals for the Second Circuit heard oral arguments of the Company and defendant is answer. On April 21, 2016, the United States Court of Appeals for the Second Circuit heard oral arguments of the Company and defendant in connection with the Company's appeal. The Company and defendant in connection with the Company's appeal. The Company cannot be certain what amount, if any, or when, if ever, it will be able to recover for business interruption losses related to this matter.

Except as noted above, as of March 31, 2016, the Company had no material ongoing litigation in which the Company was a defendant or any material ongoing regulatory or other proceedings and had no knowledge of any investigations by government or regulatory authorities in which the Company is a target that could have a material adverse effect on its current business.

Item 1A. Risk Factors

There have been no material changes in the risk factors set forth in our Form 10-K for the fiscal year ended December 31, 2015 except as set forth below. For further information, refer to "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the fiscal year ended December 31, 2015.

We face potential supply and demand challenges for Xenon.

Currently, Nordion is our sole supplier, and we believe the principal supplier on a global basis, of Xenon, which is captured by Nordion as a by-product of the Moly production process. In January 2015, we entered into a new strategic agreement with IRE for the future supply of Xenon. Under the terms of the agreement, IRE will provide bulk Xenon to us for processing and finishing once development work has been completed and all necessary regulatory approvals have been obtained. We currently estimate commercial production will occur in 2016. If we are not able to begin providing commercial quantities of Xenon prior to the NRU reactor's transition in October 2016 from providing regular supply of medical isotopes to providing only emergency back-up supply of medical isotopes through March 2018, there may be a period of time during which we are not able to offer Xenon in our portfolio of commercial products, which would have a negative effect on our business, results of operations, financial condition and cash flows. For the year ended December 31, 2015, Xenon represented approximately 17% of our revenues.

Currently, we obtain Xenon from Nordion on a purchase order basis. If we are not able to pass along to our customers any change of terms from our supplier, there could be a negative effect on our business, results of operations, financial condition and cash flows.

Historically, several companies, including Mallinckrodt, sold packaged Xenon as a pulmonary imaging agent in the U.S., but starting in 2010 we have been the only supplier of this imaging agent in the U.S. In March 2016, Mallinckrodt received regulatory approval from the FDA to again sell packaged Xenon in the U.S. and has begun to do so. Depending upon the pricing, extent of availability and market penetration of Mallinckrodt's offering, we believe we are at risk for volume loss and price erosion from those customers that are not subject to price or volume commitments with us.

In addition to Mallinckrodt again selling packaged Xenon in the U.S., if there is an increase in the use of other imaging modalities in place of packaged Xenon, our current sales volumes would decrease, which could have a negative effect on our business, results of operations, financial condition and cash flows.

Xenon is frequently administered as part of a ventilation scan to evaluate pulmonary function prior to a perfusion scan with microaggregated albumin, or MAA, a technetium-based radiopharmaceutical used to evaluate blood flow to the lungs. Currently, Draxis is the sole supplier of MAA on a global basis. In 2014, Draxis announced substantial price increases for MAA. The increased price of MAA, or difficulties in obtaining MAA, could decrease the frequency in which MAA is used for lung perfusion evaluation, in turn, decreasing the frequency that Xenon is used for pulmonary function evaluation, resulting in a negative effect on our business, results of operations, financial condition and cash flows.

We face significant competition in our business and may not be able to compete effectively.

The market for diagnostic medical imaging agents is highly competitive and continually evolving. Our principal competitors in existing diagnostic modalities include large, global companies with substantial financial, manufacturing, sales and marketing and logistics resources that are more diversified than ours, such as GE Healthcare, Bracco, Mallinckrodt, Bayer and Draxis, as well as other competitors. We cannot anticipate their actions in the same or competing diagnostic modalities, such as significant price reductions on products that are comparable to our own, development or introduction of new products that are more cost-effective or have superior performance than our current products, the introduction of generic versions when our proprietary products lose their patent protection or the new entry into a generic market in which we are already a participant. In addition, distributors of our products could attempt to shift end-users to competing diagnostic modalities and products. Our current or future products could be rendered obsolete or uneconomical as a result of these activities. Our failure to compete effectively could cause us to lose market share to our competitors and have a material adverse effect on our business, results of operations, financial condition and cash flows.

In October 2014, Bracco received FDA approval in the United States for its echocardiography agent, Lumason (known as SonoVue outside of the U.S.), which is already approved for sale in Europe and certain Asian markets, including China, Japan and Korea. Bracco now has one of three FDA-approved echocardiography contrast agents in the United States, together with GE Healthcare's Optison and our DEFINITY. If Bracco successfully commercializes Lumason in the United States without otherwise increasing the overall usage of ultrasound contrast agents, our current and future sales volume could suffer, which would have a material adverse effect on our business, results of operations, financial condition and cash flows.

Xenon for lung ventilation diagnosis is our third largest product by revenue. Historically, several companies, including Mallinckrodt, sold packaged Xenon as a pulmonary imaging agent in the U.S., but starting in 2010 we have been the only supplier of this imaging agent in the U.S. In March 2016, Mallinckrodt received regulatory approval from the FDA to again sell packaged Xenon in the U.S. and has begun to do so. Depending upon the pricing, extent of availability and market penetration of Mallinckrodt's offering, we believe we are at risk for volume loss and price erosion from those customers that are not subject to price or volume commitments with us.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds Unregistered Sales of Equity Securities

None.

Issuer Purchase of Equity Securities

None.

Item 4. Mine Safety Disclosures

Not applicable.

Item 6. Exhibits

EXHIBIT NUMBER	DESCRIPTION OF EXHIBITS	INCORPORATED BY REFERENCE			
		FORM	FILE NUMBER	EXHIBIT	FILING DATE
2.1*	Amended and Restated Asset Purchase Agreement, effective January 7, 2016, by and between Lantheus MI Canada, Inc. and Isologic Innovative Radiopharmaceuticals Ltd.				
10.1*	Employment Agreement, effective August 12, 2013, by and between Lantheus Medical Imaging, Inc. and John Crowley.				
10.2*	Amendment to Employment Agreement, effective June 22, 2015, by and between Lantheus Medical Imaging, Inc. and John Crowley.				
10.3*	Amendment to Employment Agreement, effective March 25, 2016, by and between Lantheus Medical Imaging, Inc. and John Crowley.				
31.1*	Certification of the Chief Executive Officer Pursuant to Rule 13a-14(a) or 15d- 14(a) of the Securities Exchange Act of 1934, as amended, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.				
31.2*	Certification of the Chief Financial Officer Pursuant to Rule 13a-14(a) or 15d- 14(a) of the Securities Exchange Act of 1934, as amended, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.				
32.1*	Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted by Section 906 of the Sarbanes-Oxley Act of 2002.				
101.INS*	XBRL Instance Document				
101.SCH*	XBRL Taxonomy Extension Schema Document				
101.CAL*	XBRL Taxonomy Extension Calculation Linkbase Document				
101.DEF*	XBRL Taxonomy Extension Definition Linkbase Document				
101.LAB*	XBRL Taxonomy Extension Label Linkbase Document				
101 DD E*	VPDI Tayonomy Extension Presentation Linkhase Document				

101.PRE* XBRL Taxonomy Extension Presentation Linkbase Document

* Filed herewith

Table of Contents

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

LANTHEUS HOLDINGS, INC.

By:/s/ MARY ANNE HEINOName:Mary Anne HeinoTitle:President and Chief Executive OfficerDate:May 3, 2016

LANTHEUS HOLDINGS, INC.

By:/s/ JOHN CROWLEYName:John CrowleyTitle:Chief Financial OfficerDate:May 3, 2016

35

Table of Contents

EXHIBIT INDEX

FILING DATE

EXHIBIT NUMBER	DESCRIPTION OF EXHIBITS	INCORPORATED BY REFERENCE		
		FORM	FILE NUMBER	EXHIBIT
2.1*	Amended and Restated Asset Purchase Agreement, effective January 7, 2016, by and between Lantheus MI Canada, Inc. and Isologic Innovative Radiopharmaceuticals Ltd.			
10.1*	Employment Agreement, effective August 12, 2013, by and between Lantheus Medical Imaging, Inc. and John Crowley.			
10.2*	Amendment to Employment Agreement, effective June 22, 2015, by and between Lantheus Medical Imaging, Inc. and John Crowley.			
10.3*	Amendment to Employment Agreement, effective March 25, 2016, by and between Lantheus Medical Imaging, Inc. and John Crowley.			
31.1*	Certification of the Chief Executive Officer Pursuant to Rule 13a-14(a) or 15d- 14(a) of the Securities Exchange Act of 1934, as amended, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.			
31.2*	Certification of the Chief Financial Officer Pursuant to Rule 13a-14(a) or 15d- 14(a) of the Securities Exchange Act of 1934, as amended, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.			
32.1*	Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted by Section 906 of the Sarbanes-Oxley Act of 2002.			
101.INS*	XBRL Instance Document			
101.SCH*	XBRL Taxonomy Extension Schema Document			
101.CAL*	XBRL Taxonomy Extension Calculation Linkbase Document			
101.DEF*	XBRL Taxonomy Extension Definition Linkbase Document			
101.LAB*	XBRL Taxonomy Extension Label Linkbase Document			
101.PRE*	XBRL Taxonomy Extension Presentation Linkbase Document			

* Filed herewith

36

AMENDED AND RESTATED ASSET PURCHASE AGREEMENT

January 13, 2016

This AMENDED AND RESTATED ASSET PURCHASE AGREEMENT is made by and between (i) Lantheus MI Canada, Inc., a Canadian corporation ("<u>Seller</u>"), on the one hand, and (ii) Isologic Innovative Radiopharmaceuticals Ltd., a Canadian corporation ("<u>Buyer</u>"), on the other hand, amends and restates in its entirety the original Asset Purchase Agreement by and between Seller and Buyer executed as of December 31, 2015 (the "<u>Execution Date</u>") and made effective only as of January 7, 2016 (this "<u>Agreement</u>").

Buyer and Seller also may be referred to in this Agreement each as a "<u>Party</u>" and collectively as the "<u>Parties</u>." All capitalized terms used in this Agreement are defined in Section 1.1 below.

RECITALS

WHEREAS, as of the Execution Date, Seller conducts its business of Seller (i) owning and operating three (3) radiopharmacies in Quebec City, Quebec, Canada, Dorval, Quebec, Canada and Mississagua, Ontario, Canada that prepare individual, patient-ready unit doses of SPECT-based radiopharmaceuticals (including bulk unit doses and iodine-based products) for sale to healthcare providers (for administration to patients by those healthcare providers) in the Territory, (ii) operating a radiopharmacy in Vancouver, British Columbia, Canada that prepares individual, patient-ready unit doses of SPECT-based radiopharmaceuticals (including bulk unit doses and iodine-based products) for sale to healthcare providers (for administration to patients by those healthcare providers) in the Territory, (iii) owning and operating a direct business involved in selling the specific third party, SPECT-based cold kits set forth on Schedule 1.1(a) to healthcare providers as set forth on Schedule 1.1(a) (for those healthcare providers to prepare unit doses for administration to patients by those healthcare providers) in the Territory and (iv) operating a business involved in having Gludef® (Fludeoxyglucose F18 Injection) manufactured by third parties and selling individual, patient-ready unit doses of Gludef® (Fludeoxyglucose F18 Injection) manufactured by third parties and selling individual, patient-ready unit doses of Gludef® (Fludeoxyglucose F18 Injection) to healthcare providers) in the Territory (the businesses described in clauses (i) through (iv), collectively, the "<u>Radiopharmacey Business</u>"), as well as (v) owning and/or operating entertors, hot products and cold kits other than those cold kits referred to in clause (ii) above) to healthcare providers (or those healthcare providers to prepare unit doses for administration to patients and bulk SPECT-based radiopharmaceuticals (i.e., technetium generators, hot products and cold kits other than those cold kits referred to in clause (ii) above) to healthcare providers (or those healt

WHEREAS, Seller is willing to transfer to Buyer all rights in, and certain assets and liabilities relating to, the Radiopharmacy Business on the terms and conditions set forth in this Agreement; and

Page 1 of 59

WHEREAS, the Parties desire that Seller sell, transfer and assign to Buyer, and Buyer acquire and assume, all of the Purchased Assets and Assumed Liabilities, subject to the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual promises in this Agreement, and in consideration of the representations, warranties, and covenants in this Agreement, the Parties agree as follows:

ARTICLE 1

DEFINITIONS; INTERPRETATION

1.1 Definitions. The following terms will have the following meanings for purposes of this Agreement:

(a) "Accounts Payable" means all current accounts payable, regardless of when asserted, billed or imposed, of Seller (other than Excluded Liabilities) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles.

(b) "Accounts Receivable" means all accounts receivable and other receivables (including trade accounts, refunds and rebates owing to the Seller) of the Seller (other than Excluded Assets) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles.

(c) "<u>Accrued Expenses</u>" means all current accrued expenses of Seller (other than Excluded Liabilities and accrued expenses relating to employees) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles.

(d) "<u>Accrued Rebates</u>" means all accrued rebates, grants to customers and sales rebates of Seller (other than Excluded Liabilities) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles.

(e) "<u>Affiliate</u>" as applied to any Person, means any other Person directly or indirectly controlling, controlled by, or under common control with, that Person. For the purposes of this definition, "<u>control</u>" (including, with correlative meanings, the terms "<u>controlling</u>," "<u>controlled by</u>" and "<u>under common</u> <u>control with</u>"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities or by Contract or otherwise. Notwithstanding the foregoing, (i) neither Avista Capital, nor its personnel, nor its investment funds, nor any portfolio company of any of its investment funds (other than Lantheus Holdings, Inc. and its Subsidiaries) will be considered to be an Affiliate of Seller, and (ii) neither DW Healthcare Partners, nor its personnel, nor its investment funds (other than Isologic Innovative Radiopharmaceuticals Ltd. and its Subsidiaries) will be considered to be an Affiliate of Buyer.

Page 2 of 59

(f) "<u>Agreement</u>" has the meaning set forth in the Preamble.

(g) "Allocation Schedule" has the meaning set forth in Section 3.3.

(h) "Applicable Accounting Principles" has the meaning set forth on Schedule 1.1(h).

(i) "<u>Approvals</u>" means any approvals, permits, licenses or similar approvals or rights issued by Governmental Authorities (including Health Canada and the CNSC) that are required to be obtained or maintained under any applicable Law, in each case, in connection with the Radiopharmacy Business or Purchased Assets.

(j) "Assigned Approvals" has the meaning set forth in Section 2.1(e).

- (k) "Assigned Contracts" has the meaning set forth in Section 2.1(c).
- (1) "Assigned Intellectual Property" has the meaning set forth in Section 2.1(d).
- (m) "<u>Assigned Real Property Leases</u>" has the meaning set forth in Section 2.1(b).
- (n) "Assumed Liabilities" has the meaning set forth in Section 2.3.
- (o) "Basket Amount" has the meaning set forth in Section 7.3(b).
- (p) "Benefit Plans Transition Date" has the meaning set forth in Section 6.5(j).
- (q) "Bill of Sale" means a Bill of Sale in substantially the form of Exhibit A to this Agreement.
- (r) "Bulk SPECT Products" has the meaning set forth in Section 6.8(a)(I).

(s) "<u>Business Day</u>" means, any day other than (i) a Saturday or a Sunday or (ii) a day on which banks are authorized to close in New York, New York in the United States of America or in Toronto, Ontario, Canada.

(t) "<u>Buyer</u>" has the meaning set forth in the Preamble.

(u) "Buyer Benefit Plans" has the meaning set forth in Section 6.5(h).

(v) "Buyer Customers" has the meaning set forth in Section 6.8(b).

(w) "Buyer's Indemnified Persons" has the meaning set forth in Section 7.1.

(x) "<u>Canadian Nuclear Safety Commission</u>" or "<u>CNSC</u>" means the Governmental Authority of Canada with responsibility regulating nuclear energy, facilities and materials, and any Governmental Authority successor thereto.

Page 3 of 59

(y) "<u>Cap</u>" has the meaning set forth in Section 7.3(b).

(z) "Closing" has the meaning set forth in Section 3.5.

(aa) "<u>Closing Date</u>" has the meaning set forth in Section 3.5.

(bb) "Closing Time" has the meaning set forth in Section 3.5.

(cc) "<u>Closing Working Capital</u>" means: (a) Current Assets, minus (b) Current Liabilities, in each case, at the Closing Time and determined using the Applicable Accounting Principles.

(dd) "Closing Working Capital Statement" has the meaning set forth in Section 3.2(a)(i)(A).

(ee) "Competing Business" has the meaning set forth in Sections 6.8(a)(i)(A) and (B).

(ff) "Competition Act" means the Competition Act (Canada), as amended, and includes the regulations promulgated thereunder.

(gg) "Competition Tribunal" means the Competition Tribunal established by subsection 3(1) of the Competition Tribunal Act (Canada).

(hh) "Confidential Information" has the meaning set forth in Sections 9.1(a) and 9.1(b).

(ii) "<u>Contract</u>" means any contract, agreement, lease, license, commitment, indenture, mortgage, note, bond loan or other arrangement, understanding, undertaking, commitment or obligation, whether written or oral.

(jj) "Contracting Parties" has the meaning set forth in Section 10.14.

(kk) "Covered Employees" has the meaning set forth in Section 6.5(j).

(II) "<u>Current Assets</u>" means the current assets of Seller (other than Excluded Assets) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles.

(mm) "<u>Current Liabilities</u>" means the current liabilities of Seller (other than Excluded Liabilities) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles, and including the amount of vacation days in respect of the Transferred Employees assumed by Buyer calculated as of the day before the Closing Date.

(nn) "DEL" means a drug establishment license issued by Health Canada.

(oo) "Disclosing Party" has the meaning set forth in Section 9.1(a).

Page 4 of 59

(pp) "Disputed Amounts" has the meaning set forth in Section 3.2(c).

(qq) "<u>Employee on Leave</u>" means the Ontario Employees who do not report for work on the Closing Date because they are on leave (for the avoidance of doubt, an Employee on vacation or absent due to a temporary sickness on the Closing Date is not considered an Employee on Leave).

(rr) "<u>Employees</u>" means the individuals employed by the Seller in the Radiopharmacy Business as of the Closing Date and set forth on Schedule 4.14(a).

(ss) "Employee Benefit Plan" means, with respect to any Person, each material plan, fund, program, agreement, arrangement or scheme which (i) is sponsored, maintained or required to be sponsored or maintained by such Person or to which such Person makes or has made, or has or has had an obligation to make, contributions at any time and (ii) provides benefits to the current and former employees or the dependents of any of them (whether written or oral), including (A) each deferred compensation, bonus, incentive compensation, pension, retirement, employee stock ownership, stock purchase, stock option, profit sharing or deferred profit sharing, stock appreciation, stock option plan and other equity compensation plan, or welfare plan, (B) each retirement, pension or supplemental pension plan, (C) each severance plan or agreement, and each other plan providing health, vacation, supplemental employment benefit, hospitalization insurance, medical, dental, disability, sick pay, life insurance, death or survivor benefits, fringe benefits or legal benefits and (D) each other employee benefit plan, fund, program, agreement or arrangement, in the case of each of clauses (A)-(D), to the extent material.

(tt) "Environmental Law" means any Law in any way relating to the environment or natural resources.

- (uu) "ETA" means the Excise Tax Act (Canada), as amended.
- (vv) "Excluded Assets" has the meaning set forth in Section 2.2.
- (ww) "Excluded Employees" has the meaning set forth in Section 6.5(c).
- (xx) "Excluded Liabilities" has the meaning set forth in Section 2.4.
- (yy) "Execution Date" has the meaning set forth in the Preamble.
- (zz) "FDG" means Fludeoxyglucose F18 Injection, including Gludef®.
- (aaa) "Financial Statements" has the meaning set forth in Section 4.4.
- (bbb) "Fundamental Representations" has the meaning set forth in Section 7.3(a).

(ccc) "General IP Assignment" means a general assignment of the Assigned Intellectual Property in substantially the form of Exhibit B to this Agreement.

Page 5 of 59

(ddd) "<u>Governmental Authority</u>" means any government or governmental or regulatory body thereof, or political subdivision thereof, whether domestic or foreign, federal, provincial or local, or any department (or subdivision thereof), commission, bureau, tribunal, agency, board, instrumentality, minister, governor-in-council or authority thereof, any court or arbitrator (public or private), or any applicable stock exchange.

(eee) "GST/HST" means the goods and services tax / harmonized sales tax imposed pursuant to the ETA.

(fff) "Health Canada" means the Governmental Authority of Canada with responsibility for national public health, and any Governmental Authority successor thereto.

(ggg) "<u>Indebtedness</u>" of any Person means the principal of and accreted value and unpaid interest in respect of: (i) indebtedness for borrowed money; (ii) amounts owing as the deferred purchase price for property or services; and (iii) indebtedness evidenced by any note, bond, debenture or other debt instrument or debt security, the payment of which such Person is responsible or liable. For greater clarity, Indebtedness will not include any amount included in the calculation of the Closing Working Capital.

(hhh) "Indemnification Claim" has the meaning set forth in Section 7.4(b).

(iii) "Indemnifying Party" has the meaning set forth in Section 7.4(b).

(jjj) "Indemnified Persons" means the Buyer's Indemnified Persons or the Seller's Indemnified Persons, as the case may be.

(kkk) "Independent Accountant" has the meaning set forth in Section 3.2(c).

(III) "Individual Bonus Amounts" means the aggregate amount of all outstanding annual cash bonus awards payable to each Transferred Employee and calculated in the Ordinary Course of Business, in respect of his or her service with the Seller for the 2015 calendar year, including all income tax, payroll tax and other similar amounts exigible on such cash bonus awards.

(mmm) "Intellectual Property" means all right, title and interest in or relating to intellectual property, whether protected, created or arising under the Laws of Canada or any other jurisdiction, including: (i) all patents and applications therefor, including all continuations, divisionals, and continuationsin-part thereof and patents issuing thereon, along with all reissues, reexaminations and extensions thereof; (ii) all trademarks, service marks, trade names, service names, brand names, trade dress rights, logos, corporate names, trade styles, logos and other source or business identifiers and general intangibles of a like nature, together with the goodwill associated with any of the foregoing, along with all applications, registrations, renewals and extensions thereof (collectively, "Marks"); (iii) all Internet domain names; (iv) all copyrights and all mask work, database and design rights, whether or not registered or published, all registrations and recordations thereof and all applications in connection therewith, along with all reversions, extensions and renewals thereof; (v) all trade secrets and all proprietary and confidential know-how, inventions, discoveries, concepts, ideas, research and development, manufacturing and production processes and techniques, technical data, procedures, designs, drawings, specifications, databases, and other proprietary and confidential information.

Page 6 of 59

(nnn) "<u>Intercompany Payables</u>" means all accounts payable, regardless of when asserted, billed or imposed, of Seller owed to any Affiliate of Seller to the extent relating to the Radiopharmacy Business or the Purchased Assets at the Closing Time.

(000) Reserved.

(ppp) "Interim Reference Date" means September 30, 2015.

(qqq) "<u>Inventory</u>" means all raw materials, work-in-process, supplies, finished goods and other inventories owned by Seller (other than Excluded Assets) and physically located at the Quebec City, Quebec, Dorval, Quebec, Mississagua, Ontario or Vancouver, British Columbia radiopharmacies (excluding the distribution center in Dorval, Quebec) at the Closing Time, the value of which will be determined using the Applicable Accounting Principles.

(rrr) "Key Employees" means the Employees set forth on Schedule 1.1(rrr).

(sss) "Law" means any law, statute, regulation, ordinance, rule, Order or requirement enacted, promulgated, entered into, or imposed by, any Governmental Authority (including, for the sake of clarity, common law).

(ttt) "Leased Real Property" has the meaning set forth in Section 4.11.

(uuu) "Legal Proceeding" means any judicial, administrative or arbitral actions, suits, proceedings, hearings or investigations (in each case, whether public or private and whether civil, criminal or administrative) by or before a Governmental Authority.

(vvv) "<u>Liability</u>" means, collectively, any Indebtedness, guaranties, endorsements, claims, losses, damages, deficiencies, costs, expenses, fines, penalties, liabilities, obligations or responsibilities of any nature or kind, whether accrued or fixed, known or unknown, absolute or contingent, matured or not or determined or determinable, and whether due or to become due, including any liability arising under any Law, action or Order and under liability arising under any Contract or undertaking.

(www) "Lien" means any lien, encumbrance, pledge, mortgage, deed of trust, security interest, claim, lease, charge, option, right of first refusal, easement, servitude, transfer restriction, hypothec, hypothecation, and security under Section 426 or 427 of the *Bank Act* (Canada).

(xxx) "Loss" and "Losses" have the meanings set forth in Section 7.1.

(yyy) "Marks" has the meaning set forth in the definition of the term Intellectual Property.

(zzz) "<u>Material Adverse Effect</u>" means any result, occurrence, fact, change, event or effect that is materially adverse to, or would reasonably be expected to be materially

Page 7 of 59

adverse to, the condition (financial or otherwise), assets, liabilities, obligations, operations or business of the Radiopharmacy Business and Purchased Assets, taken as a whole, except for any result, occurrence, fact, change, event or effect that arises out of, results from or is attributable to: (i) changes that are the result of factors generally affecting radiopharmacy and nuclear medicine department markets in the Territory; (ii) any adverse change, effect or circumstance arising out of or resulting from actions contemplated by the Parties in connection with this Agreement or the pendency or announcement of the Transactions (including any adverse change, effect or circumstance arising out of or resulting from any contact, discussions, requests or negotiations involving Buyer or its Representatives, on the one hand, and any counterparty to an Assigned Contract, on the other hand, including with respect to notices or consents to any of the Transactions); (iii) changes or proposed changes in applicable Laws, regulatory conditions, policies or government programs or the interpretation, application or non-application of applicable Laws, conditions, policies or programs by any Governmental Authority; (iv) any change to generally accepted accounting principles or in the interpretation thereof; (v) any failure, in and of itself, of Seller to meet any projection or forecast (it being understood that the causes underlying the failure may be taken into account in determining whether a Material Adverse Effect has occurred); or (vi) changes that are the result of described in clauses (i), (iii), (iv) or (vi) above will not apply to the extent that such result, occurrence, fact, change, event or effect has a material and disproportionate effect on the Radiopharmacy Business as whole, relative to radiopharmaces and/or nuclear medicine departments in the Territory.

(aaaa) "Non-Solicitation Period" has the meaning set forth in Section 6.8(b).

(bbbb) "Nonparty Affiliates" has the meaning set forth in Section 10.14.

(cccc) "Nuclear License" means a nuclear substance and radiation device License issued by the Canadian Nuclear Safety Commission (CNSC).

(dddd) "Offer Letters" has the meaning set forth in Section 6.5(a).

(eeee) "OHSA" has the meaning set forth in Section 4.14(f).

(ffff) "Order" means, except as otherwise provided in this Agreement, any final and enforceable order, injunction, judgment, decree, ruling, writ, assessment, award or arbitration award of a Governmental Authority.

(gggg) "Ordinary Course of Business" means the ordinary and usual course of normal day-to-day operations of the Radiopharmacy Business, as conducted by or on behalf of Seller as of immediately prior to the Closing Date and consistent in all material respects with past practice.

(hhhh) "Party" and "Parties" have the meanings set forth in the Preamble.

(iiii) "Payroll Schedule" has the meaning set forth in Section 6.5(g).

Page 8 of 59

(jjjj) "Permitted Liens" means:

(i) Liens for Taxes, assessments, Governmental Authority charges or levies not yet due and payable; or

(ii) statutory Liens relating to obligations not due and payable; or

(iii) Liens for public utilities not due and payable; or

(iv) mechanics', carriers', workmen's, repairmen's or other like Liens arising or incurred in the Ordinary Course of Business with respect to amounts that are not delinquent and which are not, individually or in the aggregate, material to the Radiopharmacy Business or the Purchased Assets, taken as a whole; or

(v) easements, rights of way, zoning ordinances and other similar encumbrances affecting Leased Real Property which are not, individually or in the aggregate, material to the Radiopharmacy Business or the Purchased Assets and which do not prohibit or interfere with the current operation of any Leased Real Property; or

(vi) liens arising under original purchase price conditional sales contracts and equipment leases with third parties entered into in the Ordinary Course of Business; or

(vii) any privilege in favor of any lessor, licensor or permitter for rent to become due or for other obligations or acts, the performance of which is required under Assigned Contracts; or

(viii) other imperfections of title or Liens which are not, individually or in the aggregate, material to the Radiopharmacy Business and the Purchased Assets, taken as a whole.

(kkkk) "<u>Person</u>" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization, any other business entity or a Governmental Authority.

(IIII) "Post-Closing Adjustment" has the meaning set forth in Section 3.2(a)(ii).

(mmmm) "<u>Prepaid Expenses</u>" means all deposits, advances, credits and prepaid expenses of Seller (other than Excluded Assets) at the Closing Time, in each case, to the extent relating to the Radiopharmacy Business or the Purchased Assets and determined using the Applicable Accounting Principles.

(nnnn) "Purchase Price" has the meaning set forth in Section 3.1.

(0000) "Purchased Assets" has the meaning set forth in Section 2.1.

Page 9 of 59

(pppp) "OST" means the Quebec sales tax imposed pursuant to the QSTA.

(qqqq) "OSTA" means the Act Respecting the Quebec Sales Tax, as amended.

(rrrr) "Quebec Employee" means an Employee employed in the Province of Quebec.

(ssss) "<u>Radiopharmacy Business</u>" has the meaning set forth in the Recitals.

(tttt) "Receiving Party" has the meaning set forth in Section 9.1(a).

(uuuu) "<u>Records</u>" means all files, reports, books, records, accounts, documents and similar materials owned by Seller, in each case, to the extent exclusively relating to the Radiopharmacy Business or the Transferred Employees, in whatever form, in each case, to the extent permitted by applicable Law to be transferred; <u>provided</u>, <u>however</u>, that "<u>Records</u>" will not include any files, reports, accounts, books, records, documents and similar materials (i) prepared in connection with the Transaction, (ii) which Seller is prohibited from disclosing or transferring to Buyer under applicable Law or is required by applicable Law to retain or (iii) maintained by Seller and/or its Representatives, agents, or licensees in connection with their respective ongoing financing, compliance, legal or Tax requirements (provided that any copies of any materials relating to regulatory matters or any Legal Proceedings (for the avoidance of doubt, including only those investigations to which Seller, to Seller's Actual Knowledge, was subject) initiated against the Seller in respect of the operations of the Radiopharmacy Business shall be included in "<u>Records</u>").

(vvvv) "<u>Reference Time</u>" means December 31, 2015.

(www) "<u>Representatives</u>" means, with respect to any Person, the Affiliates, directors, officers, employees, agents or advisors (including attorneys, accountants, financial advisors and consultants) of such Person and representatives of any of the foregoing; and, with respect to Buyer, the term "<u>Representatives</u>" will also include Buyer's lenders and their respective Representatives.

(xxxx) "Required Lease Consents" has the meaning set forth in Schedule 1.1(xxxx).

(yyyy) "Resolution Period" has the meaning set forth in Section 3.2(b)(ii).

(zzzz) "<u>Restricted Period</u>" has the meaning set forth in Section 6.8(a).

(aaaaa) "<u>Retained Business</u>" has the meaning set forth in the Preamble.

(bbbbb) "Review Period" has the meaning set forth in Section 3.2(b)(i).

(ccccc) "<u>Scheduled Approvals</u>" has the meaning set forth in Section 4.12(a).

(dddd) "Seller" has the meaning set forth in the Preamble.

Page 10 of 59

(eeeee) "Seller Benefit Plan" means each Employee Benefit Plan which provides benefits to Employees or former employees of the Radiopharmacy Business or their dependents.

(fffff) "Seller's Indemnified Persons" has the meaning set forth in Section 7.2.

(ggggg) "Seller's Actual Knowledge" means the actual knowledge, without any inquiry, of those individuals set forth on Schedule 1.1 (ggggg).

(hhhhh) "Seller's Knowledge" means the actual knowledge, after reasonable inquiry within the Seller and its Affiliates taking into account their position with the Seller and/or its Affiliates, of those individuals set forth on Schedule 1.1(ggggg).

(iiiii) "Statement of Objections" has the meaning set forth in Section 3.2(b)(ii).

(jjjjj) "Supply Agreement" means the Supply Agreement in substantially the form of Exhibit C to this Agreement, as may be amended, modified and/or supplemented from time to time in accordance with its terms.

(kkkkk) "Target Working Capital" has the meaning set forth in Section 3.2(a)(ii).

(IIII) "Tax" or "Taxes" means any and all taxes, assessments, levies, tariffs, duties or other charges or impositions in the nature of a tax imposed by any Governmental Authority, including, without limitation, income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, real property transfer, recording, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever (whether payable directly or by withholding and whether or not requiring the filing of a Tax Return), including any interest or penalty thereon or addition thereto and any interest in respect of such additions or penalties, and for greater certainty will also include all Canada Pension Plan and other pension plan premiums or contributions imposed by any Governmental Authority, employment insurance premiums, and provincial workers' compensation payments.

(mmmm) "Tax Act" means the Income Tax Act (Canada).

(nnnn) "<u>Tax Return</u>" means any report, return (including any information return), claim for refund, election, estimated Tax filing or payment, request for extension, document, declaration or other information or filing supplied or required to be supplied to any Governmental Authority with respect to Taxes.

(00000) "Territory" means Canada.

(ppppp) "Total Consideration" has the meaning set forth in Section 3.1.

(qqqqq) "<u>Transaction Documents</u>" means this Agreement, the Bill of Sale, the General IP Assignment, the Supply Agreement, the Transition Services Agreement and the other documents, instruments, exhibits, annexes, schedules or certificates contemplated by this Agreement and thereby.

Page 11 of 59

(rrrrr) "Transactions" means the transactions contemplated by the Transaction Documents.

(sssss) "Transfer Date" has the meaning set forth in Section 6.5(a).

(ttttt) "Transfer Taxes" has the meaning set forth in Section 8.2.

(uuuuu) "Transferred Employees" has the meaning set forth in Section 6.5(a).

(vvvvv) "Transition Period Benefit Plans" has the meaning set forth in Section 6.5(j).

(wwww) "<u>Transition Services Agreement</u>" means the transition services agreement in substantially the form of <u>Exhibit D</u> to this Agreement, as may be amended, modified and/or supplemented from time to time in accordance with its terms.

(xxxxx) "Undisputed Amounts" has the meaning set forth in Section 3.2(c).

1.2 Interpretation. References in this Agreement to any gender include references to all genders, and references to the singular include references to the plural and vice versa. The words "include," "includes" and "including" when used in this Agreement will be deemed to be followed by the phrase "without limitation." Unless the context otherwise requires, references in this Agreement to Articles, Sections, Exhibits and Schedules will be deemed references to Articles and Sections of, and Exhibits and Schedules to, this Agreement. The headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement. Any reference to any Law will be deemed to refer to the Law as amended and also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise.

ARTICLE 2

PURCHASE AND SALE OF ASSETS AND ASSUMPTION OF LIABILITIES

2.1 <u>Purchase and Sale of Assets</u>. On the terms and subject to the conditions set forth in this Agreement, on the Closing Date, for the Total Consideration, Seller will sell, transfer, assign, convey and deliver to Buyer, and Buyer will purchase, acquire and accept from Seller or its designee, all right, title and interest in, to and under the Purchased Assets, free and clear of all Liens (other than Permitted Liens). "<u>Purchased Assets</u>" means the following assets and rights owned by Seller at the Closing Time, in each case, to the extent pertaining to the Radiopharmacy Business:

(a) all Accounts Receivable, Inventory and Prepaid Expenses (including the benefit of the instrument described on Schedule 3.6(b)(iv) and including any security deposits in respect of any Assigned Real Property Leases);

Page 12 of 59

(b) all Real Property Leases set forth on Schedule 2.1(b) (collectively, the "Assigned Real Property Leases");

(c) all other Contracts (or portions thereof) set forth on Schedule 2.1(c) (collectively, the "Assigned Contracts");

(d) (i) any Intellectual Property (including all standard operating procedures, trade secrets and know-how) that is owned by the Seller at the Closing Time, in each case, solely to the extent (A) it is used exclusively in the conduct of the Radiopharmacy Business and (B) except with respect to Gludef® (i.e., FDG solely with respect to the Territory), it relates only to radiopharmacy preparation of patient-ready unit doses of radiopharmaceuticals to healthcare providers (for administration to patients by those healthcare providers) and is not otherwise Intellectual Property used to make or prepare, have made or prepared or otherwise used in any proprietary radiopharmaceutical of Seller, and (ii) the Gludef® Mark (collectively, the "Assigned Intellectual Property");

(e) all of Seller's rights in the Scheduled Approvals identified on Schedule 2.1(e) (the "Assigned Approvals");

(f) all furniture, fixtures, equipment, supplies and other tangible personal property owned by Seller and physically located at the Quebec City, Quebec, Dorval, Quebec, Mississagua, Ontario or Vancouver, British Columbia radiopharmacies (excluding the distribution center and data center in Dorval, Quebec);

(g) all of Seller's rights under warranties, indemnities and similar rights against third parties, in each case, to the extent related to any of the other Purchased Assets;

(h) originals (or, where not readily available, copies) of all Records (other than those included in the Excluded Assets); provided that Seller will be entitled to retain copies of all Records; and

(i) the goodwill associated with any of the assets described above, including lists of all distributors, customers and suppliers of the Radiopharmacy Business.

2.2 <u>Excluded Assets</u>. The Parties acknowledge and agree that Seller will not convey, transfer, deliver or assign to Buyer, and Buyer will not purchase, take delivery of, or acquire, any rights to any assets, properties, interests or rights of Seller other than the Purchased Assets specifically enumerated in Section 2.1 (collectively the "<u>Excluded Assets</u>"), including all of the rights, title and interests of Seller and its Affiliates in the following:

(a) all cash and cash equivalents (including restricted cash), bank accounts, bank balances, moneys in possession of banks, or other depositories, term deposits and securities (but the benefit of the instrument described on Schedule 3.6(b)(iv) and any security deposits in respect of any Assigned Real Property Leases are not Excluded Assets);

Page 13 of 59

(b) all Contracts (or portions thereof) that are not Assigned Contracts;

(c) all Intellectual Property other than the Assigned Intellectual Property;

(d) all items expressly excluded from the definition of the term Records, retained copies (or originals) of the Records and the corporate seals, organizational documents, minute books, stock record books, Tax Returns, books of account or other records having to do with the corporate organization of Seller or its Affiliates;

(e) all insurance policies and insurance proceeds and insurance awards receivable thereunder;

(f) all Approvals other than the Assigned Approvals;

(g) all Seller Benefit Plans and trusts or other assets attributable to those Seller Benefit Plans;

(h) all rights to any action, suit or claim of any nature available to, or being pursued by, Seller or any of its Affiliates, whether arising by counterclaim or otherwise, arising out of or relating to events prior to the Closing Date or to the extent pertaining to any other Excluded Asset or Excluded Liability;

(i) any and all Tax assets of Seller and its Affiliates, including, without limitation, any Tax claims or rights to Tax refunds, other than Tax claims or rights to Tax refunds in respect of Taxes for which Buyer is liable pursuant to Section 8.2;

(j) the assets, properties and rights set forth on Schedule 2.2(j) necessary in order for Seller to perform its obligations under the Transition Services Agreement (provided that such assets, properties and rights will become Purchased Assets at the time such obligations are fulfilled);

(k) the assets, properties and rights to the extent relating to the Retained Business (including positron emission tomography synthesis boxes) or any other business of Seller or its Affiliates, other than the Purchased Assets specifically enumerated in Section 2.1; and

(l) all rights of Seller under this Agreement or any of the other Transaction Documents.

2.3 <u>Assumption of Liabilities</u>. On the terms and subject to the conditions set forth in this Agreement, at the Closing, Buyer will assume, effective as of the Closing Date, and will, from and after the Closing Date, perform, satisfy and discharge, only the following Liabilities of Seller (collectively, the "<u>Assumed Liabilities</u>"):

(a) all Accounts Payable, Accrued Expenses, all accrued but unpaid vacation days up to the Closing Date or the applicable Transfer Date for each Employee who becomes a Transferred Employee, and Accrued Rebates whether arising prior to, on or after the Closing Date;

Page 14 of 59

(b) all Liabilities of Seller under the Assigned Real Property Leases, Assigned Contracts and Assigned Approvals (including all Liabilities arising out of, or relating to, any termination or announcement or notification by any third party of an intent to terminate any such Contract or Approval), and all Liabilities of Seller in respect of the instrument described on Schedule 3.6(b)(iv), but in each case only to the extent relating to the period from and after the Closing Date;

(c) all Liabilities arising out of or relating to employment, or service or compensation, employee benefits or termination of employment or service with respect to any Transferred Employees and their respective dependents and beneficiaries, in each case, in respect of service after the Closing Time, except as expressly set forth to the contrary in Section 6.5 (but subject to the reimbursement obligations of Buyer therein); and

(d) all Liabilities arising out of, relating to, or otherwise in respect of, the Radiopharmacy Business or Purchased Assets in respect of the period from and after the Closing Date.

2.4 <u>Excluded Liabilities</u>. Buyer will not assume or be liable for any Excluded Liabilities. Seller will retain, be responsible for, perform, satisfy and discharge all Excluded Liabilities. "<u>Excluded Liabilities</u>" means all Liabilities of Seller other than the Assumed Liabilities, including all of the following Liabilities:

(a) all Indebtedness (except for the Liabilities expressly assumed by Buyer in respect of the instrument described on Schedule 3.6(b)(iv) as set forth in Section 2.3(b) above);

(b) all Intercompany Payables;

(c) all Liabilities arising out of, relating to, or otherwise in respect of, the Radiopharmacy Business or Purchased Assets in respect of the period before the Closing Date (other than as specified in Section 2.3);

(d) all Liabilities under the Assigned Real Property Leases, Assigned Contracts and Assigned Approvals in respect of the period before the Closing Date;

(e) except as otherwise contemplated by Sections 6.1(a)-(b) and 7.2(d), all Liabilities incurred as a result of any Legal Proceedings (regardless of when asserted or initiated) to the extent that such Legal Proceeding arises out of, or relates to, (A) the operation of the Radiopharmacy Business or the ownership of the Purchased Assets before the Closing Date or (B) Seller's Retained Business;

(f) all Liabilities related to any Employee who does not become a Transferred Employee (other than Liabilities resulting from or arising out of Buyer's breach of Section 6.5);

(g) the Liabilities in respect of Transferred Employees expressly retained by Seller or the Seller Benefit Plans pursuant to Section 6.5 (but subject to the reimbursement obligations of Buyer therein);

Page 15 of 59

(h) all Liabilities in respect of Seller's employment of each Transferred Employee for any period prior the time he or she becomes a Transferred Employee (other than accrued but unpaid vacation days), including in respect of salary, wages, incentive pay, retirement plan contributions, equity compensation (including, without limitation, with respect to any awards granted under the Lantheus MI Holdings, Inc. 2008 Equity Incentive Plan and the Lantheus MI Holdings, Inc. 2013 Equity Incentive Plan) and in respect of change of control, retention, severance or other similar payments arising solely as a result of the consummation of the transactions contemplated by this Agreement;

(i) all Taxes of Seller and its Affiliates for any period (other than those specified in Section 2.3(d)), including (x) all Taxes relating to the Excluded Assets or Excluded Liabilities for any period and (y) all Taxes relating to the Purchased Assets or the Assumed Liabilities for any taxable period prior to the Closing Time and, with respect to any taxable period beginning before and ending after the Closing Time, for the portion of such taxable period ending on the Closing Date, other than Buyer's obligation to pay any Transfer Taxes to the Seller in addition to the Purchase Price as set forth in Section 8.2; and

(j) except as otherwise contemplated by Sections 6.1(a)-(b) and 7.2(d), arising or incurred in connection with the negotiation, preparation and execution of this Agreement and the transactions contemplated hereby and any fees and expenses of counsel, accountants, brokers, financial advisors or other experts of the Seller in connection with any of the foregoing; and

(k) all Liabilities arising out of, relating to, or otherwise in respect of, the Excluded Assets.

2.5 Non-Assignable Assets.

(a) Notwithstanding anything to the contrary in this Agreement, and subject to the provisions of this Section 2.5, to the extent that the sale, assignment, transfer, conveyance or delivery, or attempted sale, assignment, transfer, conveyance or delivery, to Buyer of any Purchased Asset or Assumed Liability would result in a violation of applicable Law, or would require the consent, authorization, approval or waiver of a Person who is not a party to this Agreement or an Affiliate of a party to this Agreement (including any Governmental Authority), and such consent, authorization, approval or waiver will not have been obtained prior to the Closing Date, then this Agreement will not be deemed to constitute a sale, assignment, transfer, conveyance or delivery, or an attempted sale, assignment, transfer, conveyance or delivery, thereof (and for the sake of clarity, notwithstanding anything to the contrary in this Section 2.5, all such assets and Liabilities and obligations will be treated as Purchased Assets and Assumed Liabilities, respectively, for all other purposes of this Agreement); provided, however, that the Closing will occur notwithstanding the foregoing without any adjustment to the Purchase Price and without giving rise to any rights of indemnification under Article 7 in favor of any of the Buyer's Indemnified Parties in respect of any becach of this Agreement to the extent that such breach was caused only by the *failure to obtain* such consent, authorization, approval or waiver (and Section 7.2(e) will apply in such situation); provided that, for further clarity, any breach resulting from Seller's *failure to disclose* in the Schedules the requirement to obtain any such consent, authorization, approval or waiver or the Seller's *failure to comply* with its covenants in

Page 16 of 59

this Section 2.5 and in Sections 6.2 and 6.3 with respect to such consent, authorization, approval or waiver, shall in each case remain subject to the rights of indemnification under Article 7 in favor of any of the Buyer's Indemnified Parties. Following the Closing Date, Buyer will use commercially reasonable efforts to obtain any such required consent, authorization, approval or waiver, and Buyer and Seller will cooperate with each other to obtain consents from the third parties that are described in Schedule 6.3; provided, however, that neither Seller nor Buyer will be required to pay any consideration therefor. Once such consent, authorization, approval, waiver, release, substitution or amendment is obtained, Seller will then sell, assign, transfer, convey and deliver to Buyer the relevant Purchased Asset or Assumed Liability to which such consent, authorization, approval, waiver, release for no additional consideration. Applicable Transfer Taxes in connection with such sale, assignment, transfer, conveyance or license will be paid by Buyer in accordance with Section 8.2, except to the extent applicable Transfer Taxes were already paid by the Buyer at the Closing Time.

(b) To the extent that any Purchased Asset and/or Assumed Liability has not or cannot be transferred to Buyer on or following the Closing Date pursuant to this Section 2.5, Buyer and Seller will use commercially reasonable efforts to enter into such arrangements (such as subleasing, sublicensing or subcontracting) to provide to the Parties the economic and, to the extent permitted under applicable Law, operational equivalent of the transfer of such Purchased Asset and/or Assumed Liability to Buyer from and after the Closing Date and the performance by Buyer of its obligations with respect thereto (and for the sake of clarity, notwithstanding anything to the contrary in this Section 2.5, all such assets and Liabilities and obligations will be treated as Purchased Assets and Assumed Liabilities, respectively, for all other purposes of this Agreement). Subject to applicable Law, Buyer will, as agent or subcontractor for Seller pay, perform and discharge fully the Liabilities and obligations of Seller thereunder from and after the Closing Date (and for the sake of clarity, notwithstanding anything to the contrary in this Section 2.5, all such assets and Liabilities and obligations will be treated as Purchased Asset of clarity, notwithstanding anything to the contrary in this Section 2.5, all such assets and bigations will be treated as Purchased Assets and Assumed Liabilities, respectively, for all other purposes of this Agreement). To the extent permitted under applicable Law, Seller will, at Buyer's expense, hold in trust for and pay to Buyer promptly upon receipt thereof, such Purchased Asset and all income, proceeds and other monies received by Seller to the extent related to such Purchased Asset in connection with the arrangements under this Section 2.5. Seller will be permitted to set off against such amounts all direct costs associated with the retention and maintenance of such Purchased Assets. Notwithstanding anything in this Agreement to the contrary, the provisions of this Section 2.5 will not apply to a

ARTICLE 3

PAYMENT; CLOSING

3.1 <u>Payment</u>. Subject to any adjustment hereunder, the aggregate consideration for the Purchased Assets will be (a) an amount in cash equal to US\$9,000,000 (the "<u>Purchase Price</u>") payable at Closing by the Buyer to the Seller, <u>plus</u> (b) the assumption, at Closing by the Buyer, of the Assumed Liabilities, <u>plus or minus</u> (c) the Post-Closing Adjustment (collectively, the "<u>Total Consideration</u>").

Page 17 of 59

3.2 Purchase Price Adjustment.

(a) Post-Closing Adjustment.

(i) Within *** (***) days after the date on which Buyer receives a copy of the balance sheet of the Radiopharmacy Business as of the Closing Date prepared by Seller under the Transition Services Agreement, Buyer will prepare and deliver to Seller (A) a statement setting forth its calculation of Closing Working Capital, which statement will be prepared in accordance with the Applicable Accounting Principles and will be substantially in the form of Schedule 3.2(a)(i)(1) (the "<u>Closing Working Capital Statement</u>"), and (B) a certificate of the Chief Financial Officer of Buyer that the Closing Working Capital Statement was prepared using the Applicable Accounting Principles. For purposes of facilitating interpretation of this Section 3.2, Schedule 3.2(a)(i)(2) contains an illustrative Closing Working Capital Statement calculating Closing Working Capital as of the Reference Time in accordance with the Applicable Accounting Principles.

(ii) The "<u>Post-Closing Adjustment</u>" will be an amount equal to the Closing Working Capital, minus \$*** (the "<u>Target Working Capital</u>"). If the Post-Closing Adjustment is a positive number, Buyer will pay to Seller an amount equal to the Post-Closing Adjustment. If the Post-Closing Adjustment is a negative number, Seller will pay to Buyer an amount equal to the absolute value of the Post-Closing Adjustment.

(b) Examination and Review.

(i) After receipt of the Closing Working Capital Statement, Seller will have *** (***) days from the date on which Buyer has provided to Seller all access and information reasonably requested for such purposes (the "<u>Review Period</u>") to review the Closing Working Capital Statement. During the Review Period, Seller and its accountants will have full access to the relevant books and records of Buyer, the personnel of, and work papers prepared by, Buyer and/or Buyer's accountants to the extent that they relate to the Closing Working Capital Statement and to such historical financial information (to the extent in Buyer's possession) relating to the Closing Working Capital Statement as Seller may reasonably request for the purpose of reviewing the Closing Working Capital Statement and to prepare a Statement of Objections.

(ii) On or prior to the last day of the Review Period, Seller may object to the Closing Working Capital Statement by delivering to Buyer a written statement setting forth Seller's objections in reasonable detail, indicating each disputed item or amount and the basis for Seller's disagreement therewith (the "<u>Statement of Objections</u>"). If Seller fails to deliver the Statement of Objections before the expiration of the Review Period, then the Closing Working Capital Statement and the Post-Closing Adjustment, as the case may be, reflected in the Closing Working Capital Statement will be deemed to have been accepted by

Page 18 of 59

Seller. If Seller delivers the Statement of Objections before the expiration of the Review Period, Buyer and Seller will negotiate in good faith to resolve such objections within *** (***) days after the delivery of the Statement of Objections (the "<u>Resolution Period</u>"), and, if the same are so resolved within the Resolution Period, then the Post-Closing Adjustment and the Closing Working Capital Statement with such changes as may have been previously agreed in writing by Buyer and Seller, will be final and binding.

(c) <u>Resolution of Disputes</u>. If Seller and Buyer fail to reach an agreement with respect to all of the matters set forth in the Statement of Objections before expiration of the Resolution Period, then any amounts remaining in dispute ("<u>Disputed Amounts</u>"; and any amounts not so disputed, the "<u>Undisputed Amounts</u>") will be submitted for resolution to Ernst & Young (as long as the individuals at such firm involved in resolving the dispute are independent of Buyer and Seller and their respective Affiliates) (the "<u>Independent Accountant</u>") which, acting as experts and not arbitrators, will resolve the Disputed Amounts only and make any adjustments to the Post-Closing Adjustment, as the case may be, and the Closing Working Capital Statement. The Independent Accountant will only decide the specific items under dispute by the Parties and its decision for each Disputed Amount must be in accordance with this Section 3.2 and the Applicable Accounting Principles and within the range of values assigned by Buyer and Seller to each such item in the Closing Working Capital Statement of Objections, respectively.

(d) <u>Fees of the Independent Accountant</u>. The fees and expenses of the Independent Accountant will be paid by Seller, on the one hand, and Buyer, on the other hand, based upon the percentage that the amount actually contested but not awarded to Seller or Buyer, respectively, bears to the aggregate amount actually contested by Seller and Buyer, respectively.

(e) <u>Determination by Independent Accountant</u>. The Independent Accountant will make a determination as soon as practicable within *** (***) days (or such other time as the Parties will agree in writing) after its engagement, and its resolution of the Disputed Amounts and their adjustments to the Closing Working Capital Statement and/or the Post-Closing Adjustment will be conclusive and binding upon the Parties.

(f) <u>Payments of Post-Closing Adjustment</u>. Except as otherwise provided in this Agreement, any payment of the Post-Closing Adjustment will (A) be due (x) within *** (***) Business Days of acceptance of the applicable Closing Working Capital Statement or, (y) if there are Disputed Amounts, then within *** (***) Business Days of the resolution thereof; and (B) be paid by wire transfer of immediately available funds to such account as is directed by Buyer or Seller, as the case may be.

(g) Adjustments for Tax Purposes. Any payments made pursuant to Section 3.2 will be treated as an adjustment to the Purchase Price by the parties for Tax purposes, unless otherwise required by Law.

3.3 <u>Allocation of Purchase Price</u>. Seller and Buyer agree that the Purchase Price and those Assumed Liabilities that are due, owing and outstanding as of the Closing Time (plus other relevant items) will be allocated among the Purchased Assets for all purposes (including Tax and

Page 19 of 59

financial accounting) as shown on the final allocation schedule (the "<u>Allocation Schedule</u>"). A draft of the Allocation Schedule will be prepared by Buyer and delivered to Seller within *** (***) days following the Closing Date. If Seller notifies Buyer in writing that Seller objects to one or more items reflected in the Allocation Schedule, Seller and Buyer will negotiate in good faith to resolve such dispute; provided, however, that if Seller and Buyer are unable to resolve any dispute with respect to the Allocation Schedule within *** (***) days following the Closing Date, such dispute will be resolved by the Independent Accountant. The fees and expenses of such accounting firm will be borne equally by Seller and Buyer. Buyer and Seller will file all Tax Returns (including amended returns and claims for refund) and information reports in a manner consistent with the Allocation Schedule. Any adjustments to the Purchase Price pursuant to Section 3.2 or Article 7 will be allocated in a manner consistent with the Allocation Schedule.

3.4 <u>Payment Method</u>. On the Closing Date, Buyer will pay the Purchase Price to, or on behalf of, Seller by wire transfer of immediately available funds denominated in U.S. dollars, into the accounts designated by Seller in writing not fewer than *** (***) Business Days before the Closing Date.

3.5 <u>The Closing</u>. The Transactions contemplated in this Agreement, including the execution of the Transaction Documents other than this Agreement (the "<u>Closing</u>") will take place remotely on January 13, 2016 or any other date on which the Parties mutually agree (the "<u>Closing Date</u>"), it being understood that the Closing will be effective as of 00:01 a.m. on the Closing Date (the "<u>Closing Time</u>"). The Parties will cooperate in good faith to finalize and mutually agree on all Exhibits and Schedules to this Agreement (including the other Transaction Documents, and the exhibits and schedules to those other Transaction Documents). and the finalized forms thereof will be attached to this Agreement at Closing. Each Party has the right to terminate this Agreement without any liability at any time prior to Closing upon written notice to the other Party.

3.6 Closing Deliveries.

(a) <u>Deliveries by Seller</u>. At the Closing Date, Seller will deliver or cause to be delivered to Buyer the following:

(i) a certificate of compliance under the Canada Business Corporations Act;

(ii) a certificate duly executed by Secretary of Seller in substantially the form of Exhibit E attached;

(iii) (A) a duly executed Bill of Sale; (B) a duly executed General IP Assignment; (C) a duly executed Supply Agreement; and (D) a duly executed Transition Services Agreement;

(iv) fully executed employment agreements with each of the Key Employees, and all duly executed Offer Letters for delivery to Employees other than Key Employees;

(v) Seller's Schedules;

Page 20 of 59

(vi) the elections referred to in Section 8.3; and

(vii) instruments and documents necessary to release any and all Liens on the Purchased Assets (except the Permitted Liens).

(b) Deliveries by Buyer. At the Closing Date, Buyer will deliver to Seller the following:

(i) a certificate of compliance under the Canada Business Corporations Act;

(ii) a certificate duly executed by Secretary of Seller in substantially the form of Exhibit F attached;

(iii) (A) a duly executed Bill of Sale; (B) a duly executed Supply Agreement; and (C) a duly executed Transition Services Agreement;

(iv) evidence reasonably satisfactory to Seller that Buyer has assumed the Seller's Liabilities, to the extent relating to the period from and after the Closing Date, in respect of the instrument described on Schedule 3.6(b)(iv);

(v) the elections referred to in Section 8.3; and

(vi) evidence of wire transfer(s) of the Purchase Price referred to in Section 3.1.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in the Schedules (such that a disclosure set forth in any section of the Schedules numbered to correspond with any section of the Agreement shall apply to that section only), Seller represents and warrants to Buyer as follows as of the Closing Date (or as of any other specifically referenced date):

4.1 <u>Organization and Corporate Power</u>. Seller is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of formation. Seller has the requisite corporate power and authority to own, operate or lease the properties that it purports to own, operate or lease and to carry on its business as it is now being conducted and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not have, or would not be reasonably expected to have, a Material Adverse Effect. The Seller does not own, directly or indirectly, any share or other equity, securities or ownership interest in any other corporation or in any other Person.

4.2 Due Authorization. Seller has the requisite corporate power and authority to execute and deliver the Transaction Documents and to consummate the Transactions. The

Page 21 of 59

execution, delivery and performance by Seller of the Transaction Documents and the consummation by Seller of the Transactions have been duly authorized by all necessary corporate action on the part of Seller, and no other corporate proceeding is necessary for the execution and delivery of the Transaction Documents by Seller, the performance by Seller of its obligations thereunder and the consummation by Seller of the Transactions. Each of the Transaction Documents has been duly executed and delivered by Seller and constitutes the legal, valid and binding obligation of Seller, enforceable against Seller in accordance with its terms, except as the same may be limited by bankruptcy, insolvency, moratorium, reorganization or other Laws of general applicability relating to or affecting the enforcement of creditor's rights and general principles of equity.

4.3 No Violation; Consents.

(a) Except as set forth in Schedule 4.3(a), but without taking into account the provisions of Section 2.5, the execution, delivery and performance by Seller of the Transaction Documents do not and will not: (i) subject to Buyer's satisfaction of its obligations under Sections 6.1(a)-(b) and 7.2(d), violate any material Law or Order applicable to Seller or any of its properties or assets (including the Purchased Assets); (ii) violate or conflict with, result in a breach of, constitute a default (or an event which, with or without notice or lapse of time or both, would constitute a default) under, permit cancellation of, or result in the creation of any Lien upon any of Seller's properties or assets (including the Purchased Assets) under, any material Contract to which Seller is a party or by which it or its properties and assets are bound (including the Assigned Contracts); or (iii) violate or conflict with any provision of Seller's organizational documents.

(b) Except for the approvals, notices and consents of, and filings with, the Governmental Authorities or other Persons not a party to this Agreement set forth on Schedules 4.3(b), but without taking into account the provisions of Section 2.5, no consents, notices or approvals of, or filings or registrations by Seller with, any Governmental Authority or any other Person not a party to this Agreement are necessary in connection with the execution, delivery and performance of the Transaction Documents or the Transactions

4.4 <u>Financial Statements</u>. Schedule 4.4(a) sets forth unaudited financial statements of the Radiopharmacy Business consisting of (a) the balance sheets of the Radiopharmacy Business as at December 31, 2014 and December 31, 2015 and the related statements of income for the fiscal years ended December 31, 2013, December 31, 2014 and December 31, 2015 (collectively, the "<u>Financial Statements</u>"). The Financial Statements have been specially prepared for purposes of this Agreement from the books and records of Seller and using the Applicable Accounting Principles and (taking into account the Applicable Accounting Principles) fairly present, in all material respects, the financial position and results of operations of the Radiopharmacy Business as of the dates and for the periods presented. The Accounts Payable, Accrued Expenses, and Accrued Rebates (in each case, as of the Reference Time) are set forth and listed on Schedule 4.4(b), together with aging and past due reports thereon, and such listing and reports are true, complete and correct in all material respects.

4.5 <u>Undisclosed Liabilities</u>. Except as set forth in Schedule 4.5, Seller has no Liabilities of the type required to be reflected or reserved against on a balance sheet prepared in

Page 22 of 59

accordance with U.S. generally accepted accounting principles with respect to the Radiopharmacy Business, except (a) those which are adequately reflected or reserved against in the balance sheet for the year ended December 31, 2015 included in the Financial Statements, (b) for Excluded Liabilities, (c) for future executory obligations arising under any Assigned Contracts or Assigned Approvals and (d) those that have been incurred in the Ordinary Course of Business since December 31, 2015 and which are not, individually or in the aggregate, material in amount (except to the extent included in the calculation of Closing Working Capital).

4.6 Absence of Certain Changes. Except as expressly contemplated by this Agreement or as set forth in Schedule 4.6, since the Interim Reference Date and through immediately prior to the Closing Date, Seller has operated the Radiopharmacy Business in the Ordinary Course of Business in all material respects and there has not been, with respect to the Radiopharmacy Business, (i) any event, occurrence or development that has had a Material Adverse Effect, (ii) any sale, transfer or disposition of any properties or assets, or any grant of a license to Assigned Intellectual Property, in either case, outside of the Ordinary Course of Business, (iii) any purchase, lease or otherwise acquisition of the right to own, use or lease any property or assets or any capital expenditures or commitments for an amount in excess of \$*** individually (in the case of a lease, per annum), or \$*** in the aggregate (in the case of a lease, for the entire term of the lease, not including any option term), except for purchases of raw materials, inventory or supplies in the Ordinary Course of Business, (iv) any damage, destruction, property loss or other similar event (in each case, whether or not covered by insurance) involving \$*** or more in damages, losses, replacement or repair costs, (v) any material change in accounting practices and policies, practices and procedures with respect to the manner of its billings, or the credit terms made available by it, to any of its customers, the collection of Accounts Receivable, establishment of reserves for uncollectible Accounts Receivable, accrual of Accounts Receivable, inventory control, prepayment of expenses, deferral of revenue and acceptance of customer deposits, (vi) any material change in the terms of employment or engagement or in compensation or other benefits payable or to become payable (or any acceleration of any compensation or benefit), in each case with respect to Employees or consultants of the Radiopharmacy Business, (vii) any hiring or termination of any Employees or consultants of the Radiopharmacy Business outside the Ordinary Course of Business, (viii) any material change to, or entering into or termination of, any Employee Benefit Plan, (ix) any material write-down of the value of its assets or any write-off as uncollectible of its Accounts Receivable or any portion thereof, in each case, in an amount in excess of \$*** in the aggregate, (x) any acceleration, termination, material modification, material waiver of, or failure to perform any of its material obligations under any material Assigned Contract or Assigned Approval, (xi) any entering into of any new material Contract, (xii) any settlement or compromise of any material Legal Proceeding, or (xiii) any board authorization or legally binding agreement or commitment in respect of any of the foregoing.

4.7 Intellectual Property.

(a) Schedule 4.7 lists all registered Assigned Intellectual Property. Payments of all necessary registration, maintenance and renewal fees currently due in connection with such registered Assigned Intellectual Property have been made and all necessary documents, recordations and certifications in connection with such registered Assigned Intellectual Property have been filed with the relevant Governmental Authorities for the purpose of maintaining such

Page 23 of 59

registered Assigned Intellectual Property. There has been no claim asserted or, to the Seller's Actual Knowledge, threatened, and there are no proceedings of any kind pending or in progress, challenging the scope, validity or enforceability of any such registered Assigned Intellectual Property.

(b) To Seller's Knowledge: (i) the conduct of the Radiopharmacy Business as currently conducted, does not infringe, misappropriate, dilute or otherwise violate the Intellectual Property of any Person; and (ii) no Person is infringing, misappropriating or otherwise violating any material Assigned Intellectual Property. Notwithstanding anything to the contrary in this Agreement, this Section 4.7 constitutes the sole representation and warranty of the Seller under this Agreement with respect to any actual or alleged infringement, misappropriation or other violation by Seller of any Intellectual Property of any other Person.

(c) The Seller has taken commercially reasonable steps to protect its rights in that portion of its Confidential Information relating exclusively to the Radiopharmacy Business and any trade secret or confidential information of third parties used by the Seller in the conduct of the Radiopharmacy Business.

(d) All Assigned Intellectual Property developed by or for the Seller was done so by employees or contractors who have assigned all of their Intellectual Property rights therein to the Seller and have waived all non-assignable Intellectual Property rights.

(e) The Seller has a written privacy policy and procedures, a copy of which has been made available to Buyer.

(f) Any material information technology (including material computer hardware, software, websites, databases, networks, telecommunications equipment and facilities and other information or communication technology systems) included in the Purchased Assets and necessary to the conduct of the Radiopharmacy Business as it was conducted by Seller immediately prior to the Closing Date (i) operates and performs in all material respects as required to conduct the Radiopharmacy Business as it was conducted by Seller immediately prior to the Closing Date and (ii) has been subject to commercially reasonable standards intended to (A) protect against unauthorized use and (B) provide backup services, in the case of each of clause (A) and (B), in a manner intended to prevent material adverse interruptions of the conduct of the Radiopharmacy Business. Except as set forth on Schedule 4.7(f), the Radiopharmacy Business' ownership of, or lease rights or licenses to such information technology will not be lost by virtue of the Transactions contemplated by this Agreement.

4.8 <u>Litigation and Claims</u>. There is no Legal Proceeding (whether in or outside the Territory) pending or, to Seller's Knowledge, threatened in writing, against or affecting (i) Seller that would prohibit or materially hinder, delay or otherwise impair Seller's ability to perform its obligations under the Transaction Documents, including the assignment of the Purchased Assets, that would affect the legality, validity or enforceability of the Transaction Documents, or that would prevent or materially delay the consummation of the Transactions, (ii) the Radiopharmacy Business or (iii) the Purchased Assets.

Page 24 of 59

4.9 Good and Valid Title. Except as set forth on Schedule 4.9, Seller has the right to acquire and transfer to Buyer, good and valid title to the Purchased Assets, in each case free and clear of all Liens (other than Permitted Liens).

4.10 <u>Assigned Contracts</u>. Seller has provided Buyer with true, complete and unredacted copies of each Assigned Contract, including all amendments, supplements and modifications to each such Assigned Contract. All of the Assigned Contracts are the valid, binding obligations of Seller, enforceable by and against Seller in accordance with their respective terms, except as the same may be limited by bankruptcy, insolvency, moratorium, reorganization or other Laws of general applicability relating to or affecting the enforcement of creditor's rights and general principles of equity. Neither Seller nor, to Seller's Knowledge, any other party thereto is in material breach or default under any provision of any Assigned Contract.

4.11 <u>Real Property</u>. Seller owns no real property that is included in the Purchased Assets. Schedule 2.1(b) sets forth (a) all real property that is leased by Seller and primarily used in connection with the Radiopharmacy Business (collectively, the "Leased Real Property") and (b) the Assigned Real Property Leases pertaining to the Leased Real Property. Seller has made available to Buyer true, complete, unredacted and correct copies of the Assigned Real Property Leases, including, without limitation, all modifications, extensions or amendments thereto.

4.12 Approvals; Compliance with Laws.

(a) Schedule 4.12(a) sets forth a true and complete list of all material Approvals held by Seller primarily relating to the Radiopharmacy Business (the "<u>Scheduled Approvals</u>"). Seller validly holds, possesses or has rights to, the Scheduled Approvals, and all Scheduled Approvals are in full force and effect in all material respects. The Scheduled Approvals constitute all material Approvals necessary and sufficient for the operation of the Radiopharmacy Business in the manner operated as of immediately prior to the Closing Date, including all Approvals necessary for the lawful manufacturing, distribution, storage, containment, disposal, transport, handling, sale or promotion, if and as applicable, of each of the products that are manufactured, distributed, compounded or sold by the Radiopharmacy Business. Since ***, Seller has been in compliance in all material noncompliance with any Scheduled Approvals or that any Governmental Authority is considering any Legal Proceeding to limit, revoke, suspend or modify any Scheduled Approval.

(b) Seller is, and since *** has been, in compliance, in all material respects with those Laws, including the Food And Drugs Act and the regulations of Health Canada promulgated thereunder, the Nuclear Safety and Control Act and the regulations of the CNSC promulgated thereunder, applicable to (x) the operation of the Radiopharmacy Business or (y) the Purchased Assets. Seller has not received since *** written notice of material noncompliance with any such Laws.

(c) Since ***, all material reports, documents, claims, permits, applications, accreditations and notices required by applicable Law to be filed, maintained or furnished to Health Canada, the CNSC or any Governmental Authority by the Seller with respect to the

Page 25 of 59

Radiopharmacy Business have been so filed, maintained or furnished. All such reports, documents, claims, permits, applications, and notices were complete and accurate in all material respects on the date filed (or were corrected in or supplemented by a subsequent filing).

(d) Other than normal-course facility audits conducted by Governmental Authorities, the Radiopharmacy Business is not, to Seller's Knowledge, the subject of any investigation by Health Canada, the CNSC or any other Governmental Authority regarding violations or alleged violations of Law. Since ***, the Seller has maintained in all material respects all material records as required by applicable Laws with respect to the manufacturing, compounding, distribution or sale of products in the conduct of the Radiopharmacy Business.

(e) Since ***, Seller has not received any written notice of material adverse finding, notice of violation, warning letter, or other correspondence or notice from Health Canada, the CNSC or any other Governmental Authority with respect to the Radiopharmacy Business.

(f) Since ***, the Seller has not received any written notice that Health Canada, the CNSC or any other Governmental Authority has commenced, or threatened to initiate, any action to cause the recall, market withdrawal or replacement of any product manufactured, compounded, distributed or sold or intended to be sold by the Radiopharmacy Business.

4.13 Environmental Laws.

(a) The Radiopharmacy Business and the Leased Real Properties are in compliance, in all material respects, with all Environmental Laws. There is no pending claim or, to the Seller's Knowledge, threatened claims pursuant to any Environmental Laws with respect to any of the Leased Real Properties. In the last *** (***) years, the Seller has not been convicted of an offense for non-compliance with Environmental Laws, been fined or received a penalty for non-compliance with Environmental Laws or settled a lawsuit relating to non-compliance with Environmental Laws, in respect to the Radiopharmacy Business.

(b) The Seller has not received, in the last *** (***) years, any directive, inquiry, notice, Order, warning or other communication from any Governmental Authority or other Persons that relates to any offence or failure or any non-compliance real, alleged or potential in connection with any applicable Environmental Laws.

4.14 Employee Matters.

(a) Schedule 4.14(a) contains a true, correct and complete list of (i) all of the Employees, and in each case as of the Closing Date, specifying their position or title, annual salary, bonus and other compensation, hourly wages, date of hire, work location (i.e., province where services are performed), length of service, full or part time employment, equity compensation, employee benefits (including type of coverage under the Seller Benefit Plans), vacation entitlements, accrued and unused vacation days as of the Closing Date, any other annual paid time off entitlements in days and each employee's accrued and unused days of such other paid time off, annual bonus and/or commission for *** and their status as active or inactive, and whether the listed Employee is subject to a written employment agreement; provided that,

Page 26 of 59

notwithstanding the foregoing, Schedule 4.14(a) will not contain any nominal information associated with any Quebec Employee; and (ii) all independent contractors who provided services to the Seller specifying the name of the independent contractor, type of services, length of time services have been provided, work location (i.e., province where services have been provided), and fees (including any commissions, royalties and bonuses) paid to such independent contractor for ***.

(b) Since ***, Seller has not received a written claim from any Governmental Authority to the effect that the Seller has improperly classified any Person providing services to the Radiopharmacy Business as an independent contractor and, to the Seller's Knowledge, no such claim has been threatened in writing. The Seller has not made any written or verbal legally binding commitments to any Employee, former Employee, consultant or independent contractor of the Seller with respect to compensation, promotion, retention, termination, severance or similar matters in connection with the Transactions. The Seller has delivered to the Buyer true, correct and complete copies of all Contracts with each Employee and each independent contractor.

(c) Seller is not a party to any collective bargaining Contract with a labor union or labor organization applicable to any Employee, nor is Seller under any current obligation to bargain with any bargaining agent on behalf of any such Employee. Seller is not the subject of any Legal Proceeding pertaining to the Employees asserting that it has committed an unfair labor practice or is seeking to compel it to bargain with any labor union or labor organization and, to Seller's Knowledge, no such Legal Proceeding is pending or has been threatened in writing. No labor strike, dispute, walkout, work stoppage, slow-down or lockout involving any Employee has occurred since ***. Since ***, no labor union, labor organization, or group of Employees has made a written demand for recognition or certification, and there are no representation or certification proceedings or petitions seeking a representation proceeding presently pending or threatened in writing to be brought or filed with any labor relations tribunal or authority by, or for the benefit of, any Employees. To Seller's Knowledge, there are no labor union organizing activities occurring with respect to any Employee.

(d) Seller is in compliance in all material respects with all applicable Laws pertaining to labor, employment and employment practices to the extent they relate to the Employees.

(e) There are no outstanding material assessments, penalties, fines, liens, charges, surcharges or other amounts due or owing in respect of the Employees pursuant to any applicable workers' compensation legislation, and the Seller has not been assessed or reassessed in any material respect under such legislation since *** and, to Seller's Knowledge, no audit of the Seller is currently being performed pursuant to any applicable workers' compensation legislation. Except as set forth on Schedule 4.14(e), there are no claims or pending claims which may materially adversely affect the accident cost experience in respect of the Radiopharmacy Business.

(f) Seller has provided to the Buyer all orders and inspection reports under applicable occupational health and safety legislation ("<u>OHSA</u>") received by the Seller since *** pertaining to the Radiopharmacy Business. There are no charges pending under OHSA

Page 27 of 59

pertaining to the Radiopharmacy Business and the Seller has complied in all material respects with any orders issued under OHSA pertaining to the Radiopharmacy Business and there are no appeals of any orders under OHSA pertaining to the Radiopharmacy Business currently outstanding.

(g) There is no Legal Proceeding or pending Legal Proceeding by an Employee or an independent contractor of the Radiopharmacy Business under any Law against the Seller, or, to Seller's Knowledge, threatened in writing against the Seller, by any Employee or independent contractor of the Radiopharmacy Business.

(h) Schedule 4.14(h) contains a true, correct and complete list of each Seller Benefit Plan. True, correct, current and complete copies of each Seller Benefit Plan or where oral, written summaries of the material terms thereof have been made available to Buyer.

(i) Except as set forth on Schedule 4.14(i), Seller has not announced or entered into any plan or binding commitment to (x) create or cause to exist any additional Seller Benefit Plan or improve any benefits thereunder, or (y) adopt, amend or terminate any Seller Benefit Plan.

(j) None of the Seller Benefit Plans provides for retiree or post-employment benefits.

4.15 <u>Sufficiency of Assets and Rights</u>. Except for the rights and assets set forth on Schedule 4.15(a), the Purchased Assets, together with the rights granted to Buyer under the Transaction Documents (including Section 2.5 and the Transition Services Agreement) and taking into account the Excluded Assets and Seller's obligations under Section 6.13, are adequate and sufficient to operate the Radiopharmacy Business on the Closing Date in substantially the same manner as carried on by Seller as of immediately prior to the Closing Date. Except for the rights and assets set forth on Schedule 4.15(b), no Person other than the Seller owns any assets used by Seller in the Radiopharmacy Business immediately prior to the Closing Date in substantially the same manner as carried on by Seller as of immediately the same manner as carried on by Seller as of immediately the same manner as carried on by Seller as of insubstantially the same manner as carried on by Seller as of immediately the same manner as carried on by Seller as of immediately the same manner as carried on by Seller as of immediately prior to the Closing Date in substantially the same manner as carried on by Seller as of immediately prior to the Closing Date.

4.16 Tax Matters.

(a) The Seller is not a "non-resident" of Canada within the meaning of the Tax Act.

(b) The Seller is registered for GST/HST under Part IX of the ETA and for QST under the QSTA. The Seller's GST/HST registration number is *** and QST registration number is ***.

(c) The Seller has timely paid in full all Taxes and all installments of Taxes due on or before the Closing Date (whether or not shown on any Tax Return), which, if unpaid, could result in a Lien upon any of the Purchased Assets.

Page 28 of 59

(d) All assessments and reassessments received by the Seller in respect of Taxes have been paid. There are no assessments or reassessments of Taxes for the Seller that have been issued and are under dispute, which, if unpaid, could result in a Lien upon any of the Purchased Assets.

(e) Seller has withheld or collected, and will have at the Closing Time, withheld or collected any Taxes that are required by Law to be withheld or collected prior to the Closing Time and has timely paid or remitted, and will have at the Closing Time paid and remitted, on a timely basis, the full amount of any Taxes that have been withheld or collected and which are due on or before the Closing Time, to the applicable Governmental Authority.

4.17 <u>Broker's Fees</u>. Seller has not incurred any Liability for any brokerage, finder's or other fee or commission, in connection with the Transactions (other than such fees or commissions for which Seller is solely responsible).

4.18 <u>Personal Property</u>. All of the material tangible Purchased Assets (i) are in good operating condition and in state of good maintenance and repair, in each case, subject to normal wear and tear and taking into account their use, age and useful life and (ii) were acquired in the Ordinary Course of Business. Except as set forth on Schedule 4.18(a), all of the material tangible Purchased Assets are located at the Leased Real Property. Schedule 4.18(b) sets forth a true, correct and complete list and general description of each category of tangible Purchased Asset having a book value (for that category) as of the Reference Time of more than \$***, indicating the purchase value and purchase date of each such category of tangible Purchased Assets, including the net book value for each category of such Purchased Assets as at the Reference Time. All material tangible Purchased Assets which are leased or financed are set forth on Schedule 4.18(c).

4.19 <u>Inventory</u>. The Seller has good and valid title to the Inventory included in the Purchased Assets, free and clear of all Liens (other than Permitted Liens). All Inventory included in the Purchased Assets (other than such Inventory having no material shelf life in the Ordinary Course of Business) is in good condition. A true, correct and complete list of the quantity of each item of Inventory as of the Reference Time is set forth on Schedule 4.19.

4.20 <u>Accounts Receivable</u>. To Seller's Actual Knowledge, the debtors to which the Accounts Receivable of the Seller relate are not in or subject to a bankruptcy or insolvency proceeding. All Accounts Receivable represent monies due for goods sold and delivered or services rendered, in each case in the Ordinary Course of Business. Except as set forth on Schedule 4.20(a), to Seller's Knowledge, there are no material disputes regarding the collectability of any such Accounts Receivable. The Accounts Receivable and Prepaid Expenses (in each case, as of the Reference Time) are set forth and listed on Schedule 4.20(b) together with, where applicable, aging and past due reports thereon (in each case, as of the Reference Time), and such listing and reports are true, complete and correct in all material respects.

4.21 <u>Affiliate Matters</u>. Except as set forth on Schedule 4.21, no (a) Affiliate of Seller or, (b) in each case to Seller's Knowledge, any director or officer of Seller or its Affiliates, is a party to any (i) Assigned Contract, (ii) Assumed Liability that is due, owing and outstanding as of the Closing Time or (iii) lease of property (real, personal or mixed, tangible) included in the Purchased Assets.

Page 29 of 59

4.22 <u>Customer and Supplier Relations</u>. Schedule 4.22(a) contains a true, correct and complete list of the names of (i) the customers of the Radiopharmacy Business who purchased in excess of \$*** of the Seller's products per calendar year, either in the calendar year of ***, (ii) any Person who supplied in excess of \$*** of goods to the Radiopharmacy Business per calendar year, either in the calendar year of ***. Except as set forth on Schedule 4.22(b), no such customer or supplier since *** has affirmatively exercised a right by providing a written notice of cancellation or termination for any of the Assigned Contracts, and to Seller's Actual Knowledge, no such customer or supplier since *** has made any bona fide threat (whether orally or in writing) to cancel or otherwise terminate any of the Assigned Contracts.

4.23 Disclaimer of Other Representations.

(A) NEITHER SELLER NOT ANY OF ITS REPRESENTATIVES HAS MADE ANY REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, OF ANY NATURE WHATSOEVER RELATING TO SELLER OR ANY OF ITS AFFILIATES OR THE BUSINESS OF SELLER OR ANY OF ITS AFFILIATES (INCLUDING THE RADIOPHARMACY BUSINESS OR PURCHASED ASSETS) OR OTHERWISE IN CONNECTION WITH THE TRANSACTIONS, OTHER THAN THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN THIS ARTICLE 4.

(B) Without limiting the generality of clause (a) above, except as expressly set forth in this Article 4, neither Seller nor any of its Representatives has made, and will not be deemed to have made, any representations or warranties in the materials relating to the business of Seller or its Affiliates (including the Radiopharmacy Business and Purchased Assets) made available to Buyer and its Representatives, including due diligence materials, or in any presentation of the business of Seller or its Affiliates by management of Seller or others in connection with the Transactions, and no statement contained in any of such materials or made in any such presentation will be deemed a representation or warranty under this Agreement or otherwise or deemed to be relied upon by Buyer in executing, delivering and performing this Agreement and the Transactions. Except as expressly set forth in this Article 4, it is understood that any cost estimates, projections or other predictions, any data, any financial information or any memoranda or offering materials or presentations are not and will not be deemed to be or to include representations or warranties of Seller, and are not and will not be deemed to be relied upon by Buyer in executing, delivering and performing the Transactions.

Page 30 of 59

ARTICLE 5

REPRESENTATIONS AND WARRANTIES OF BUYER

Except as set forth in the Buyer Disclosure Schedules, Buyer represents and warrants to Seller as follows as of the Closing Date:

5.1 Organization and Corporate Power. Buyer is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of formation. Buyer has the requisite corporate power and authority to own, operate or lease the properties that it purports to own, operate or lease and to carry on its business as it is now being conducted and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not have, or would not be reasonably expected to have, a material adverse effect on Buyer's ability to consummate the Transactions.

5.2 Due Authorization. Buyer has the requisite corporate power and authority to execute and deliver the Transaction Documents and to consummate the Transactions. The execution, delivery and performance by Buyer of the Transaction Documents and the consummation by Buyer of the Transactions have been duly authorized by all necessary corporate action on the part of Buyer, and no other corporate proceeding is necessary for the execution and delivery of the Transaction Documents by Buyer, the performance by Buyer of its obligations thereunder and the consummation by Buyer of the Transactions. Each of the Transaction Documents has been duly executed and delivered by Buyer and constitutes the legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, except as the same may be limited by bankruptcy, insolvency, moratorium, reorganization or other Laws of general applicability relating to or affecting the enforcement of creditor's rights and general principles of equity.

5.3 No Violation; Consents.

(a) The execution, delivery and performance by Buyer of the Transaction Documents do not and will not: (i) violate any material Law or Order applicable to Buyer or any of its properties or assets; (ii) violate or conflict with, result in a breach of, constitute a default (or an event which, with or without notice or lapse of time or both, would constitute a default) under, permit cancellation of, or result in the creation of any Lien upon any of Buyer's properties or assets including the Purchased Assets (other than any Lien imposed by Buyer's lenders under its existing credit facilities, as may be amended from time to time) under, any material Contract to which Buyer is a party or by which it or its properties and assets are bound; or (iii) violate or conflict with any provision of Buyer's organizational documents.

(b) Except for the approvals, notices and consents of, and filings with, the Governmental Authorities or other Persons not a party to this Agreement set forth on Schedule 5.3(b), no consents, notices or approvals of, or filings or registrations by Buyer with, any Governmental Authority or any other Person not a party to this Agreement are necessary in connection with the execution, delivery and performance of the Transaction Documents or the Transactions.

(c) Buyer is a "WTO investor" and is not a "state-owned enterprise" for purposes of the Investment Canada Act.

5.4 Litigation. There is no Legal Proceeding pending or, to Buyer's knowledge, threatened in writing, against or affecting Buyer that would prohibit or materially hinder, delay

Page 31 of 59

or otherwise impair Buyer's ability to perform its obligations under the Transaction Documents, including the assumption of the Assumed Liabilities, that would affect the legality, validity or enforceability of this Agreement or the Transaction Documents, or that would prevent or materially delay the consummation of the Transactions.

5.5 <u>Financial Capability</u>. Buyer has sufficient financial resources to pay the Purchase Price and any expenses incurred by Buyer in connection with the Transactions and consummate the Transactions. Buyer has provided Seller with true, complete and correct copy of all consents required by its lenders under its debt financing arrangements.

5.6 Solvency. Immediately after giving effect to the Transactions, Buyer will be solvent and will: (a) be able to pay its debts as they become due (including those arising under the Supply Agreement and Transition Services Agreement); (b) own property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent Liabilities); and (c) have adequate capital to carry on its business. No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of Buyer or Seller. In connection with the Transactions, Buyer has not incurred, nor currently plans to incur, debts beyond its ability to pay as they become absolute and matured.

5.7 <u>Approvals</u>. Buyer holds a Nuclear License in respect of its business that is currently valid, in full force and effect and in good standing. Buyer is in compliance in all material respects with all of the terms and conditions of its Nuclear License. Buyer has no reason to believe that the CNSC will not amend its existing Nuclear License as required to continue the conduct of the Radiopharmacy Business in substantially the same manner as conducted immediately prior to the Closing Date.

5.8 Tax Matters.

(a) The Buyer is not a "non-resident" of Canada within the meaning of the Tax Act.

(b) The Buyer is registered for GST/HST under Part IX of the ETA and for QST under the QSTA. The Buyer's GST/HST registration number is *** and QST registration number is ***.

5.9 Broker's Fees. Buyer has not incurred any Liability for any brokerage, finder's or other fee or commission, in connection with the Transactions (other than such fees or commissions for which Buyer is solely responsible).

5.10 Inspection: No Other Representations. Buyer is an informed and sophisticated Person, and has engaged expert advisors experienced in the evaluation and acquisition of assets such as the Purchased Assets as contemplated under this Agreement. Buyer acknowledges that, except as expressly set forth in Article 4, (a) neither Seller nor its Representatives makes any representation or warranty with respect to (i) any projections, estimates or budgets delivered to or made available to Buyer or (ii) any other information or documents made available to Buyer or its Representatives with respect to the Purchased Assets or Seller, its Affiliates and their respective businesses and (b) Buyer has not relied and will not rely upon any of the information

Page 32 of 59

described in subclauses (i) and (ii) of clause (a) above or any other information, representation or warranty except those representations and warranties set forth in Article 4 of this Agreement in executing, delivering and performing this Agreement and the Transactions.

ARTICLE 6

COVENANTS AND AGREEMENTS

6.1 <u>Efforts</u>. Upon the terms and subject to the conditions of this Agreement, each of the Parties will use reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Laws to consummate and defend the Transactions. From the Execution Date through the Closing Date, Seller will operate the Radiopharmacy Business in the Ordinary Course of Business in all material respects. Not in limitation of the foregoing, and notwithstanding any limitation in the foregoing:

(a) If an application (not including, for greater certainty, mere investigations or inquiries) is initiated by the Commissioner under section 92 of the Competition Act seeking an Order from the Competition Tribunal that would rescind, dissolve or otherwise unwind any of the Transactions or any provision of any Transaction Document, in whole or in part, then in each such case, Buyer and its Affiliates and Subsidiaries will, prior to an Order being made by the Competition Tribunal, take any and all actions (at its and their own cost and expense) including by selling or divesting, or agreeing to sell or divest, assets (including any of the Purchased Assets) to a third party or modifying or terminating, or agreeing to modify or terminate, any existing commercial relationships (other than with Seller) or Contracts (other than any of the Transaction Documents) or entering into, or agreeing to enter into, supply agreements and/or agreements for the licensing of intellectual property, in each case, solely to the extent required to avoid an Order from the Competition Tribunal requiring rescission, dissolution or other unwinding of any of the Transactions in a manner that involves Seller or its Affiliates or adversely affects any of Seller's rights under the Transaction Documents, or that would involve the put or other sale of any of the Purchased Assets or Assumed Liabilities back to Seller or its Affiliates; provided further that Buyer will, and will cause its Affiliates, to take any and all actions (at its and their own cost and expense) to have vacated, lifted, reversed, repealed or rescinded any Order from the Competition Tribunal requiring rescission, dissolution or other unwinding of any of the Transactions in a manner that involves Seller or its Affiliates or adversely affects any of Seller's rights under the Transaction Documents, or that would involve the put or other sale of any of the Purchased Assets or Assumed Liabilities back to Seller or its Affiliates; provided further that Buyer will not consent to any agreement with the Commissioner (whether under Section 92 of the Competition Act or otherwise) that involves rescission, dissolution or unwinding of any of the Transactions, in whole or in part, or that involves the put or sale of any of the Purchased Assets or Assumed Liabilities back to Seller or its Affiliates or that adversely affects any of Seller's rights under the Transaction Documents; and provided further that Buyer will not be required to take actions that, in the aggregate, would cause a material adverse effect on its combined business (including the Radiopharmacy Business). In addition, Buyer will provide Seller on a timely basis with a copy of all communications and correspondence with respect to the Transactions, including attachments and briefs, between (A) Buyer, its Affiliates and its advisors, on the one hand, and (B) the Commissioner of Competition, on the other.

Page 33 of 59

(b) Buyer agrees, and will cause its Affiliates, to exercise reasonable best efforts to take any and all actions (at its and their own cost and expense) necessary to avoid, contest and defend against any dissolution, rescission or other unwinding or modification of the Transactions, in whole or in part, under the *Investment Canada Act*, and to have vacated, lifted, reversed, repealed or rescinded any Order for dissolution, rescission or other unwinding or modification of the Transactions that may be initiated, asserted or issued by any Governmental Authority with respect to the Transactions, under the *Investment Canada Act*, including by way of notice, requirement, referral, investigation, inquiry, Order or Legal Proceeding.

6.2 Maintenance of Assigned Approvals.

(a) Buyer's Obligations.

(i) Until all Assigned Approvals have been transferred into the name of Buyer and Buyer has obtained all of its own Approvals necessary to own and operate the Radiopharmacy Business and Purchased Assets, as applicable, Buyer covenants to operate the Radiopharmacy Business in a manner (including with respect to personnel, equipment, facilities, processes, products, ingredients, manufacturing processes, manufacturing controls, tests or labels used in the operation of the Radiopharmacy Business) that does not, and reasonably would not, adversely affect its ability to have transferred into its name, or obtain for itself, any such Approvals.

(ii) Buyer will obtain, promptly on or after the Closing Date, all Approvals required by the Governmental Authorities to enable Buyer to conduct the Radiopharmacy Business and hold the Purchased Assets independently from Seller.

(iii) Subject to Seller's obligations under Section 6.2(b), until such time as all Assigned Approvals have been transferred to Buyer, Buyer will be responsible for maintaining the Assigned Approvals (including payment of any fees) as if such Assigned Approvals have been transferred to Buyer on the Closing Date.

(iv) For the avoidance of doubt, following the date on which all of the Assigned Approvals are transferred, Buyer will be solely responsible for (i) determining the regulatory plans and strategies for the Radiopharmacy Business and the Purchased Assets, (ii) (either itself, through its Affiliates or through its authorized designee) seeking all consents, notices or approvals of, and making all filings or registrations with any Governmental Authority or any other Person not a party to this Agreement that are necessary in connection with the continuation of the Radiopharmacy Business and (iii) maintaining all of the Assigned Approvals.

(b) Seller's Obligations.

(i) As soon as practicable on or after the Closing Date and in any event within one (1) week of the Closing Date, Buyer will make all applications, filings and notifications required (or, if deemed advisable by the Parties,

Page 34 of 59

requested by applicable Governmental Authorities) with applicable Governmental Authorities to cause the issuance of new, or the amendment of its existing, Approvals in the name of Buyer as required to enable Buyer to conduct the Radiopharmacy Business and hold the Purchased Assets independently from Seller. Buyer shall diligently pursue such transfers, issuances and amendments, including responding promptly to any requests from Governmental Authorities for further information or documents.

(ii) Until all Assigned Approvals have been transferred into the name of Buyer and Buyer has obtained all of its own Approvals necessary to own and operate the Radiopharmacy Business and Purchased Assets, as applicable:

(A) Seller will use its commercially reasonable efforts to maintain any Scheduled Approvals that have not been transferred to or otherwise obtained by Buyer;

(B) Seller will be free to continue to pursue those on-going variations and amendments to the Scheduled Approvals which are pending at the Closing Date or withdraw them, subject to the prior written approval of the Buyer (not to be unreasonably withheld, conditioned or delayed);

(C) Seller will not be required to initiate any additional variations or amendments to the Scheduled Approvals, except in the event they are indispensable for the continuation of the Radiopharmacy Business and only then upon Buyer's written request; and

(D) Seller does not warrant that it will obtain approval of any variation or amendment that is sought and will not be responsible for any failure in this regard.

(iii) For the avoidance of doubt, Seller does not warrant the successful issuance, maintenance or renewal of any Approval (including any Assigned Approvals) on or after the Closing Date and Seller will not be responsible and will have no liability for any failure to obtain, maintain or renew any Approval (including any Assigned Approval) on or after the Closing Date, in each case, except if the Governmental Authority cancels any of Seller's Approvals or refuses its renewal or issuance as a result of Seller's failure to use commercially reasonable efforts to maintain any Scheduled Approvals that have not been transferred to or otherwise acquired by Buyer (or where Buyer has not obtained its own equivalent Approval) as a result of Seller's gross negligence or willful misconduct.

(iv) Furthermore, Seller is not responsible for conducting any studies, including clinical and stability studies, concerning any Purchased Assets, which may be requested by a Governmental Authority on or after the Closing Date, regardless of whether the transfer of all Assigned Approvals has occurred or not.

Page 35 of 59

(c) <u>Costs and Expenses</u>. From and after the Closing Date, Buyer or its Affiliates will bear all fees levied by the relevant Governmental Authorities or any other Persons and any other relevant out-of-pocket costs for: (i) the issuance, maintenance or renewal of all such Approvals (including the Assigned Approvals) and for the transfer of the Assigned Approvals to Buyer (or its Affiliates) on or after the Closing Date; and (ii) any variations and amendments made pursuant to Sections 6.2(b)(ii) and 6.2(b)(iii) above.

6.3 <u>Consents</u>. Seller and Buyer will use commercially reasonable efforts to give notices to, and obtain consents from, the third parties that are described in Schedule 6.3; provided, however, that Seller will not be obligated to pay any consideration therefor to any third party from whom consent is requested (it being understood that Buyer will be solely responsible for the payment of any such consideration, if any) and Seller will not be responsible for the actions or inactions of any third parties in the process of delivering notices or seeking consents to the Transactions or otherwise. Nothing in this Section 6.3 denigrates or otherwise affects the provisions of Section 2.5.

6.4 <u>Bulk Sales Laws</u>. Each of the Parties hereby waives on behalf of itself and its respective Affiliates compliance with the requirements and provisions of the "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer or sale of any or all of the Purchased Assets to the Buyer, as applicable.

6.5 Employees and Employee Benefits.

(a) Buyer represents, warrants, covenants and agrees that it (or one of its designated Affiliates) has, on or immediately prior to the scheduled Closing Date, (i) offered in writing to employ, effective as of the Closing Date, all Employees (other than Quebec Employees and Key Employees) on terms and conditions that are, in respect of each such Employee, no less favorable in the aggregate than those in effect immediately prior to the Closing Date and set forth on Schedule 4.14(a) and (ii) continued to employ (and confirmed the same in writing) all Quebec Employees (other than Key Employees) on terms and conditions that are, in respect of each such Employee, no less favorable in the aggregate than those in effect immediately prior to the Closing Date and set forth on Schedule 4.14(a) (the writings described in clauses (i) and (ii), collectively, the "<u>Offer Letters</u>"). The Offer Letters are conditional on Closing. Employees who commence employment with the Buyer pursuant to this Section 6.5(a) will be referred to, collectively, as the "<u>Transferred Employees</u>." Notwithstanding the foregoing, an Employee on Leave shall not become a Transferred Employee unless the Employee on Leave becomes a Transferred Employee shall be referred to herein as the "<u>Transfer Date</u>").

(b) Prior to the Closing Date, Buyer and each of the Key Employees shall have entered into an employment agreement on terms and conditions that are, in respect of the Key Employees, substantially similar in the aggregate than those in effect immediately prior to the Closing Date, to be effective and conditional on Closing.

(c) For all individuals who are employed by the Seller in the Radiopharmacy Business who are not Employees (the "Excluded Employees"), the Seller shall, prior to Closing,

Page 36 of 59

either (i) transfer the employment of such Excluded Employee to the Retained Business or an Affiliate of the Seller, or (ii) terminate the employment of such Excluded Employee, and the Seller shall be solely responsible for all costs associated with such transfers and terminations. Seller agrees that it will be entirely responsible for, and pay, perform and discharge and fully satisfy all liabilities for severance pay, notice of termination of employment or pay in lieu of such notice, damages for wrongful or unjust dismissal and any obligations in connection therewith (statutory, contractual, or at civil or common law) or other employee benefits or claims made by (x) any Excluded Employee, and (y) any Employee who does not accept an Offer Letter and/or does not become a Transferred Employee (in the case of this clause (y), other than as a result of Buyer's breach of this Section 6.5).

(d) Buyer will recognize the past service of Transferred Employees with Seller for all purposes required by Law, including any required notice of termination, termination or severance pay (contractual, statutory, at common-law or otherwise under applicable Law).

(e) Buyer will make the Key Employees reasonably available for consultation with Seller with respect to Seller allocating Individual Bonus Amounts. Seller shall pay the Individual Bonus Amounts to eligible Transferred Employees on or before *** and shall pay, withhold and/or remit all income tax, payroll tax and other similar amounts exigible on such cash bonus awards as required under applicable Law. Seller shall disclose the Individual Bonus Amounts to the Buyer at least five business days prior to such payment to eligible Transferred Employees.

(f) The Seller covenants and agrees that the only cash bonuses to which Transferred Employees will at Closing be entitled (regardless of whether or not the relevant cash bonuses are payable on the fulfillment of conditions that have not at the relevant time been satisfied and whether those bonuses become payable before or after Closing) shall be included in the Individual Bonus Amounts calculated by the Seller (provided that, for clarity, any annual bonus with respect to calendar year 2016 shall not be so included).

(g) Subject to Buyer's reimbursement obligations below, Seller will retain all Liability for any compensation that accrues in the single payroll period beginning prior to, and ending after, the Closing Date. For this purpose, any such pay (including all income Tax, payroll Tax and other similar amounts exigible on such payroll payments as required under applicable Law) that relates to the period that begins before and ends after the Closing Date or Transfer Date, as applicable, will be pro-rated based on (i) the portion of the period that fell before the Closing Date or Transfer Date, as applicable, and (ii) the period on and after the Closing Date or Transfer Date, as applicable, for which Buyer will be responsible, and (ii) the period on and after the Closing Date or Transfer Date, as applicable, for which Buyer will be responsible. Seller will, prior to the payroll payment contemplated by this Section 6.5(g) is scheduled to be made, notify and provide Buyer with a schedule of payroll payments to be made by Seller to Transfered Employees (such schedule, the "Payroll Schedule"). Seller shall pay to the Transferred Employees the aggregate amount set out on the Payroll Schedule on the scheduled payroll date. Buyer will then reimburse the Seller for the pro-rata portion of the payments made to Transfered Employees that relates to the period beginning on the Closing Date or Transfer Date, as applicable, and ending on the last day of the payroll period to which the payroll payment relates, in accordance with the Payroll Schedule (including all income Tax, payroll Tax and other similar amounts exigible on such payroll payments as required under applicable Law). Seller shall pay, withhold and/or remit all income Tax, payroll Tax and other similar amounts exigible on such payroll payments as required under applicable Law.

Page 37 of 59

(h) With respect to any Employee Benefit Plan maintained by Buyer or an Affiliate of Buyer (collectively, "<u>Buyer Benefit Plans</u>") for the benefit of any Transferred Employee, Buyer will, or will cause its Affiliate to, recognize all past service of the Transferred Employees with Seller, as if such past service were with Buyer, only to the extent required by Law.

(i) The Seller Benefit Plans shall retain responsibility for and satisfy obligations with respect to all benefit claims incurred by Transferred Employees (and their eligible spouses, beneficiaries and dependents) under the Seller Benefit Plans prior to the Closing Date or Transfer Date, as applicable, are entitled to benefits under the Seller Benefit Plans, in each case, subject to and in accordance with the terms of the applicable Seller Benefit Plan and related insurance policies. For greater certainty, a Transferred Employee who incurred a claim in respect of short-term or long-term disability prior to the Closing Date or Transfer Date, as applicable, as applicable, is entitled to benefits under the Seller's short-term or long-term disability benefit plan, shall continue to be covered by and paid from the Seller Benefit Plans for the duration of such disability.

(j) Subject to Buyer's reimbursement obligations in clause (l) below, the Seller Benefit Plans specified on Schedule 6.5(j) (the "<u>Transition Period</u> <u>Benefit Plans</u>") shall retain responsibility for and satisfy obligations with respect to all benefit claims incurred by Transferred Employees prior to the Benefit Plans Transition Date (the "<u>Covered Employees</u>") and their eligible spouses, beneficiaries and dependents, under the Transition Period Benefit Plans for the period commencing on and after the Closing Date and ending on *** (the "<u>Benefit Plans Transition Date</u>"), in each case, subject to and in accordance with the terms of the applicable Transition Period Benefit Plan, approval by the applicable Insurance Provider and the applicable insurance policies. For greater certainty, subject to Buyer's reimbursement obligations in clause (l) below, a Covered Employee who incurred a claim under the Transition Period Benefit Plans in respect of short-term disability or long-term disability during the period commencing on the Closing Date and ending on the Benefit Plans Transition Date shall be entitled to benefits under the Seller's short-term disability, long-term disability and life insurance plans for the duration of such disability.

(k) The Buyer shall be responsible under the Buyer Benefit Plans for any and all benefit claims incurred by the Covered Employees (and their eligible spouses, beneficiaries and dependents) on and after the Benefit Plans Transition Date.

(1) The Parties agree that the Transition Period Benefit Plans shall continue to provide benefit coverage for the Covered Employees during the period commencing on the Closing Date and ending on the Benefit Plans Transition Date, provided that the Buyer pays all related out-of-pocket costs and expenses incurred by Seller, including the proportion of monthly premiums required by the Seller's benefits and insurance providers (the "<u>Insurance</u> <u>Providers</u>") that relate to the benefits provided to the Transferred Employees during the Transition Period (including all income Tax, payroll Tax and other similar amounts exigible thereon as required

Page 38 of 59

under applicable Law). The Parties understand and agree that the Insurance Providers will provide the Buyer with monthly premium invoices that will reflect monthly changes in insured volumes and headcount of Transferred Employees and premium taxes.

(m) Notwithstanding the foregoing, any Quebec Employee who is on an approved leave of absence on the Benefit Plans Transition Date shall remain a participant in the Seller Benefit Plans that provide disability and life insurance benefits (as applicable) until such Quebec Employee returns to active employment with the Buyer, at which time he or she shall become a participant in the disability and life Buyer Benefit Plans (as applicable).

(n) For purposes of Section this 6.5, a claim for benefits will be deemed to have been incurred, whether or not reported:

(i) with respect to death or dismemberment, on the actual date of death or of dismemberment;

(ii) with respect to short-term and long-term disability, on the date the claimant became disabled as determined in accordance with the applicable plan;

(iii) with respect to all medical, dental or vision claims, on the date a service or supply giving rise to the claim under the applicable benefit plan is purchased or received by the claimant or his/her eligible dependent; and

(iv) where a claim includes more than one service or supply, the incurred date of the claim will be determined in accordance with the guidelines issued by the Canadian Life and Health Insurance Association ("<u>CLHIA</u>") for such claims. Reimbursement will then be made by the applicable Employee Benefit Plan, based on the date on which the claim was incurred, and will be subject to the terms and conditions of the applicable plan.

(o) Buyer and Seller intend that the transactions contemplated by this Agreement will not constitute a separation, termination or severance of employment of any Transferred Employee who accepts an employment offer by Buyer that is consistent with the requirements of this Section 6.5, including for purposes of any Employee Benefit Plan that provides for separation, termination or severance benefits, and that each Transferred Employee will have continuous employment immediately before and immediately after the Closing Date.

(p) Subject to compliance with applicable Law, Seller and Buyer will reasonably cooperate in good faith, and Seller and the Buyer and their respective Affiliates will provide reasonable access to such information as is reasonably necessary or appropriate, to facilitate implementation of the provisions of this Section 6.5.

(q) This Section 6.5 will be binding upon and inure solely to the benefit of each of the Parties, and nothing in this Section 6.5, express or implied, will confer upon any other Person (including any Employees) any rights or remedies of any nature whatsoever under or by reason of this Section 6.5. Nothing contained in this Agreement, express or implied, will be construed to establish, amend or modify any benefit plan, program, agreement or arrangement. The Parties acknowledge and agree that the terms set forth in this Section 6.5 will not create any right in any Employee or any other Person to any continued employment with Buyer or any of its Affiliates or compensation or benefits of any nature or kind whatsoever.

Page 39 of 59

6.6 <u>Further Assurances</u>. Each of Buyer and Seller will, at the request of the other Party and at such other Party's expense, promptly execute and deliver to such other Party all such further instruments, assignments, filings and other documents, and do all such things, as such other Party may reasonably request in connection with the carrying out and effectuating the Transaction Documents and the Transactions.

6.7 <u>Retention of Books and Records and Information</u>. Seller and Buyer agree that each will preserve and keep the records held by it relating to the Purchased Assets for a period of *** (***) years from the Closing Date, or any longer retention period prescribed by applicable Laws, and will make such records and personnel available to the other as may be reasonably required by such Party in connection with, among other things, any insurance claims by, legal or regulatory compliance requirements of, Legal Proceedings or Tax audits against, or Governmental Authority investigations of, Seller or Buyer or in order to enable Seller or Buyer to comply with their respective obligations under the Transaction Documents, provided, however, that all such records relating to Tax matters for any given year will be retained until the later of (i) the required retention period prescribed by any Tax law for such fiscal period, and (ii) the expiry of all Tax reassessment periods for such fiscal period.

6.8 Non-Compete and Non-Solicit.

(a) Seller and its Affiliates will be permitted to conduct the Retained Business without limitation, except as otherwise set forth in this Section 6.8. Neither Seller nor any of its Affiliates will, for a period commencing on the Closing Date and ending *** (the "<u>Restricted Period</u>"), directly or indirectly, individually, in partnership, jointly or in collaboration with any Person:

(i) open, own, acquire or operate (A) a radiopharmacy business in the Territory that prepares, sells or distributes (1) individual, patientready unit doses (including bulk unit doses and iodine-based products) of SPECT-based radiopharmaceuticals to healthcare providers that administer those unit doses to patients, or (2) subject to the exception in Section 6.8(a)(V) below, the specific third party SPECT-based cold kits set forth in Schedule 1.1(a), or (B) a business in the Territory that manufactures, or has manufactured by third parties, FDG or that prepares, sells or distributes FDG (the businesses described in clauses (A) and (B), a "<u>Competing Business</u>"),

(ii) teach, instruct or assist any Person to open, acquire, establish, operate, develop, reorganize or re-establish a Competing Business,

(iii) have an equity or profit interest in any Person that materially engages in any Competing Business (in each case, other than up to a one percent interest in a publicly traded company engaging in a Competing Business),

(iv) lend money to any Person that materially engages in any Competing Business, or

(v) participate in, be awarded Contracts pursuant to, or otherwise perform Contracts awarded pursuant to, any customer supply contract tender process that, throughout the duration of that tender process, only seeks bids for unit doses (including bulk unit doses and iodine-based products) of SPECT-based radiopharmaceuticals or FDG.

Page 40 of 59

For the avoidance of doubt, nothing in this Agreement prohibits or restricts Seller or its Affiliates from:

- (I) offering to sell, selling or distributing to any Person (including any Person conducting a Competing Business, but excluding any Buyer Customer solely to the extent provided in clause (b) below) any bulk Products (as defined in the Supply Agreement) or any other competing, bulk SPECT-based radiopharmaceuticals manufactured, sold or distributed by or for Seller or any of its Affiliates (collectively, "<u>Bulk SPECT Products</u>");
- (II) teaching, instructing or assisting any Person (including any Buyer Customer or any Person conducting a Competing Business) in actually preparing or administering unit doses from Bulk SPECT Products or from any PET-based radiopharmaceuticals (other than FDG) or non-radiopharmaceutical products manufactured, sold or distributed by or for Seller or any of its Affiliates (with respect to Bulk SPECT Products, in accordance with the package inserts included with such products, but only to the extent arising out of specific customer inquiries relating to actually preparing or administering unit doses of such products, and excluding teaching, instructing or assisting any Person in conducting or operating a Competing Business generally);
- (III) manufacturing or having manufactured by any Person (including any Buyer Customer or Person conducting a Competing Business) any PET-based radiopharmaceuticals (other than FDG) or non-radiopharmaceutical products;
- (IV) offering to sell, selling or distributing to any Person (including any Buyer Customer or Person conducting a Competing Business) any PET-based radiopharmaceuticals (other than FDG) or non-radiopharmaceutical products manufactured, sold or distributed by or for Seller or any of its Affiliates; or
- (V) offering to sell, selling or distributing the specific third party SPECT-based cold kits set forth in Schedule 1.1(a) to

Page 41 of 59

(1) Bulk SPECT Product customers in accordance with its obligations under any of its customer contracts that existed as of the Closing Date in accordance with their respective terms and conditions (including the pricing and pricing adjustment provisions, volume and volume adjustment provisions, contract term and renewal options in favor of the customer, in the case of each of the provisions described in this parenthetical, as those provisions exist as of the Closing Date) for the term of those contracts (including any renewal terms exercised by the applicable customers, in each case, pursuant to the renewal provisions existing as of the Closing Date), or (2) any other Persons that are not Buyer Customers.

(b) During the period commencing on the Closing Date and ending on *** (the "<u>Non-Solicitation Period</u>"), Seller and its Affiliates will not actively solicit any SPECT unit dose (including bulk unit dose) customers of the Radiopharmacy Business or of the Buyer's radiopharmacy business, in each case, in the Territory as of the Closing Date (collectively, "<u>Buyer Customers</u>") to purchase from Seller or any of its Affiliates any Bulk SPECT Products or FDG; provided, however, that:

- (A) Seller and its Affiliates will be permitted to sell and distribute any Bulk SPECT Products to (i) Buyer Customers that initiate discussions with Seller without Seller's active solicitation, (ii) Buyer Customers that are also Bulk SPECT Product customers of Seller only for purposes of supplying Bulk SPECT Products for weekend or overnight use at current levels; provided, however, Seller and its Affiliates may sell additional volumes requested by such Buyer Customers without Seller's active solicitation and (iii) any other Person that is not a Buyer Customer, including any Person conducting a Competing Business (and Seller may actively solicit such Persons);
- (B) Seller agrees that it shall give reasonably prompt written notice to Buyer (to the extent it is not restricted under applicable Laws or obligations of confidentiality) if any of the following events occur from time to time: (i) any Person solicits help from Seller to, or expressly indicates its intent to Seller to, engage in a Competing Business or (ii) if any Buyer Customer increases its volumes of purchases of Bulk SPECT Products from the Retained Business by more than \$*** in any calendar quarter; and
- (C) for clarity, responding to, participating in, being awarded one or more Contracts pursuant to, and otherwise performing Contracts awarded pursuant to, any customer supply contract tender process that seeks bids for bulk radiopharmaceuticals (including Generators, as defined in the Supply Agreement) does not

Page 42 of 59

constitute active solicitation for purposes of this Section 6.8. However, during the Non-Solicitation Period, Seller and its Affiliates shall not ***.

(c) During the Non-Solicitation Period, neither Buyer nor any of its Affiliates will sell any Bulk SPECT Products manufactured, sold or distributed by or for any Person to any customers of the Seller's Retained Business as of the Closing Date.

(d) The Parties agree that the foregoing restrictive covenants are an integral part of this Agreement and those covenants are granted to maintain or preserve the fair market value of the Purchased Assets acquired by Buyer under this Agreement, on the one hand, and the Retained Business of Seller, on the other hand. The Parties further acknowledge and confirm that no amount of the Purchase Price is received or receivable by any Party for granting any of its covenants contained in this Agreement.

6.9 <u>Non-Solicitation and No Hire</u>. During the Restricted Period, Seller will not, directly or indirectly, (a) solicit for employment or hire any person who is an Employee immediately prior to the Closing Date and who becomes a Transferred Employee or (b) induce or attempt to influence any such person to terminate his or her employment with Buyer; <u>provided</u> that the foregoing will not apply to the solicitation or hiring of any Person (i) via general solicitations not targeted at such employees or (ii) who is terminated from his or her employment with Buyer or who otherwise is no longer, and for the prior ******* (*******) months has not been, employed by Buyer.

6.10 Investment Canada Act. Buyer will complete and file a notice of the Transactions following the Closing Date and in the form and within the time period prescribed by Section 12 of the Investment Canada Act.

6.11 Certain Intellectual Property Matters.

(a) Seller hereby grants an exclusive in the Territory, fully paid and assignable license to use solely in the Territory in the conduct of Buyer's radiopharmacy business (including the Radiopharmacy Business) the Intellectual Property (including all standard operating procedures, trade secrets and know-how) that is owned by the Seller at the Closing Time solely to the extent it is (i) protected, created or arising under the Laws of the Territory, (ii) is actually used in the conduct of the Radiopharmacy Business as of the Closing Date, (iii) used primarily for purposes of conducting the Radiopharmacy Business or the other radiopharmacy businesses of the Seller and its Affiliates in Australia and Puerto Rico, and (iv) except with respect to Gludef®, it relates only to radiopharmacy preparation of patient-ready unit doses of radiopharmaceuticals to healthcare providers (for administration to patients by those healthcare providers) and is not otherwise Intellectual Property used to make or prepare, have made or prepared or otherwise used in any proprietary radiopharmaceutical of Seller. Seller (and its assignees) will keep all such Intellectual Property confidential and will not disclose such Intellectual Property to any third parties (other than assignees thereof).

(b) Buyer will upon Closing cease using the "Lantheus" name (and all derivations thereof) and any related Marks and revise marketing, labeling and other literature to

Page 43 of 59

(a) delete references thereto and (b) all references to Seller or any of its Affiliate's customer service addresses or telephone numbers; provided that, for up to *** (***) months after the Closing Date, Buyer may continue to distribute product literature and labeling and use customer interfaces that use any such names, Marks, addresses or telephone numbers to the extent existing on the Closing Date and, in respect of tangible property, included in Inventory.

6.12 <u>Accounts Receivable</u>. From and after the Closing Time, (a) if the Seller or its Affiliates receive or collect any Accounts Receivable, the Seller or such Affiliate, as applicable, will remit any such amounts to Buyer, for the first three months after the Closing Date, on a weekly basis, and thereafter, on a monthly basis, and (b) if the Buyer or its Affiliates receive or collect any amounts in respect of Excluded Assets, the Buyer or such Affiliate, as applicable, will remit any such amounts to Seller, for the first three months after the Closing Date, on a weekly basis, and thereafter, on a monthly basis; provided that, for the sake of convenience, the Parties may with mutual consent net such amounts against each other.

6.13 <u>Transition Matters</u>. Buyer will develop and implement a transition for the Radiopharmacy Business from the services specified in the Transition Services Agreement to a standalone basis as soon as reasonably practicable after Closing (and, in any event, no later than ******* (*******) days after the Closing Date, unless explicitly specified otherwise in the Transition Services Agreement) and in a manner that is otherwise consistent with the terms and conditions of the Transition Services Agreement. Each of the Parties agrees that it will reasonably cooperate with the other Party in good faith in implementing that transition in a timely manner in accordance with the terms of this Section 6.13 and the Transition Services Agreement. Each of the Parties will make available to the other Party all data (including, for greater certainty, all personnel files) and other information as may be reasonably required by such Party in connection with any legal or regulatory compliance requirements. As soon as practical after Closing (and, in any event, no later than the applicable date specified on Schedule 6.13), Seller shall deliver or cause to be delivered to Buyer all Records included in the Purchased Assets, and Seller will reasonably cooperate with Buyer to provide to Buyer such Records as Buyer may reasonably request to be provided on an expedited basis as set forth Schedule 6.13. The Seller covenants that *******. Not in limitation of any of the foregoing, by no later than *******, Buyer will reasonably cooperate with Seller and provide to Seller, at Buyer's expense, all data from the Radiopharmacy Business (including data on dosimetry and inventory of sealed and unsealed sources for the 2015 calendar year) that Seller requests in writing in order for Seller to complete and submit its annual compliance report to CNSC in a timely manner.

6.14 <u>Collaborator Meetings</u>. Within *** (***) days following the Closing Date, Seller shall arrange for Buyer and its Representatives to meet with, at a mutually convenient time, representatives of customers specified on Schedule 6.14 having relevant decision-making authority.

ARTICLE 7

INDEMNIFICATION

7.1 Indemnification by Seller. Subject to all of the limitations set forth in this Article 7, from and after Closing, Seller agrees to indemnify, defend and hold Buyer, its

Page 44 of 59

Affiliates and each of their respective directors, officers, employees, agents, attomeys, representatives, successors and permitted assigns (Buyer and such Persons are collectively hereinafter referred to as "<u>Buyer's Indemnified Persons</u>"), harmless from and against any and all loss, liability, damage or deficiency, including interest, penalties, reasonable costs of preparation and investigation, and reasonable attorneys' fees and disbursements (individually a "Loss," and collectively, "Losses") that Buyer's Indemnified Persons may suffer, sustain, incur or become subject to, to the extent arising out of or due to: (a) any inaccuracy, misrepresentation or breach of any representation or warranty made or given by the Seller in Article 4 of this Agreement as of the Closing Date (or any other date specified in such representation or warranty); (b) from and after Closing, the breach of any covenant, undertaking, agreement or other obligation of Seller under this Agreement; (c) any Excluded Asset or Excluded Liability (excluding, for such purposes, all assets and liabilities intended to be transferred or assumed as contemplated in Section 2.5); (d) in respect of any Employee who is not a Transferred Employee (except if Buyer breached its obligations in Section 6.5 with respect to such Employee); (e) the application of the "bulk-sale" or "bulk-transfer" Laws of any jurisdiction in Canada by creditors of Seller's Retained Business with respect to the purchase and sale of the Purchased Assets pursuant to this Agreement (for the sake of clarity, in each case, except to the extent the Loss arises from any failure of Buyer to fully satisfy the Assumed Liabilities); or (f) any failure by Seller to comply with section 6 of the *Retail Sales Tax Act* (Ontario) or any comparable Laws of any other jurisdiction in Canada in respect of the purchase and sale of the Purchased Assets pursuant to this Agreement.

7.2 Indemnification by Buyer. Subject to all of the limitations set forth in this Article 7, from and after Closing (except with respect to clause (d), which shall apply from and after the Execution Date), Buyer agrees to indemnify, defend and hold Seller, its Affiliates and each of their respective directors, officers, employees, controlling Persons, agents, attorneys, representatives, successors and permitted assigns (Seller and such Persons are hereinafter collectively referred to as "Seller's Indemnified Persons"), harmless from and against any and all Losses that Seller's Indemnified Persons may, suffer, sustain, incur or become subject to, to the extent arising out of, or due to: (a) any inaccuracy, misrepresentation or breach of any representation or warranty made or given by the Seller in Article 5 of this Agreement as of the Closing Date (or any other date specified in such representation or warranty); (b) from and after Closing, the breach of any covenant, undertaking, agreement or other obligation of Buyer under this Agreement; (c) any Purchased Asset or Assumed Liability (including, for such purposes, all assets and Liabilities intended to be transferred or assumed as contemplated in Section 2.5), or Transferred Employee; (d) any matters (whether before or after the Closing Date) arising from or relating to the application or enforcement of, or claims brought under, antitrust/competition Laws or the Investment Canada Act (including the matters contemplated by Sections 6.1(a) and (b)), including the costs of defending the Transactions, costs associated with any Legal Proceeding (including investigations and inquiries, whether formal or informal), or Losses that Seller's Indemnified Persons may suffer or incur arising in connection with any related Order; (e) Seller's or Buyer's performance of its respective obligations under Section 2.5 or under any Contract treated as an Assigned Contract pursuant to Section 2.5 or the sale, assignment, transfer, conveyance and delivery of any of the Assigned Contracts (including any treated as Assigned Contracts pursuant to Section 2.5) to the extent that any consent, authorization, approval or waiver described in Section 2.5 with respect to the applicable Assigned Contract has not been obtained at Closing (except to the extent Seller has breached its obligations to disclose in the

Page 45 of 59

Schedules the requirement to obtain any such consent, authorization, approval or waiver and Seller incurs actual economic harm as a result of not obtaining consent under that applicable Assigned Contract or the Seller has failed to comply with its covenants in Section 2.5, 6.2 or 6.3 with respect to such consent, authorization, approval or waiver); or (f) any of the Agreements specified in Schedule 7.2(f), in each case only to the extent such Losses relate to the period from and after the Closing Date.

7.3 Survival of Representations and Warranties; Limitations.

(a) The representations and warranties of the Parties contained in this Agreement will survive the Closing Date and will remain in full force and effect thereafter for a period of *** (***) months and will be effective with respect to any claimed breach of the representations and warranties timely made pursuant to Section 7.4, after which period the representations and warranties of the Parties contained in this Agreement will terminate and be of no further force or effect. Notwithstanding the foregoing, the representations and warranties set forth in Section 4.1 (Organization and Corporate Power), Section 4.2 (Due Authorization), Section 4.3(a)(iii) (No Violation; Consents), Section 4.9 (Good and Valid Title), Section 4.16 (Tax Matters), Section 5.8 (Broker's Fees), Section 5.1 (Organization and Corporate Power), Section 5.2 (Due Authorization), 5.3(a)(iii) (No Violation; Consents) and Section 5.8 (Broker's Fees) will survive indefinitely (collectively, the "<u>Fundamental Representations</u>"); provided that Section 4.16 (Tax Matters) will survive only until the applicable statute of limitations has run.

(b) Notwithstanding anything to the contrary in this Agreement, no Indemnified Person will be entitled to any recovery from an Indemnifying Party with respect to any breach of such representations and warranties unless and until the amount of such Losses suffered, sustained or incurred by the asserting Party, or to which such Party becomes subject, by reason of such breach, will exceed *** U.S. Dollars (US\$***) calculated on a cumulative basis and not a per item basis (the "<u>Basket Amount</u>") and, in such event, the indemnifying party will be required to pay the full amount of such Losses including the Basket Amount. In no event will either Party be liable to the other Party under this Agreement in an aggregate amount in excess of *** Dollars (\$***), or in the case of a breach of Section 6.8 by the Seller or its Affiliates, ***Dollars (\$***) (in each case, the "<u>Cap</u>"), except that the Basket Amount and the Cap will not be applicable to (i) any breach of any Fundamental Representations, (ii) Losses indemnifiable under any of Sections 7.1(b)-(f) or 7.2(b)-(f) or (iii) Losses based on fraud of the Indemnifying Party; provided that, except in the case of Losses based on fraud of the Indemnifying Party; in no event will either Party be liable to the other Party under this Agreement in an aggregate amount in excess of *** Dollars (\$***). For purposes of determining the amount of any Losses under this Article 7, each representation and warranty will be read without reference to any materiality or Material Adverse Effect qualification contained therein (but for the avoidance of doubt, a materiality or a Material Adverse Effect qualification will be used for determining whether a breach occurred and for determining whether a material item was explicitly required to be listed or disclosed in a Schedule).

(c) Seller will not be required to indemnify any Buyer's Indemnified Persons, and Buyer will not be required to indemnify any Seller's Indemnified Persons, to the extent of any Losses that any court of competent jurisdiction will have determined by final judgment to have resulted from the bad faith, fraud, gross negligence or willful misconduct of any of the Buyer's Indemnified Persons or Seller's Indemnified Persons, respectively.

Page 46 of 59

(d) Notwithstanding anything to the contrary in this Agreement, no breach of any representation, warranty, covenant or agreement contained in this Agreement will give rise to any right on the part of Party or any Indemnified Person, after the consummation of the Transactions, to rescind this Agreement or any of the Transactions.

(e) Any liability for indemnification under this Article 7 will be determined without duplication of recovery by reason of the state of facts giving rise to such liability constituting a breach of more than one representation, warranty, covenant or agreement.

(f) Notwithstanding any other provision in this Agreement, neither Seller nor Buyer will in any event be liable to the other Party or any of the other Party's Indemnified Persons on account of any indemnity obligation set forth in this Article 7 for any special, incidental, treble or punitive damages, in each case, unless such Losses are paid pursuant to a third party claim.

7.4 Indemnification Procedure.

(a) A claim for indemnification for any matter not involving a third-party claim may be asserted by notice issued in accordance with Section 10.3 to the Party from whom indemnification is sought. The failure by any Indemnified Person so to notify the Indemnifying Party will not relieve the Indemnifying Party from any liability that it may have to such Indemnified Person with respect to any such claim, except to the extent that the Indemnifying Party is materially prejudiced as a result of such failure, it being understood that notices in respect of a breach of a representation or warranty must be delivered prior to the expiration of the survival period for such representation or warranty. In the event the Indemnifying Party does not notify the Indemnified Person within *** (***) days following its receipt of such Notice of Claim that the Indemnifying Party disputes its liability to the Indemnified Person under this Article 7 or the amount thereof, the claim specified by the Indemnified Person in such notice will be conclusively deemed a Loss of the Indemnifying Party under this Article 7, and the Indemnifying Party will pay the amount of the Losses relating to such claim to the Indemnified Person on demand or, in the case of any notice in which the amount of Losses related to such the claim (or any portion of the claim) is estimated, on such later date when the amount of such claim as provided above, as promptly as reasonably practicable, such Indemnified Person and the appropriate Indemnifying Party will east the Indemnifying Party will east the Indemnifying Party will east the Indemnifying Party will pay to the Indemnifying Party will east the Indemnifying Party will east the Losses related to such claim, where applicable, the Indemnifying Party will pay to the Indemnifying Party will east the Indemnifying Party will east the Losses related to such claim, where applicable, the Indemnifying Party will pay to the Indemnifying Party will east the such claim as mount of the Losses related to such claim, where app

(b) In the event that any Legal Proceeding is instituted, or that any claim is asserted, by any third party in respect of which payment may be sought under Article 7 (regardless of the limitations set forth in Section 7.3) (an "<u>Indemnification Claim</u>"), the

Page 47 of 59

Indemnified Person will promptly cause written notice of the assertion of any Indemnification Claim of which it has knowledge that is covered by this indemnity to be forwarded to the Party against whom indemnification is sought (the "Indemnifying Party"). The failure of the Indemnified Person to give reasonably prompt notice of any Indemnification Claim will not release, waive or otherwise affect the Indemnifying Party's obligations with respect thereto, except to the extent that the Indemnifying Party is materially prejudiced as a result of such failure. The Indemnifying Party will have the right, at its sole option and expense, to be represented by counsel of its choice and to defend against, negotiate, settle or otherwise deal with, any Indemnification Claim that relates to any Losses indemnified by it under this Agreement. If the Indemnifying Party elects to defend against, negotiate, settle or otherwise deal with any Indemnification Claim that relates to any Losses indemnified by it under this Agreement, it will, within thirty (30) days (or sooner, if the nature of the Indemnification Claim so requires), notify the Indemnified Person of its intent to do so. If the Indemnifying Party elects not to defend against, negotiate, settle or otherwise deal with, any Indemnification Claim that relates to any Losses indemnified by it under this Agreement, the Indemnified Person may defend against, negotiate, settle or otherwise deal with, such Indemnification Claim. If the Indemnifying Party assumes the defense of any Indemnification Claim, the Indemnified Person may participate, at his or its own expense, in the defense of such Indemnification Claim; provided, however, that such Indemnified Person will be entitled to participate in any such defense with separate counsel at the expense of the Indemnifying Party if (i) so requested by the Indemnifying Party to participate, (ii) in the reasonable opinion of counsel to the Indemnified Person, a conflict or potential conflict exists between the Indemnified Person and the Indemnifying Party that would make such separate representation advisable or (iii) the Indemnifying Person fails to prosecute such defense actively and diligently; and provided, further, that the Indemnifying Party will not be required to pay for more than one such counsel (plus any appropriate local counsel) for all Indemnified Persons in connection with any Indemnification Claim. The Parties agree to cooperate fully with each other in connection with the defense, negotiation or settlement of any such Indemnification Claim. Notwithstanding anything in this Section 7.4 to the contrary, neither the Indemnifying Party nor any Indemnified Person will, without the written consent of the other Party, settle or compromise any Indemnification Claim or permit a default or consent to entry of any judgment. If the Indemnifying Party makes any payment on any Indemnification Claim, the Indemnifying Party will be subrogated, to the extent of such payment, to all rights and remedies of the Indemnified Persons to any insurance benefits or other claims of the Indemnified Person with respect to such Indemnification Claim for payment thereof.

(c) After any final decision, judgment or award will have been rendered by a Governmental Authority of competent jurisdiction and the expiration of the time in which to appeal therefrom, or a settlement will have been consummated, or the Indemnified Persons and the Indemnifying Party will have arrived at a mutually binding agreement with respect to an Indemnification Claim under this Agreement, the Indemnified Persons will forward to the Indemnifying Party pursuant to this Agreement with respect to such matter.

7.5 <u>Character of Payments</u>. To the extent permitted by applicable Law, the Parties agree that any indemnification payments (and/or payments or adjustments) made with respect to this Agreement will be treated for all Tax purposes as an adjustment to the Purchase Price.

Page 48 of 59

7.6 <u>Calculation of Losses</u>. The amount of any Losses for which indemnification is provided under this Article 7 will be net of any amounts actually recovered by the Indemnified Person under insurance policies with respect to such Losses (net of any Tax or expenses incurred in connection with such recovery). Each Indemnified Person will take, and will cause its Affiliates to take, all commercially reasonable efforts to mitigate and otherwise minimize the Losses to the maximum extent reasonably possible upon, and after becoming aware of, any event which would reasonably be expected to give rise to any Losses. Each Indemnified Person will use commercially reasonable efforts to collect any amounts available under insurance coverage, or from any other Person alleged to be responsible, for any Losses to the same extent that such Indemnified Person would if such Loss were not subject to indemnification under this Agreement.

7.7 <u>Tax Benefits</u>. The obligation of indemnification will be reduced to the extent the Indemnified Person is entitled to any Tax benefits, including the reduction of any income Tax by reason of the Losses being claimed as a deduction or reduction of the income Tax payable after taking into account any Tax consequences arising as a result of the receipt of the indemnification payment.

7.8 <u>Minimize Tax.</u> If an indemnity payment would otherwise be included in the Indemnified Person's income, the Indemnified Person covenants and agrees to make all such elections and take such actions as are available, acting reasonably, to minimize or eliminate Taxes with respect to the indemnity payment.

7.9 <u>Appeal.</u> In addition to the right provided to the Indemnifying Party to participate in or assume control of the defense of a third party claim as provided in this Article 7, the Indemnified Person will not permit any right of appeal in respect of any third party claim to terminate without giving the Indemnifying Party notice thereof and an opportunity to contest such third party claim.

Disclosure. The Indemnified Party will not be entitled to indemnification in respect of any Loss to the extent that this Agreement or the Schedules set forth an exception to the representation or warranty which gave rise to the claim is made in this Agreement or in the Schedules. A disclosure set forth in any section of the Schedules numbered to correspond with any section of the Agreement shall apply to that section only.

7.11 Provisions. The amount of Losses will be reduced by any allowance, provision or reserve in respect of and to the extent the matter giving rise to such claim was included in the Closing Working Capital Statement.

7.12 Exclusive Remedy. From and after the Closing, the sole and exclusive remedies for (a) any breach or failure to be true and correct, or alleged breach or failure to be true and correct, of any representation or warranty in this Agreement will be indemnification in accordance with this Article 7, or (b) any breach, or alleged breach, of any covenant or agreement in this Agreement (other than the covenants set forth in Sections 6.1(a)-(b), 6.8 and 6.9 and Article 8) will be indemnification in accordance with this Article 7 and specific performance, injunction or other equitable relief; provided, however, that no Party will be deemed to have waived any rights, claims, causes of action or remedies if and to the extent that (i) such rights, claims, causes of action or remedies may not legally be waived under applicable Law or (ii) such Party proves the other Party's fraud.

Page 49 of 59

7.13 <u>Right of Set-Off</u>. Except as provided in Section 6.12, the Parties expressly agree that Buyer will perform its obligations under this Agreement without setoff, deduction, recoupment or withholding of any kind for amounts owed or payable by Seller, whether under any of the Transaction Documents, applicable Law or otherwise and whether relating to Seller's breach, insolvency or otherwise, in each case, unless such amounts owed or payable are definitively resolved in Buyer's favor by mutual written agreement or by a final and binding Order or arbitration decision. Without limiting the foregoing, the Buyer shall be entitled to deduct and set off from any amounts due and owing to the Seller under the Transaction Documents any amounts owed by the Seller to the Buyer pursuant to Article 7 hereof that have been definitively resolved by mutual written agreement or by a final and binding Order or arbitration decision.

ARTICLE 8

TAX MATTERS

8.1 Cooperation on Tax Matters.

(a) Notwithstanding anything in Section 6.7 to the contrary, Seller and Buyer agree to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information (including access to Records) relating to the Radiopharmacy Business, the Purchased Assets and the Assumed Liabilities as is reasonably necessary for the filing of any Tax Return, the preparation for any Tax audit, or the prosecution or defense of any claim relating to any proposed Tax adjustment; provided that in no event will any Party be required to provide any Tax Returns to the other Party. Buyer and Seller will keep all such information and documents received by them confidential in accordance with Article 9.

(b) Notwithstanding anything in this Agreement to the contrary, Buyer and Seller will reasonably cooperate with each other in the conduct of any audit or other proceedings relating to the Radiopharmacy Business and the Purchased Assets or the Assumed Liabilities.

8.2 Transfer Taxes.

(a) All GST/HST, QST sales, use and similar transfer Taxes, including but not limited to any value added, gross receipts, stamp duty and personal or intangible property transfer Taxes and other such Taxes and any conveyance fees or recording charges due by reason of the transfer of the Radiopharmacy Business, Purchased Assets, including but not limited to any interest or penalties in respect thereof (the "<u>Transfer Taxes</u>") will be paid by Buyer in addition to the Purchase Price. If Seller is required by Law or by administration thereof to collect any applicable Transfer Taxes from Buyer, Buyer will pay such Transfer Taxes to Seller concurrently with the payment of any consideration payable pursuant to this Agreement, unless Buyer qualifies for an exemption from any such applicable Transfer Taxes to Seller, deliver to Seller such certificates, elections or other documentation as may be required by Law or administration thereof to substantiate and effect the exemption claimed by Buyer.

Page 50 of 59

(b) In the event an amount payable by either Party to the other pursuant to this Agreement is not consideration for a taxable supply pursuant to the ETA and/or the QSTA but rather is deemed to include an amount of GST/HST and/or QST in accordance with section 182 of the ETA and/or section 318 of the QSTA, such amount will be increased to take into consideration the GST/HST and/or QST deemed to be included in the payment, such that the net amount received by the recipient, after remittance of such GST/HST and/or QST to the appropriate Governmental Authority, is equal to the amount that would have been received by the recipient if no GST/HST and/or QST were deemed to be included.

8.3 Tax Elections.

(a) If applicable, Buyer and Seller will jointly execute the elections provided for in subsection 167(1) of the ETA and section 75 of the QSTA to have subsection 167(1.1) of the ETA and section 75.1 of the QSTA apply so that no GST/HST or QST will be payable by reason of the sale of the Purchased Assets. Buyer will file such elections in the manner and within the time prescribed by the ETA and the QSTA. Buyer will, at all times, indemnify and hold harmless Seller, its directors, officers, and employees against and in respect of any and all amounts, including interest and penalties, assessed by a Governmental Authority determining, for any reason, that an election contemplated in this Section 8.3 is inapplicable, invalid, or not properly made, as well as all legal and professional fees incurred by Seller, its directors, officers, and employees as a consequence of, or in relation to, any such assessment. This indemnity will survive and continue in full force and effect until 60 days after the expiration of the period during which any Tax assessment may be issued by a Governmental Authority in respect of such particular liability for Taxes.

(b) Buyer and Seller will, if applicable, jointly execute and file an election under subsection 20(24) of the Tax Act in the manner required by subsection 20(25) of the Tax Act and under the equivalent or corresponding provisions of any other applicable provincial or territorial statute, in the prescribed forms and within the time period permitted under the Tax Act and under any other applicable provincial or territorial statute, as to such amount paid by Buyer to Seller for assuming future obligations. In this regard, Buyer and Seller acknowledge that a portion of the Purchased Assets transferred by Seller pursuant to this Agreement and having a value equal to the amount elected under subsection 20(24) of the Tax Act and the equivalent provisions of any applicable provincial or territorial statute, is being transferred by Seller as a payment for the assumption of such future obligations by Buyer.

(c) Buyer and Seller will each execute and file a joint election under section 22 of the Tax Act and the corresponding provisions of any applicable provincial taxing statute or regulation, within the prescribed time periods, in respect of the accounts receivable to which such election may apply and will designate in such election an amount equal to the portion of the Purchase Price allocated to such accounts receivable as the consideration of and by Buyer therefor.

Page 51 of 59

ARTICLE 9

CONFIDENTIALITY

9.1 Confidential Information.

(a) Each of the Parties (the "<u>Receiving Party</u>") acknowledges its possession of Confidential Information (as defined below) of the other Party (the "<u>Disclosing Party</u>"). From and after the Closing Date, the Receiving Party will not, and will cause its Representatives not to, directly or indirectly, disclose, reveal, divulge or communicate the Confidential Information of the Disclosing Party to any third Person other than the Receiving Party's Representatives. The Receiving Party and its Representatives will use the same degree of care to prevent and restrain the unauthorized use or disclosure of the Confidential Information as the Receiving Party and such Representatives, respectively, currently uses for its own Confidential Information of a like nature, but in no event less than a commercially reasonable standard of care. "<u>Confidential Information</u>" of the Disclosing Party means any confidential or proprietary information, data, material or documents, including information, data, material or documents that, (a) with respect to the Buyer as the Disclosing Party, primarily pertains to the Radiopharmacy Business, the Purchased Assets and/or the Assumed Liabilities, (b) with respect to Seller as the Disclosing Party, pertains to the Excluded Assets and/or the Excluded Business and/or financial information of the Seller or, (c) with respect to either Party as the Disclosing Party, pertains to the Transaction Documents and/or the Transactions and/or otherwise disclosed under the Transaction Documents, in each case, irrespective of the form of communication, and all notes, analyses, compilations, forecasts, data, material or documents.

(b) Notwithstanding the foregoing, "<u>Confidential Information</u>" does <u>not</u> include, and there will be no obligation under this Agreement with respect to, information, data, material or documents that (i) are or become generally available to the public, other than as a result of a disclosure by the Receiving Party or its Representatives in breach of this Agreement, (ii) was available to the Receiving Party or its Representatives on a non-confidential basis prior to disclosure by the Disclosing Party or its Representatives, (iii) becomes available to the Receiving Party or its Representatives on a non-confidential basis from a Person who, to the Receiving Party's or such Representative's knowledge, is not bound by a confidentiality agreement with the Disclosing Party or (iv) was or is independently developed by the Receiving Party or its Representatives without use of the Confidential Information of the Disclosing Party.

(c) For the sake of clarity, nothing in this Agreement will impose any obligation on Seller as the Receiving Party and its Representatives with respect to their disclosure for Seller's own legitimate business or other purposes of information, data, material or documents that pertains (but does not exclusively pertain) to the Radiopharmacy Business, the Purchased Assets or the Assumed Liabilities.

9.2 Disclosure Required by Law. If the Receiving Party or any of its Representatives is required by any Governmental Authority (whether by oral question, interrogatories, requests for information or documents, subpoena, civil investigative demand or similar process) or

Page 52 of 59

pursuant to applicable Law to disclose or provide any Confidential Information, the Receiving Party and its Representatives will use commercially reasonable efforts to provide (if not prohibited by such applicable Law or Governmental Authority) the Disclosing Party with written notice of such request or demand as promptly as practicable under the circumstances so that the Disclosing Party will have an opportunity to seek an appropriate protective Order. The Receiving Party agrees to take, and cause its Representatives to take, at the Disclosing Party's expense, all other commercially reasonable steps necessary to obtain confidential treatment. Subject to the foregoing, the Receiving Party and its Representatives may thereafter disclose or provide any such Confidential Information, as the case may be, to the extent required by such Law (as so advised by counsel) or such Governmental Authority.

9.3 <u>Publicity</u>. Neither Party will issue any publication, press release or other public announcement relating to the Transaction Documents or the Transactions without the other Party's prior written consent, unless such disclosure is required by Law (in which case, the Party required to make such disclosure will, in addition to complying with Section 9.2 and subject to any restrictions of applicable Law, give reasonable advance notice of any such legally required disclosure to the other Party, and such other Party may provide any comments on the proposed disclosure, and such disclosing Party will use its good faith efforts to consider any reasonable comments that such other Party may have with respect thereto). The Parties will reasonably cooperate, each at its own expense, in such disclosure, filing or registration, including such confidential treatment request, and will execute all documents reasonably required in connection therewith.

9.4 <u>Permitted Disclosure</u>. Notwithstanding anything to the contrary in this Article 9, the approval by the other Party will be unnecessary if such disclosure is necessary, as in the reasonable opinion of the Disclosing Party's counsel, in order to implement the provisions of this Agreement.

9.5 Duration of Confidentiality Obligations. This Article 9 will survive for a period of *** (***) years following the Closing Date.

ARTICLE 10

MISCELLANEOUS

10.1 Expenses. Except as specifically provided in this Agreement, Seller and Buyer will each pay its own expenses (including the fees and expenses of their respective agents, representatives, counsel and accountants) incidental to the preparation, negotiation, and consummation of the Transaction Documents and the Transactions.

10.2 <u>Specific Performance</u>. The Parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms of this Agreement and that the Parties will be entitled to specific performance of the terms of this Agreement (without having to post any bond or prove actual damages), in addition to any other remedy to which they are entitled at law or in equity.

Page 53 of 59

10.3 Notices. Any notice, request, demand or other communication given by any Party under this Agreement will be in writing, may be given by a Party or its legal counsel, and will be deemed to be duly given upon actual receipt or refusal if (i) personally delivered, or (ii) delivered by an internationally recognized express courier service which provides evidence of delivery, or (iii) transmitted by registered or certified mail, postage prepaid, return receipt requested, addressed to the Party to whom directed at that Party's address as it appears below or another address of which that Party has given notice. Notices of address change will be effective only upon receipt notwithstanding the provisions of the foregoing sentence.

If to Seller, to:

Lantheus Medical Imaging, Inc. 331 Treble Cove Road North Billerica, MA 01862 Attn: Michael Duffy, General Counsel and Senior Vice President, Strategy and Business Development

If to Buyer, to:

Isologic Innovative Radiopharmaceuticals Ltd. c/o DW Healthcare Partners 66 Wellington St. W., Suite 4030 Toronto, Ontario, Canada M5K 1J5 Attn: Sameer Mathur

provided, however, that if any Party will have designated a different address by notice to the other Party, then to the last address so designated.

10.4 Successors and Assigns: Assignment. This Agreement will be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. This Agreement or any part thereof, may not be assigned, in whole or in part, without the prior written consent of the other Party, which consent may be withheld in the sole discretion of the other Party; provided, however, that either Party may assign this Agreement without the consent of the other Party (a) in whole or in part to any Affiliate of such Party; it being agreed that no such assignment to a Party's Affiliate will release the assigning Party from its obligations under this Agreement and provided that the assigning Party will give prior written notice of any such assignment to the other Party, (b) in connection with the transfer and sale of all or substantially all of the assets or business lines to which this Agreement, or (c) in the case of the Buyer, in whole or in part to it lenders (as collateral security) under the Buyer's current or proposed credit facilities, it being agreed that no such assignment under this Agreement, or (c) in the case of the Buyer, in whole or in part to it lenders (as collateral security) under the Buyer's current or proposed credit facilities, it being agreed that no such assignment will release the Buyer from its obligations under this Agreement. In the event of any assignment under this Agreement, the assignee will enter into a joinder agreement pursuant to which it will be subject to the terms, conditions and covenants of this Agreement, and will be deemed to be "Buyer" or "Seller," as applicable, in the same capacity as the transferring Party.

Page 54 of 59

10.5 Entire Agreement; Modification. The Transaction Documents supersede all prior Contracts between the Parties relating to the subject matter of the Transaction Documents, including any term sheets (in each case, other than any confidential disclosure agreements previously entered into among the Parties), and the Transaction Documents are the entire and complete statement of the terms of the agreement between the Parties with respect to such subject matter. This Agreement may be amended, modified or supplemented only in a writing signed by Seller and Buyer.

10.6 <u>Waivers</u>. The failure of a Party to this Agreement at any time or times to require performance of any provision of this Agreement will in no manner affect its right at a later time to enforce the same. No waiver by a Party of any condition or of any breach of any term, covenant, representation or warranty contained in this Agreement will be effective unless in writing, and no waiver in any one or more instances will be deemed to be a further or continuing waiver of any such condition or breach in other instances or a waiver of any other condition or breach of any other term, covenant, representation or warranty.

10.7 Section and Other Headings. The section and other headings contained in this Agreement are for reference purposes only and will not in any way affect the meaning or interpretation of this Agreement.

10.8 Currency. Except as expressly provided in this Agreement, all references to currency contained in this Agreement are to lawful money of United States of America.

10.9 <u>Governing Law</u>. Subject to Section 10.11, any controversy, dispute or claim arising under or out of, or in connection with, or otherwise related to this Agreement (including the existence, validity, interpretation or breach of this Agreement and any claim based on contract, tort or statute) (each, a "<u>Dispute</u>") will be exclusively interpreted in accordance with, and governed by, the Laws of the State of New York, without regard to the conflicts of law rules thereof.

10.10 Submission to Jurisdiction; Consent to Service of Process; Waiver of Jury Trial.

(a) Subject to Section 10.11, the Parties by this Agreement irrevocably and unconditionally submit to the exclusive jurisdiction of any federal or state court located within the Borough of Manhattan of the City, County and State of New York over any disputes, controversies or claims arising from, relating to, or in connection with, this Agreement or any of the Transactions, and each Party by this Agreement irrevocably agrees that all claims in respect of such dispute or any suit, action proceeding related thereto will be heard and determined in such courts. The Parties by this Agreement irrevocably and unconditionally waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the laying of venue of any such dispute brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the Parties agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law.

Page 55 of 59

(b) Each of the Parties by this Agreement consents to process being served by any party to this Agreement in any suit, action or proceeding by the delivery of a copy thereof in accordance with the provisions of Section 10.3.

(c) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT OR THE OTHER TRANSACTION DOCUMENTS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES AND, THEREFORE, EACH SUCH PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LEGAL ACTION ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE OTHER TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY TO THIS AGREEMENT CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT SEEK TO ENFORCE THE FOREGOING WAIVER IN THE EVENT OF A LEGAL ACTION, (B) SUCH PARTY HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.10(c).

10.11 Arbitration. Any Dispute will be exclusively resolved by binding arbitration, which arbitration will be commenced by sending a written notice to the other Party demanding arbitration of that Dispute (the "Demand"). In that event, the Dispute will be finally resolved by arbitration in accordance with the United States Arbitration Act and the Commercial Arbitration Rules of the American Arbitration Association. The place of the arbitration will be New York, New York. The arbitration will be conducted in the English language before a panel of three arbitrators. Each Party will name one arbitrator, and the two so named will name the third arbitrator, who will act as chairperson. If the two party arbitrators cannot agree on a third arbitrator within *** (***) days after the Demand, the third arbitrator will be selected by the American Arbitration Association. The arbitrators will promptly meet, fix the time, date and place of the hearing (as specified in Section 10.10) and notify the Parties. The arbitration will be conducted within ninety (90) days after receipt of any Demand. The panel of arbitrators will promptly transmit an executed copy of its decision to the Parties. The decision of the arbitrators will be final, binding and conclusive upon the Parties. Judgment on the award rendered by the arbitrators may be entered in any court specified in Section 10.10 having jurisdiction thereof. Each Party retains the right to seek from a court specified in Section 10.10 any interim or provisional relief that may be necessary to protect the rights or property of that Party pending the establishment of the arbitrators' determination of the merits of the controversy, and any such action will not be deemed incompatible with this Agreement to arbitrate or a waiver of the right to arbitration. The obligations of the Parties under this Section 10.11 are specifically enforceable and will survive any termination of this Agreement. All awards are subject to Sections 7.3, 7.6, 7.7, 7.8 and 7.12; provided that the arbitrators may award to the party prevailing in the arbitration its reasonable out-of-pocket costs, including the reasonable fees and expenses of the arbitrators and legal counsel incurred in the arbitration proceedings, or the arbitrators may award the costs on a distributive basis that apportions costs on an issue-by-issue basis and based on the inverse proportion that any amount actually contested but not awarded to Indemnifying Party or the Indemnified Person bears to the aggregate amount actually contested by Indemnifying Party and Indemnified Person, respectively.

Page 56 of 59

10.12 <u>Severability</u>. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction will, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Agreement, and any such prohibition and unenforceability in any jurisdiction will not invalidate or render unenforceable such provision in any other jurisdiction.

10.13 <u>No Third Party Beneficiaries</u>. Neither this Agreement nor any provision of this Agreement is intended to confer upon any Person (other than the Parties to this Agreement and the Indemnified Persons, which are third party beneficiaries entitled to enforce the provisions of Article 7 as if an original party to this Agreement) any rights or remedies under this Agreement.

10.14 No Recourse Against Nonparty Affiliates. All claims, obligations, liabilities, or causes of action (whether in contract or in tort, in law or in equity, or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to this Agreement, or the negotiation, execution, or performance of this Agreement (including any representation or warranty made in, in connection with, or as an inducement to, this Agreement), may be made only against (and are those solely of) the entities that are expressly identified as Parties ("Contracting Parties"). No Person who is not a Contracting Party, including without limitation any director, officer, employee, incorporator, member, partner, manager, stockholder, affiliate, agent, attorney, or representative of, and any financial advisor or lender to, any Contracting Party, or any director, officer, employee, incorporator, member, partner, manager, stockholder, affiliate, agent, attorney, or representative of, and any financial advisor or lender to, any of the foregoing ("Nonparty Affiliates"), will have any liability (whether in contract or in tort, in law or in equity, or granted by statute) for any claims, causes of action, obligations, or liabilities arising under, out of, in connection with, or related in any manner to this Agreement or based on, in respect of, or by reason of this Agreement or its negotiation, execution, performance, or breach; and, to the maximum extent permitted by law, each Contracting Party hereby waives and releases all such liabilities, claims, causes of action, and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Law, (a) each Contracting Party hereby waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at law or in equity, or granted by statute, to avoid or disregard the entity form of a Contracting Party or otherwise impose liability of a Contracting Party on any Nonparty Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in, in connection with, or as an inducement to this Agreement.

10.15 <u>Construction</u>. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.

Page 57 of 59

10.16 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which will be deemed to be an original, and such counterparts will together constitute one and the same instrument. A facsimile or other electronic transmission of an executed counterpart signature page will be deemed an original.

10.17 <u>Incorporation of Schedules and Exhibits</u>. The schedules and exhibits to this Agreement are incorporated into this Agreement and will be deemed a part of this Agreement as if set forth in this Agreement in full. References in this Agreement to "this Agreement" and the words "in this Agreement," "of this Agreement" and words of similar import refer to this Agreement (including its schedules and exhibits as an entirety). In the event of any conflict between the provisions of this Agreement and any such exhibit, the provisions of this Agreement will control.

[The remainder of this page is left blank intentionally.]

Page 58 of 59

IN WITNESS WHEREOF, the Parties to this Agreement have executed this Agreement on the day and year first above written.

SELLER:

LANTHEUS MI CANADA, INC.

By: /s/ Mary Anne Heino

Name: Mary Anne Heino Title: President and Chief Executive Officer

BUYER:

ISOLOGIC INNOVATIVE RADIOPHARMACEUTICALS LTD.

By: /s/ Sameer Mathur

Name: Sameer Mathur Title: Director

Page 59 of 59

<u>Exhibit A to</u> <u>Amended and Restated Purchase Agreement</u>

FORM OF BILL OF SALE (Canadian Radiopharmacies)

January 13, 2016 effective at 00:01 a.m.

This BILL OF SALE is made by and between (i) Lantheus MI Canada, Inc., a Canadian corporation ("<u>Seller</u>"), on the one hand, and (ii) Isologic Innovative Radiopharmaceuticals Ltd., a Canadian corporation ("<u>Buyer</u>"), on the other hand, as of the date first written above (this "<u>Bill of Sale</u>").

Buyer and Seller also may be referred to in this Bill of Sale each as a "Party" and collectively as the "Parties."

RECITALS

WHEREAS the Seller and Buyer entered into an asset purchase agreement as of the date hereof (the "Asset Purchase Agreement") pursuant to which, among other things, the Buyer agreed to purchase and the Seller agreed to sell certain assets used in the operation of the Radiopharmacy Business;

WHEREAS the Parties wish to confirm the closing of the transaction contemplating the purchase of the Purchased Assets, the whole as provided for in the Asset Purchase Agreement;

NOW, THEREFORE, in consideration of the premises and mutual agreements herein contained, and for other good and valuable consideration (the receipt and sufficiency of which are hereby acknowledged by each Party), les Parties agree as follows:

ARTICLE 1

INTERPRETATION

1.1 Definitions. Unless otherwise defined in this Bill of Sale, the capitalized words and expressions used herein shall have the meaning given thereto in the Asset Purchase Bill of Sale.

1.2 <u>Articles, Sections and Headings</u>. The division of this Bill of Sale into Articles, Sections and Schedules and the insertion of headings are for convenience of reference only and will not affect the construction or interpretation of this Bill of Sale. The terms "hereof", "hereinder", "herein" and similar expressions refer to this Bill of Sale and not to any particular Article, Section, Schedule or other portion hereof. References herein to Articles, Sections or Schedules are to Articles, Sections and Schedules of this Bill of Sale unless otherwise expressly stated herein.

Page 1 of 7

1.3 Extended Meaning. In this Bill of Sale, words importing the singular number also include the plural and vice versa and words importing any gender include all genders. The term "including" means "including, without limiting the generality of the foregoing".

1.4 Schedules. Schedule 2.1 attached hereto shall be deemed to form a part hereof.

ARTICLE 2

PURCHASE AND SALE

2.1 <u>Purchase and Sale of Purchased Assets</u>. In accordance with and subject to the terms and conditions set forth in the Asset Purchase Agreement, the Seller hereby sells, transfers and conveys to the Buyer, which accepts, effective on January 12, 2016 at 00:01 a.m. (the "<u>Closing Date</u>"), all of the rights and titles to and interests in the Purchased Assets held by the Seller.

As for the Accounts Receivable, the Seller agrees to immediately sign, at the Buyer's request, a notice to the debtors of the Accounts Receivable in a form substantially similar to Schedule 2.1.

ARTICLE 3

OBLIGATIONS

3.1 <u>Assumed Liabilities</u>. In accordance with the terms and conditions of the Asset Purchase Agreement, the Buyer hereby assumes the Assumed Liabilities on behalf and to the entire exoneration of the Seller, effective as of the Closing Date.

ARTICLE 4

GENERAL PROVISIONS

4.1 <u>Precedence</u>. This Bill of Sale is entered into in accordance with the Asset Purchase Agreement, and it shall not operate to amend the Asset Purchase Agreement or derogate therefrom. Should a provision of this Bill of Sale be incompatible with the provisions of the Asset Purchase Bill of Sale, the provisions of the Asset Purchase Agreement shall take precedence over this Bill of Sale.

4.2 <u>Further Assurances</u>. Each of the Parties hereto shall from time to time execute and deliver all such further documents and instruments and do all acts and things as another Party may, after the Closing Date, reasonably require to give effect to this Bill of Sale or better evidence or perfect the full intent and meaning of this Bill of Sale.

4.3 <u>Successors</u>, <u>Assigns and Assignments</u>. This Bill of Sale will be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. This Bill of Sale or any part thereof, may not be assigned, in whole or in part, without the prior written consent of the other Party, which consent may be withheld in the sole discretion of the other Party; provided, however, that either Party may assign this Bill of Sale without the consent of the other Party (a) in whole or in part to any Affiliate of such Party, it being agreed that no such

Page 2 of 7

assignment to a Party's Affiliate will release the assigning Party from its obligations under this Bill of Sale and provided that the assigning Party will give prior written notice of any such assignment to the other Party, or (b) in connection with the transfer and sale of all or substantially all of the assets or business lines to which this Bill of Sale relates of such Party or any of its Affiliates, it being agreed that no such assignment will release the assigning Party from its obligations under this Bill of Sale. In the event of any assignment under this Bill of Sale, the assignee will enter into a joinder agreement pursuant to which it will be subject to the terms, conditions and covenants of this Bill of Sale, and will be deemed to be "Buyer" or "Seller," as applicable, in the same capacity as the transferring Party.

4.4 <u>Entire Agreement; Modifications</u>. This Bill of Sale, the Asset Purchase Agreement and the Transaction Documents supersede all prior Contracts between the Parties relating to the subject matter of the Transaction Documents, including any term sheets (in each case, other than any confidential disclosure agreements previously entered into among the Parties), and the Transaction Documents are the entire and complete statement of the terms of the agreement between the Parties with respect to such subject matter. This Bill of Sale may be amended, modified or supplemented only in a writing signed by Seller and Buyer.

4.5 <u>Amendments and Waivers</u>. The failure of a Party to this Bill of Sale at any time or times to require performance of any provision of this Bill of Sale will in no manner affect its right at a later time to enforce the same. No waiver by a Party of any condition or of any breach of any term, covenant, representation or warranty contained in this Bill of Sale will be effective unless in writing, and no waiver in any one or more instances will be deemed to be a further or continuing waiver of any such condition or breach in other instances or a waiver of any other condition or breach of any other term, covenant, representation or warranty.

4.6 Section and Other Headings. The section and other headings contained in this Bill of Sale are for reference purposes only and will not in any way affect the meaning or interpretation of this Bill of Sale.

4.7 <u>Notices</u>. Any demand, notice or other communication to be given in connection with this Bill of Sale shall be sent in writing and will be given and formulated in keeping with Section 10.3 of the Asset Purchase Agreement, mutatis mutandis, that applies to this Bill of Sale.

4.8 <u>Governing Law</u>. Subject to Section 4.10, any controversy, dispute or claim arising under or out of, or in connection with, or otherwise related to this Bill of Sale (including the existence, validity, interpretation or breach of this Bill of Sale and any claim based on contract, tort or statute) (each, a "Dispute") will be exclusively interpreted in accordance with, and governed by, the Laws of the State of New York, without regard to the conflicts of law rules thereof.

4.9 Submission to Jurisdiction; Consent to Service of Process; Waiver of Jury Trial.

(a) Subject to Section 4.10, the Parties by this Bill of Sale irrevocably and unconditionally submit to the exclusive jurisdiction of any federal or state court located within the Borough of Manhattan of the City, County and State of New York over any disputes, controversies or claims arising from, relating to, or in connection with, this Bill of Sale, and each

Page 3 of 7

Party by this Bill of Sale irrevocably agrees that all claims in respect of such dispute or any suit, action proceeding related thereto will be heard and determined in such courts. The Parties by this Bill of Sale irrevocably and unconditionally waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the laying of venue of any such dispute brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the Parties agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law.

(b) Each of the Parties by this Bill of Sale consents to process being served by any party to this Bill of Sale in any suit, action or proceeding by the delivery of a copy thereof in accordance with the provisions of Section 4.10.

(c) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS BILL OF SALE IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES AND, THEREFORE, EACH SUCH PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LEGAL ACTION ARISING OUT OF OR RELATING TO THIS BILL OF SALE, THE OTHER TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY TO THIS BILL OF SALE CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT SEEK TO ENFORCE THE FOREGOING WAIVER IN THE EVENT OF A LEGAL ACTION, (B) SUCH PARTY HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS BILL OF SALE BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION (c).

4.10 Arbitration. Any Dispute will be exclusively resolved by binding arbitration, pursuant to Section 10.11 of the Asset Purchase Agreement.

4.11 <u>Severability</u>. Any provision of this Bill of Sale which is prohibited or unenforceable in any jurisdiction will, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Bill of Sale, and any such prohibition and unenforceability in any jurisdiction will not invalidate or render unenforceable such provision in any other jurisdiction.

4.12 <u>Construction</u>. The Parties have participated jointly in the negotiation and drafting of this Bill of Sale. In the event an ambiguity or question of intent or interpretation arises, this Bill of Sale will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Bill of Sale.

4.13 <u>Counterparts</u>. This Bill of Sale may be executed in two or more counterparts, each of which will be deemed to be an original, and such counterparts will together constitute one and the same instrument. A facsimile or other electronic transmission of an executed counterpart signature page will be deemed an original.

Page 4 of 7

4.14 <u>Incorporation of Schedules and Exhibits</u>. The schedules and exhibits to this Bill of Sale are incorporated into this Bill of Sale and will be deemed a part of this Bill of Sale as if set forth in this Bill of Sale in full. References in this Bill of Sale to "this Bill of Sale" and the words "in this Bill of Sale", "of this Bill of Sale" and words of similar import refer to this Bill of Sale (including its schedules and exhibits as an entirety). In the event of any conflict between the provisions of this Bill of Sale and any such exhibit, the provisions of this Bill of Sale will control.

[The remainder of this page is left blank intentionally.]

Page 5 of 7

IN WITNESS WHEREOF, the Parties to this Bill of Sale have executed this Bill of Sale on the day and year first above written.

SELLER:

LANTHEUS MI CANADA, INC.

By:

Name: Cyrille Villeneuve Title: Vice President, International

BUYER:

ISOLOGIC INNOVATIVE RADIOPHARMACEUTICALS LTD.

By: _____

Name: Sameer Mathur Title: Director

SCHEDULE 2.1

NOTICE OF TRANSFER TO CLIENT ACCOUNTS

<u>Exhibit B to</u> <u>Amended and Restated Purchase Agreement</u>

TRADE-MARK ASSIGNMENT

WHEREAS Lantheus MI Canada, Inc., the full post office address of whose principal office or place of business is 1111 Dr. Frederik-Philips Boulevard, suite 100, Montreal, Quebec, H4M 2X6 (the "Assignor"), is the owner of the entire right, title and interest in and to the trade-mark identified in Schedule A attached hereto (the "Mark"), as well as the goodwill associated with and symbolised by the Mark;

WHEREAS Isologic Innovative Radiopharmaceuticals Ltd., the full post office address of whose principal office or place of business is 1855, 32e Avenue, Lachine, Montreal, Quebec, H8T 3J1 (the "Assignee") is desirous of obtaining the entire right, title and interest in and to the Mark, as well as the goodwill associated with and symbolised by the Mark;

NOW, therefore, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Assignor hereby sells, assigns and transfers to said Assignee, its successors, assigns and legal representatives, the entire right, title and interest in and to the Mark, including the goodwill associated with and symbolised by the Mark, together with the right to registration, the applications for registration and registrations set out in Schedule A attached hereto and all rights of action accrued under and by virtue thereof, including the right to sue and recover for past infringement of the Mark and any rights of opposition accrued to the Assignor and any of its predecessors in title (the "Assignment");

AND the Assignor agrees to execute all such documents and do all such acts, at the costs of the Assignee, as may be necessary to give effect to this Assignment and the recordal thereof;

AND the Assignor declares that any prior use of the Mark by it or by any of its predecessors in title enures to the exclusive benefit of the Assignee.

[SIGNATURE PAGE FOLLOWS]

The Assignor: Lantheus MI Canada, Inc.

By:

Name: Cyrille Villeneuve Title: Vice President, International

The Assignee: Isologic Innovative Radiopharmaceuticals Ltd.

By:

Name: Sameer Mathur Title: Director

<u>Exhibit B to</u> <u>Amended and Restated Purchase Agreement</u>

<u>SCHEDULE A</u> <u>CANADA</u>

<u>Exhibit C to</u> <u>Amended and Restated Purchase Agreement</u>

FORM OF SUPPLY AGREEMENT

This **Supply Agreement** (this "<u>Agreement</u>") is made by and between **Lantheus MI Canada, Inc.**, a Canadian corporation ("<u>LMIC</u>"), and **Isologic Innovative Radiopharmaceuticals, Ltd.**, a Canadian corporation ("<u>Isologic</u>"), as of January 13, 2016 (the "<u>Effective Date</u>").

Each of LMIC and Isologic are referred to individually as a "Party" and collectively as the "Parties." Capitalized terms used in this Agreement are defined in Article 1.

In consideration of the mutual covenants and agreements contained in this Agreement, the Parties, intending to be legally bound, hereby agree as follows.

ARTICLE 1 DEFINITIONS

- 1.1. <u>Defined Terms</u>. As used in this Agreement, the following terms will have the following meanings:
 - (a) "<u>Affiliate</u>" as applied to any Person, means any other Person directly or indirectly controlling, controlled by, or under common control with, that Person. For the purposes of this definition, "<u>control</u>" (including, with correlative meanings, the terms "<u>controlling</u>," "<u>controlled by</u>" and "<u>under common control with</u>"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities or by contract or otherwise. Notwithstanding the foregoing, (i) none of Avista Capital Partners, its associated companies and entities, their respective successors and assigns or their respective direct and indirect investments (other than Lantheus Holdings, Inc. and its direct and indirect subsidiaries) will be deemed to be Affiliates of LMIC, and (ii) none of DW Healthcare Partners, its associated companies and entities, their respective successors and assigns or their respective direct and indirect investments (other than Isologic and its direct and indirect subsidiaries) will be deemed to be Affiliates of LMIC, and indirect investments (other than Isologic and its direct and indirect subsidiaries) will be deemed to be Affiliates of Isologic.
 - (b) "<u>Agreement</u>" has the meaning set forth in the preamble.
 - (c) "<u>Anti-Bribery Laws</u>" has the meaning set forth in Section 5.2(b).
 - (d) "<u>APA</u>" has the meaning set forth in the definition of the term Isologic Radiopharmacies.
 - (e) "CARDIOLITE®" means CARDIOLITE® (Kit for the Preparation of Technetium Tc99m Sestamibi for Injection).

Page 1 of 27

- (f) "<u>cGMP</u>" means all current good manufacturing practices, under Title 21 of the United States Code of Federal Regulations, as amended from time to time, and under the Canadian *Food and Drugs Regulations*, as amended from time to time.
- (g) "Cold Kits" means Sestamibi and NEUROLITE®.
- (h) "Confidential Information" means, with respect to a Disclosing Party, any confidential or proprietary information (including pricing), data, materials or documents of that Party and/or its Affiliates relating to or disclosed in connection with this Agreement or the transactions contemplated under this Agreement and all notes, analyses, compilations, data, translations, studies, memoranda, operating procedures or other documents prepared by the Receiving Party and/or its Representatives to the extent containing or otherwise reflecting that information, data, material or documents, in each case, irrespective of format; provided, however, that the term "Confidential Information" does not include, and there will be no obligation under this Agreement with respect to, information, data, material or documents that (a) are or become generally available to the public, other than as a result of a disclosure by the Receiving Party and/or its Representatives in breach of this Agreement or any other commercial or confidentiality agreement between the Parties and/or their respective Affiliates (in each case, other than (i) specific information that is merely embraced by more general information in the public domain or in the Receiving Party's possession, or (ii) if it constitutes a combination which can be reconstructed from multiple sources in the public domain or the Receiving Party's possession, none of which shows the whole combination of the Confidential Information), (b) the Receiving Party or any of its Representatives can demonstrate by its written records was or became available to the Receiving Party or that Representative from a source other than the Disclosing Party and/or its Representatives (in each case, provided that the source of that information, data, material or documents was not known by the Receiving Party or any of its Representatives to be bound by a confidentiality agreement with, or other contractual, legal or fiduciary obligation of confidentiality to, the Disclosing Party and/or its Representatives with respect to that information, data, material or documents), or (c) are developed independently by the Receiving Party and/or its Representatives without reference to or use of the Confidential Information of the Disclosing Party.
- (i) "<u>Conflicting Agreement</u>" and "<u>Conflicting Agreements</u>" have the meanings set forth in Section 5.9.
- (j) "<u>Demand</u>" has the meaning set forth in Section 8.6.
- (k) "<u>Disclosing Party</u>" has the meaning set forth in Section 5.7(a).
- (l) "<u>Dispute</u>" has the meaning set forth in Section 8.6.
- (m) "<u>Effective Date</u>" has the meaning set forth in the preamble.

Page 2 of 27

- (n) "Excluded Claims" means any and all claims, obligations, liabilities, controversies and causes of action that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to, (i) any breach of Section 5.7 (Confidential Information), (ii) the gross negligence, willful misconduct or fraud of any Party, its Affiliates or any of their respective employees, independent contractors or other agents or (iii) any third party damages that are indemnifiable under Article 7 (Indemnification).
- (o) "Expected Moly Costs" has the meaning set forth in Section 2.2(a).
- (p) "<u>FCPA</u>" has the meaning set forth in Section 5.2(b).
- (q) "<u>FCPA Parties</u>" has the meaning set forth in Section 5.2(b).
- (r) "Force Majeure Event" means, with respect to an affected Party, any circumstances or events that are beyond the reasonable control of any one or more of: that Party, its respective Affiliates, or any of their respective vendors, suppliers or shipping carriers (including, in each case to the extent such circumstances or events are beyond the reasonable control of the affected Party, any (a) act of God, (b) natural disaster or severe weather condition (e.g., lightning, earthquakes, hurricanes, floods, tornadoes, drought, blizzards, ice storms, volcanic eruption, epidemic, etc.), fire or explosion, (c) war, invasion, hostilities (whether war is declared or not), terrorist threat or act, riot, rebellion, mutiny, sabotage or other civil unrest, (d) act or decision of any Governmental Authorities or change in applicable Law, (e) sinking, crashing, embargo or blockade, (f) strikes, labor disturbances, stoppages or slowdowns or other industrial disturbances, (g) failure or delay of public utilities or common carriers, (h) batch failure, supply failure or outage, equipment failure or malfunction, shortages of fuel, power or raw materials or (i) any other circumstance or event that is not under the reasonable control of the affected Party).
- (s) "<u>Gallium</u>" means Gallium 67 (Gallium Citrate Ga67 Injection).
- (t) "<u>Generators</u>" means Technetium Tc99m (regardless of technology or delivery mechanism used), including TechneLite®, any other Technetium Tc99m generators and any generic version thereof.
- (u) "<u>Governmental Authority</u>" means any government or governmental or regulatory body thereof, or political subdivision thereof, whether domestic or foreign, federal, provincial or local, or any department (or subdivision thereof), commission, bureau, tribunal, agency, board, instrumentality or authority thereof, any court or arbitrator (public or private), or any applicable stock exchange.
- (v) "GST/HST" means the goods and services tax and/or harmonized sales tax levied under Part IX of the Excise Tax Act (Canada).
- (w) "Hot Products" means Gallium, TechneLite® and Thallium.

Page 3 of 27

- (x) "Indemnified Party" has the meaning set forth in Section 7.3(a).
- (y) "<u>Indemnifying Party</u>" has the meaning set forth in Section 7.3(a).
- (z) "Initial Term" has the meaning set forth in Section 6.1.
- (aa) "Isologic" has the meaning set forth in the preamble.
- (bb) "Isologic Radiopharmacies" means all current and future radiopharmacies (and other bulk or unit dose radiopharmaceutical distribution businesses) owned and/or operated by Isologic and/or any of its Affiliates and/or any of their respective successors or assigns, including the radiopharmacies acquired by Isologic pursuant to the Asset Purchase Agreement by and between LMIC and Isologic, dated as of the date of this Agreement (as amended, modified and/or supplemented from time to time, the "<u>APA</u>").
- (cc) "Isologic Related Persons" has the meaning set forth in Section 7.1.
- (dd) "Latent Defect" means a defect in a Product not conforming to Specifications that existed at the time the Product is received by Isologic which could not have been detected by Isologic utilizing reasonable and customary inspection procedures for incoming products.
- (ee) "Law" means any law, statute, regulation, ordinance, rule, Order or requirement enacted, promulgated, entered into, or imposed by, any Governmental Authority (including, for the sake of clarity, common law).
- (ff) "Liabilities" has the meaning set forth in Section 7.1.
- (gg) "<u>LMIC</u>" has the meaning set forth in the preamble.
- (hh) "LMIC Related Persons" has the meaning set forth in Section 7.1.
- (ii) "Measurement Period" has the meaning set forth in Section 3.3(a).
- (jj) "Minimum Purchase Requirement" and "Minimum Purchase Requirements" have the meanings set forth in Section 3.1.
- (kk) "<u>MPI Products</u>" means any myocardial perfusion imaging kits or products, including Sestamibi, Myoview® and all generic versions of any of the foregoing.
- (ll) "<u>NBI Products</u>" means any nuclear brain imaging kits or products, including NEUROLITE®, CERETEC® and all generic versions of any of the foregoing.
- (mm) "<u>NEUROLITE®</u>" means NEUROLITE® (Kit of the Preparation of Technetium Tc99m Bicisate for Injection) and all generic versions of any of the foregoing.
- (nn) "Party" or "Parties" has the meaning set forth in the preamble.

Page 4 of 27

- (00) "<u>Person</u>" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization, any other business entity or a Governmental Authority.
- (pp) "Product" means any product set forth on the Product Schedule, including any generic version thereof.
- (qq) "Product Schedule" means Exhibit A attached to this Agreement (as may be amended by LMIC in accordance with Section 2.3 from time to time).
- (rr) "<u>Receiving Party</u>" has the meaning set forth in Section 5.7(a).
- (ss) "<u>Remediation Period</u>" has the meaning set forth in Section 3.3(b)(ii)(1).
- (tt) "<u>Representatives</u>" means, with respect to any Person, the Affiliates of that Person and the directors, employees, independent contractors, subcontractors, agents, lenders and consultants of that Person or of any of those Affiliates.
- (uu) "<u>Sales Taxes</u>" means the GST/HST and any similar sales taxes imposed by any Canadian Governmental Authority in connection with the purchase and supply of the Products.
- (vv) "Sestamibi" means CARDIOLITE® and any generic version thereof.
- (ww) "Shortfall Payment" has the meaning set forth in Section 3.3(b).
- (xx) "SIEA" has the meaning set forth in Section 5.6.
- (yy) "Specifications" has the meaning set forth in Section 4.6.
- (zz) "<u>Term</u>" has the meaning set forth in Section 6.1.
- (aaa) "TechneLite®" means TechneLite® (Technetium Tc99m Generator).
- (bbb) "Thallium" means Thallium 201 (Thallous Chloride Tl201 Injection).
- (ccc) "Third Party Products" means those products set forth under the heading "Third Party Products" on the Product Schedule.
- 1.2. <u>Interpretational Matters</u>. References in this Agreement to any gender include references to all genders, and references to the singular include references to the plural and vice versa. When used in this Agreement, the words "include," "includes" and "including" will be deemed to be followed by the phrase "without limitation." When used in this Agreement, the word "or" is not exclusive. Unless the context otherwise requires, references in this Agreement to Articles, Sections and Exhibits will be deemed to be references to the Articles and Sections of, and the Exhibits to, this Agreement; and the Exhibits referred to in this Agreement will be construed with, and as an integral part of,

Page 5 of 27

this Agreement to the same extent as if they were set forth verbatim in this Agreement. Unless the context otherwise requires, references in this Agreement to an agreement, instrument or other document means that agreement, instrument or other document as amended, supplemented and modified from time to time to the extent permitted by the provisions thereof. Unless the context otherwise requires, when used in this Agreement, the word "hereby" and the phrases "of this Agreement," "by this Agreement" and "in this Agreement" and words and phrases of similar meaning will be deemed to refer to this Agreement in its entirety and not to any particular Article, Section or provision of this Agreement. The section and other headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event that an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties, and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.

ARTICLE 2

GENERAL TERMS OF PURCHASES AND SALES

2.1. <u>Purchase and Sale of Products</u>. All purchases of Products by Isologic from LMIC during the Term will be governed by, and will be subject to the terms and conditions of, this Agreement. LMIC will use commercially reasonable efforts to produce and sell to Isologic the sizes and quantities of Products as requested by Isologic, taking into account LMIC's manufacturing capacity and availability of supply and subject to Section 8.5; provided that, if LMIC is unable, for any reason, to fill any size of a Product specified in a purchase order on any given day, then LMIC will be entitled to make reasonable size substitutions (e.g., by providing two (2) 2.500 curie TechneLite® generators or one (1) 6.000 curie TechneLite® generator in fulfillment of a purchase order for one (1) 5.000 curie TechneLite® generator). In no event will Isologic re-sell, transfer, give or otherwise supply any Products supplied by LMIC, or any other bulk radiopharmaceuticals (not including bulk unit dose preparations) that compete with Products supplied by LMIC as of the date of this Agreement, to any third party without LMIC's prior written approval, other than as contemplated by this Agreement for unit dose preparations) (including bulk unit dose preparations) and other than certain third party cold kits to certain unit dose customers (as contemplated by the definition of the term "Radiopharmacy Business" in the APA), in each case by Isologic Radiopharmacies.

2.2. Purchase Price and Late Delivery Discounts.

(a) The purchase price to be invoiced to, and paid by, Isologic for any Products will be the purchase price set forth on the Product Schedule corresponding to that Product; provided, however, that each calendar year (commencing in calendar year 2017) LMIC will be entitled to increase the purchase price of any or all Products by up to *** percent (***%) of the previous year's purchase price, effective upon *** (***) days' prior written notice. In addition, if at any time during the Term LMIC's and its Affiliates' molybdenum-99 costs exceed Expected Moly Costs (as defined below) by ** percent (***%) or more, in the

Page 6 of 27

aggregate, and such increase is sustained for at least a *** (***) day period, then ***, provided that any increase in pricing pursuant to the foregoing shall only be effective upon *** (***) days' prior written notice (and provided further that Isologic may request that an independent auditor (which would enter into a confidentiality agreement in the form reasonably required by, and in favor of, LMIC) confirm the calculations thereof, without divulging any specific figures to Isologic). The term "Expected Moly Costs" means ***.

For purposes of illustration, if:

- (i) If the date of this Agreement was January 1, 2016 and Expected Moly Costs on that date are \$***/curie,
- (ii) Expected Moly Costs on January 1, 2017 are \$***/curie (as a result of ***, as described above); and
- (iii) LMIC and its Affiliates incur a cost increase such that molybdenum-99 costs on May 1, 2017 are \$***/per curie,

then *** of that \$***/curie molybdenum-99 cost increase from January 1, 2017 to May 1, 2017 would be passed through under this Section 2.2(a) (because the \$*** cost increase is less than **% of then-applicable Expected Moly Costs of \$***/curie).

If, however, molybdenum-99 costs are further increased to \$***/curie on June 1, 2017 and that increase is sustained for at least sixty (60) days, then \$***/curie of the aggregate molybdenum-99 cost increase (*calculated as follows:* ***) would be passed through under this Agreement, starting on July 31, 2017 (i.e., the sixtieth day of the molybdenum-99 cost increase period, assuming notice thereof was delivered to Isologic on June 1, 2017).

- (b) In the event that delivery of any Hot Product is delayed past the agreed upon local delivery time (as specified by the Isologic Radiopharmacy and accepted by LMIC) and that delay results in an actual, negative impact on the ordering Isologic Radiopharmacy's first production run for which that Product was intended to be used, then, (i) in the case of Generators, LMIC will reduce the price Isologic pays for that Product by *** and, (ii) in the case of other Hot Products, LMIC will reduce the price Isologic pays for that Product by ***; provided, in each case, that the foregoing will not apply to delays caused by any Force Majeure Event, for which there will be no price reduction.
- 2.3. <u>Cessation of Sale of any Product or Size: New Product Sizes</u>. LMIC reserves the right to modify the Product Schedule at any time during the Term, upon at least *** (***) days' prior written notice to Isologic, to reflect that it has (a) ceased to manufacture or sell any Product or (b) introduced any new size of Product (and to establish the initial purchase price for that new size, provided that the introduction of new sizes of Product and the pricing thereof shall not be implemented in a manner so as to circumvent the pricing for Products set forth in this Agreement) or it has discontinued any size of Product.

Page 7 of 27

ARTICLE 3 MINIMUM PURCHASE OBLIGATIONS

- 3.1. <u>Minimum Purchase Obligation</u>. Throughout the Term, Isologic guarantees to purchase from LMIC at least the following quantities of Products (each of clauses (a) through (e) individually, a Product's applicable "<u>Minimum Purchase Requirement</u>" and, collectively, the "<u>Minimum Purchase Requirements</u>"):
 - (a) *** percent (***%) of the Isologic Radiopharmacies' total requirements of Generators (measured in curies purchased, regardless of size);
 - (b) *** percent (***%) of the Isologic Radiopharmacies' total requirements of Thallium (measured in millicuries purchased, regardless of size);
 - (c) *** percent (***%) of the Isologic Radiopharmacies' total requirements of Gallium (measured in millicuries purchased, regardless of size);
 - (d) All of the Third Party Products in LMIC's inventory as of the Effective Date in the quantities set forth in the Product Schedule.

Throughout the Term, Isologic agrees that ***.

Notwithstanding the foregoing, Isologic will not be deemed to be in breach of Section 3.1(a) to the extent that the specific quantities of product required to be purchased by Isologic under its Conflicting Agreement causes it to fail to meet its Minimum Purchase Requirements for the analogous Product in any Measurement Period through the ***. For the avoidance of doubt, (1) the full amount of the Minimum Purchase Commitments for all other Products for all Measurement Periods will apply at all times throughout the Term and (2) the full amount of the Minimum Purchase Commitments for all Measurement Periods product for all Measurement Periods commencing in the *** will apply at all times throughout the Term.

3.2. <u>Treatment of Undelivered Products</u>.

(a) In the event that Isologic places a purchase order for any Product that LMIC is unable, for any reason (including a Force Majeure Event), to deliver to Isologic, then that undelivered Product (to the extent representing normal quantities ordered in the ordinary course of Isologic's business) will be treated as having been purchased by Isologic for purposes of determining whether Isologic has satisfied the relevant Minimum Purchase Requirement (and, for the avoidance of doubt, Isologic shall not be required to pay any amount in respect of such Product), and Isologic shall also receive an invoice credit for the original purchase price of such undelivered Product (to the extent it was actually paid) against future purchases under this Agreement.

Page 8 of 27

- (b) In addition, to the extent that LMIC is unable to supply the quantities of Product requested by Isologic in a purchase order under this Agreement for any reason (including a Force Majeure Event), Isologic will have the right to purchase that Product from an alternate supplier for the period of that unavailability (and, for the duration of that period of unavailability, Isologic will not be in violation of its purchase obligations relating to the Minimum Purchase Requirements).
- (c) To the extent that LMIC is unable, for any reason (including a Force Majeure Event), to supply on any given day all quantities of a Product ordered by Isologic pursuant to a firm purchase order under this Agreement and by all of LMIC's other customers in the Territory pursuant to firm purchase orders, LMIC shall ***.

3.3. Compliance; Reporting; Shortfall Payments.

- (a) Compliance with the Minimum Purchase Requirement for each Product will be determined for each calendar quarter during the Term (each, a "<u>Measurement Period</u>"). Not later than *** (***) days after the end of each Measurement Period, Isologic will provide to LMIC (A) a timely and accurate written report in substantially the form set forth in <u>Exhibit B</u> that sets forth the Isologic Radiopharmacies' total requirements for curies, millicuries, doses or vials, as applicable, of each Product for that Measurement Period, (B) reasonable, true and accurate documentation supporting that report and (C) a written certification by an executive officer that Isologic has complied (or failed to comply) with all of its Minimum Purchase Requirements for that Measurement Period. Any disputes over the figures reflected in that report will be resolved by the Parties in good faith.
- (b) In the event that Isologic fails to satisfy the Minimum Purchase Requirement for any Product during any Measurement Period, Isologic will:
 - (i) with respect to any Hot Products, make the required Shortfall Payment for that Hot Product to LMIC on or before the last business day of the calendar month immediately following the end of that Measurement Period; and
 - (ii) with respect to any Cold Kit Products,
 - (1) order the shortfall quantity of that Product (i.e., reflecting the portion of the Minimum Purchase Requirement for that Product for which Isologic failed to place firm purchase orders for delivery within that Measurement Period) in the immediately following Measurement Period (the "<u>Remediation Period</u>"), in addition to complying with its Minimum Purchase Requirement for that Product for that Remediation Period (for the sake of clarity, purchases during a Remediation Period will be applied first towards the Minimum Purchase Requirement for that Remediation Period and then second towards the shortfall from the immediately preceding Measurement Period); and
 - (2) if Isologic fails to purchase any portion of that shortfall quantity during that Remediation Period, then make the required Shortfall Payment for that Product to LMIC on or before the last business day of the calendar month immediately following the end of that Remediation Period.

Page 9 of 27

With respect to any Product during a Measurement Period, "<u>Shortfall Payment</u>" means a cash payment required to be made by Isologic to LMIC for (x) the quantity of that Product (quantified in curies in the case of Generators) reflecting the portion of the Minimum Purchase Requirement for that Product for which Isologic failed to place firm purchase orders for delivery within that Measurement Period (or, for Cold Kits, that Remediation Period), (y) priced at, (1) in the case of Generators, *** or, (2) in the case of other Products, ***. Upon receipt of any Shortfall Payment for a Product with respect to a Measurement Period, Isologic will be deemed to have met the Minimum Purchase Requirement for that Product during that Measurement Period.

ARTICLE 4 PURCHASE AND SALE OF PRODUCTS: ACCEPTANCE AND REJECTION OF PRODUCTS

4.1. Purchase Orders.

- (a) For LMIC's smooth inventory and order management and so as to minimize the size and frequency of any shortfalls in meeting its Minimum Purchase Requirements, Isologic will (i) establish <u>***</u> recurring standing orders for Hot Products in the volumes necessary to satisfy its Minimum Purchase Requirement with respect to Hot Products for each calendar quarter and (ii) place <u>***</u> orders for Cold Kits in the volumes necessary to satisfy its Minimum Purchase Requirement with respect to Cold Kits for each calendar quarter. Isologic will also place a purchase order for all of the Third Party Products in LMIC's inventory as of the Effective Date in the quantities set forth in the Product Schedule in ***.
- (b) Isologic will place orders for Products under this Agreement in written, electronic or verbal form which will specify: (i) the quantity of each Product being ordered, (ii) the requested shipping date and (iii) the shipping destination (which must be a licensed Isologic Radiopharmacy). All orders are subject to (1) LMIC's customary ordering requirements and lead times as in effect from time to time, (2) LMIC's reasonable discretion to determine the method of shipment and (3) acceptance by LMIC, which will not be unreasonably withheld. The terms of this Agreement will prevail over any inconsistent terms in any purchase order, acknowledgment or invoice, and no additional terms other than those set forth in this Agreement or allowed pursuant to the terms of Section 2.1 (size substitutions) and this Section 4.1(b) in a purchase order, acknowledgment or invoice will be binding on either Party.

Page 10 of 27

- 4.2. <u>Shipments</u>. The risk of loss, delay (as and to the extent contemplated by Section 2.2) or damage in transit will remain with LMIC until delivery to the applicable Isologic Radiopharmacy. LMIC will prepay all shipping and insurance costs and will invoice Isologic for the actual costs and expenses incurred, plus *** percent (***%) of such shipping and insurance costs. LMIC agrees to use its reasonable best efforts to utilize the most cost efficient forms of shipping available that meet the requirements of the Parties and applicable Laws.
- 4.3. <u>Invoicing and Payment Terms</u>. LMIC will provide an invoice to Isologic with each shipment for the Products then delivered. All payments will be due and payable on a net *** (***) day basis. All payments will be made in *** by wire transfer as designated by LMIC, or by such other method as LMIC will notify Isologic from time to time. Interest will be payable on all amounts not paid as of *** days after the due date at a yearly rate of *** percent (***%) (or, if lower, the maximum interest rate permitted by Law) and will accrue from *** days after the due date until that sum is paid. LMIC shall ensure that each invoice submitted to Isologic under this Agreement shall separately set out all Sales Taxes imposed on Isologic, including any GST/HST payable by Isologic under this Agreement, and all prescribed information under the *Excise Tax Act (Canada)* (or any other applicable provincial tax legislation) required to claim all available refunds or credits of any Sales Taxes.
- 4.4. <u>Taxes</u>. Isologic will be responsible for and shall pay any applicable Sales Taxes charged on the sale of each Product sold pursuant to this Agreement or on any amounts payable to LMIC under this Agreement. LMIC shall reasonably cooperate with the reasonable requests of Isologic in any efforts by Isologic to obtain exemption from, or to minimize, any Sales Taxes for which Isologic is liable under this Agreement.
- 4.5. <u>Set Off.</u> Isologic will perform its obligations under this Agreement without setoff, deduction, recoupment or withholding of any kind for amounts owed or payable by LMIC, whether under this Agreement, the APA (or any transaction agreement referenced in the APA), applicable Law or otherwise and whether relating to LMIC's breach, insolvency or otherwise, in each case, unless such amounts owed or payable are definitively resolved by mutual written agreement or by a final and binding court or arbitration decision. Without limiting the foregoing, Isologic shall be entitled to deduct and set off from any amounts due and owing to LMIC under this Agreement any amounts owed by LMIC to Isologic pursuant to Article 7 of the APA that have been definitively resolved by mutual written agreement or by a final and binding court or by a final and binding court or arbitration.
- 4.6. <u>Product Warranties</u>. Each Product supplied to Isologic pursuant to this Agreement will, at the time of delivery, (a) be free from defects in material and workmanship; (b) conform to LMIC's specifications for that Product (as set forth in the applicable regulatory approval) as in effect from time to time; (c) comply with all applicable Laws relating to the production, storage, packaging, labeling, shipping and delivery of that Product; and (d) be produced in accordance with applicable cGMPs (clauses (a) through (d), collectively, the applicable "Specifications").

Page 11 of 27

- 4.7. <u>Non-Conforming Product</u>. Isologic may reject a shipment of any Product only if that Product fails to conform to (a) the type and quantity of Products ordered by Isologic in its purchase order (other than because of a size substitution permitted under Section 2.1) or (b) the applicable Specifications; provided that, in each case, Isologic notifies LMIC by telephone (or any other method agreed to by the Parties from time to time) of any such rejection within *** (***) days (for Hot Products) or *** (***) days (for Cold Kits) after receipt by Isologic of that shipment of Products; provided that, in the case of Latent Defects, such *** (***) or *** (***) day period will commence only upon discovery of such Latent Defects by Isologic or Isologic's customers. Isologic's sole and exclusive remedy with respect to any non-conforming Products will be (in each case, only to the extent such non-conforming Products were actually paid for) to receive, at Isologic's option, either (i) replacement quantities for those non-conforming Products, which replacement quantities shall be for no additional consideration, or (ii) an invoice credit for the original purchase price of those non-conforming Products from being taken into account in determining whether a material breach of this Agreement has occurred for the purposes of Section 6.2(a) of this Agreement.
- 4.8. <u>Product Recalls</u>. In the event that LMIC determines that a recall or withdrawal of the Products from the market is necessary, Isologic will take all actions appropriate in order to reasonably assist LMIC with that recall or withdrawal. The costs of the recall (including all costs of collecting, shipping and disposing of the recalled Product) will be borne by LMIC, unless the circumstances leading to the recall result from the negligence or fault of any of the Isologic Parties.

ARTICLE 5 REPRESENTATIONS, WARRANTIES AND COVENANTS

- 5.1. <u>Mutual Representations, Warranties and Covenants</u>. Each Party represents, warrants and covenants to the other Party that, as of the Effective Date and throughout the Term:
 - (a) (i) it is and will be duly organized, validly existing and in good standing under the laws of the jurisdiction of its formation; (ii) it has and will have the requisite power and authority to carry on its business as it is now being conducted and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties owned or leased by it makes that licensing or qualification necessary; (iii) this Agreement has been duly executed and delivered by it and constitutes and will continue to constitute its legal, valid and binding obligation, enforceable against it in accordance with its terms, except as the same may be limited by bankruptcy, insolvency, moratorium, reorganization or other Laws of general applicability relating to or affecting the enforcement of creditor's rights and general principles of equity; and (iv) except as disclosed on Schedule 5.1, the execution, delivery and performance by it of this Agreement does not and will not (A) violate or

Page 12 of 27

conflict with, result in a breach of, constitute a material default (or an event which, with or without notice or lapse of time or both, would constitute a default) under, any material contract to which it or any of its Affiliates is a party or by which it or any of its Affiliates or its or their respective assets is bound; (B) violate in any material respect any Law applicable to it or any of its Affiliates; or (C) violate or conflict with any provision of the organizational documents of it or any of its Affiliates; and

- (b) (i) neither it, nor any of its employees or agents, is debarred, excluded, suspended, proposed for debarment or otherwise ineligible for participation in any federal or provincial health care program; (ii) neither it, nor any of its employees or agents, has been convicted of or had a civil judgment rendered against it for commission of fraud or a criminal offense; and (iii) neither it nor any of its employees or agents is presently indicted for or otherwise criminally or civilly charged by a governmental entity or agency with commission of any of the offenses enumerated in this Section 5.1(b) (in the event of any material breach of this clause (b), the non-breaching Party may terminate this Agreement immediately upon written notice to the breaching Party).
- (c) it holds and will continue to hold all federal, provincial, municipal and other local licenses and permits necessary for the lawful conduct of its business and its activities under this Agreement, including (i) in the case of LMIC, its business of distributing Products or, (ii) in the case of Isologic, its radiopharmacy business for the preparation of the Products and for the dispensing of unit doses containing the Products; and that in each case none of those licenses or permits has or will have expired or has or will have been suspended, terminated, cancelled, not renewed or otherwise lost (in the event that any of the Isologic Radiopharmacies fails to maintain any such licenses or permits, Isologic will notify LMIC promptly of (but in no event later than *** (***) days following) that failure, and LMIC will not be required to deliver any Products to that Isologic promptly of (but in no event later than *** (***) days following) that failure, and LMIC shall not deliver any Products to Isologic where the production, storage, packaging, labelling or distribution of such Products would require such licenses or permits to be maintained (and for the avoidance of doubt, Sections 3.2(a) and (b) shall apply to any failure of LMIC to deliver Products for such reason).
- (d) it has ascertained and complies with, and will ascertain and comply with, all applicable Laws and internal policies, including in the case of Isologic those covering (i) disposal of radioactive materials, pollution, hazardous substances or the protection of human health, the environment or natural resources and (ii) handling, sales, marketing, preparation, use and distribution of the Products; and
- (e) it will not, on behalf of itself and its Affiliates, take any action that disparages the other Party or their respective products (including the Products and, in the case of Isologic, in any manner that may reduce or dilute the reputation or distinctiveness of any of the Product trademarks).

Page 13 of 27

5.2. FCPA and Anti-Bribery Law Compliance.

- (a) General. Each Party acknowledges that it intends to comply with all Laws governing its business and to conduct its activities with integrity. Each Party acknowledges and agrees that it is subject to LMIC's Foreign Corrupt Practices Act and Anti-Bribery Compliance Policy attached as <u>Exhibit C</u>; provided that Isologic shall not be bound by any amendments, modifications or supplements to such policy unless reasonable prior written notice of such amendment, modification or supplement is given to Isologic.
- (b) Anti-Bribery Laws. Each of LMIC and Isologic and their respective Affiliates and all of their respective officers, directors, employees, representatives, shareholders and agents (the foregoing, with respect to each Party, collectively, its "FCPA Parties") have complied, and will comply, with the terms of all applicable Laws, including the U.S. Foreign Corrupt Practices Act (the "FCPA") and laws of other nations that generally prohibit the payment of bribes, kickbacks and other improper payments (collectively, "Anti-Bribery Laws"). Isologic and LMIC each acknowledge that the FCPA, in particular, makes it unlawful to offer, pay, promise, or authorize to pay any money, gift or anything of value (including bribes, entertainment, kickbacks or any benefit), directly or indirectly, (i) to any foreign official or any foreign political party or (ii) to any person while knowing or suspecting that the payment or gift will be passed on to a foreign official, in connection with any business activity of the company. Isologic and LMIC each acknowledge that a "foreign official" generally means any employee of face of a government of a foreign country (*i.e.*, a country other than the United States of America), including any federal, regional or local department, agency or enterprise owned or controlled by the foreign government, any official of a foreign political party, any official or employee of a public international organization, any person acting in an official capacity for, or on behalf of, such entities, and any candidate for foreign political office.
- (c) *Isologic Compliance-Related Representations and Warranties.* Each of Isologic and LMIC (on behalf of itself and its respective FCPA Parties) makes the following representations and warranties to the other Party, and covenants and agrees as follows:
 - (i) It and its FCPA Parties have not, and will not, in connection with this Agreement or in connection with any other business transactions involving the other Party, make, promise, or offer to make any payment or transfer of anything of value, directly or indirectly, that has the purpose or effect of public or commercial bribery, or acceptance of or acquiescence in extortion, kickbacks, or other unlawful or improper means of obtaining business. This does not, however, prohibit normal and customary business entertainment or the giving of business mementos of normal value in

Page 14 of 27

connection with its or its FCPA Parties' performance under this Agreement, provided the entertainment or giving of the memento is otherwise legal under local law and does not violate or exceed the rules, code of conduct, or other ethical standards set by LMIC or Isologic or by the recipient's own agency, company or organization.

- (ii) At any time reasonably requested by a Party and upon reasonable prior notice, but no more than once per year, the other Party shall provide a compliance certification as to its compliance with this Section 5.2 (which shall be in such reasonable form as the requesting party and/or its Affiliates use for their analogous contractual counterparties at that time).
- (iii) If it or any of its respective FCPA Parties learns or has reason to know of: (i) any payment, offer or agreement to make a payment to a foreign official or political party for the purpose of obtaining or retaining business or securing any improper advantage for it or the other Party under this Agreement or otherwise, or (ii) any other development during the Term that in any way makes inaccurate or incomplete its and its respective FCPA Parties' anti-bribery-related representations, warranties or certifications at any time during the Term, then it or its respective FCPA Parties, as applicable, will promptly advise the other Party in writing of the information giving rise to its or its respective FCPA Parties' knowledge or suspicion that a violation may have occurred.
- (d) *Termination Rights.* If a Party believes, in good faith, that the other Party or any of its FCPA Parties has acted in any way that may subject the first-mentioned Party or any of its directors, officers or Affiliates to liability under the FCPA or any other Anti-Bribery Laws, or has otherwise failed to comply with the terms of Section 5.2, then the first-mentioned Party may terminate this Agreement immediately upon written notice to the other Party.
- (e) *Disclosure*. Isologic and LMIC each agrees (on behalf of itself and its respective FCPA Parties) that the other Party may make full disclosure of information relating to any possible violation of the FCPA, any Anti-Bribery Laws, or related offenses to the U.S. government and its agencies, the appropriate foreign government and its agencies, and to any other person or entity that such other Party determines has a legitimate need to know.
- 5.3. <u>Isologic's Representations, Warranties and Covenants</u>. Isologic represents, warrants and covenants to LMIC that, as of the Effective Date and throughout the Term:
 - (a) Isologic will not give or make any guarantees, warranties or representations as to the condition, quality, durability, performance, merchantability or fitness for a particular use or purpose or any other feature of any Product or other than or different from those provided by LMIC under this Agreement (any such other guarantee, warranty or condition, whether express or implied, made by Isologic to its customers will be and remain the sole responsibility of Isologic and will not impose any obligation on LMIC).
 - (b) It will not sell or distribute any expired Product.

Page 15 of 27

- 5.4. <u>Disclaimer of All Other Representations and Warranties</u>. THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS AGREEMENT ARE THE ONLY REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE SUBJECT MATTER OF THIS AGREEMENT AND ARE MADE EXPRESSLY IN LIEU OF ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, ALL OF WHICH ARE HEREBY DISCLAIMED, INCLUDING ANY IMPLIED WARRANTIES OF FITNESS FOR A PARTICULAR PURPOSE, MERCHANTABILITY, NON-INFRINGEMENT OR OTHERWISE.
- 5.5. LMIC Audit Rights. During the Term of this Agreement, upon reasonable prior written notice, LMIC may request that it or its Representatives (in the case of clause (i) below) or an independent auditor (having a nationally-recognized reputation and level of expertise, and which would enter into a confidentiality agreement in the form reasonably required by, and in favor of, Isologic) (in the case of clause (ii) below), to inspect Isologic's facilities and records and to confer with Isologic's employees and independent contractors, in each case, in a commercially reasonable manner, during normal business hours, in compliance with Section 5.7, and only to the extent reasonably necessary for purposes of conducting compliance inspections regarding (i) regulatory and quality matters in respect of the Products and (ii) confirming and ensuring the accuracy and completeness of the reports delivered by Isologic under Section 3.3(a) (and the information supporting such reports), in the case of this clause (ii), without any specific figures being divulged to LMIC. Inspections in respect of regulatory and quality matters and internal policies or under applicable Law, or in the event that LMIC has a reasonable basis for suspecting that Isologic's business is not being conducted in compliance with applicable Law or regulatory or quality requirements in respect of the Products. All inspections shall be conducted at LMIC's cost and in a manner that does not unreasonably interfere with the conduct of Isologic's normal business operations and may be conducted no more than *** per year (or more frequently, to the extent required for purposes of confirming the completion and effectiveness of any material remediation measures).
- 5.6. <u>Adverse Event Reporting</u>. Isologic and LMIC agree to comply with all of the terms and conditions of the Safety Information Exchange Agreement that is being executed and delivered simultaneously with this Agreement and attached as <u>Exhibit D</u> (the "<u>SIEA</u>"). The SIEA is incorporated by reference into the terms of this Agreement.
- 5.7. <u>Confidentiality</u>.
 - (a) All Confidential Information disclosed by or on behalf of a Party (the "<u>Disclosing Party</u>") and received by the other Party (the "<u>Receiving Party</u>") will be held in strict confidence by the Receiving Party and its relevant Representatives. From

Page 16 of 27

and after the Effective Date of this Agreement, except as otherwise contemplated by this Agreement, the Receiving Party will not, and will cause its Representatives not to, directly or indirectly, disclose, reveal, divulge or communicate the Confidential Information of the Disclosing Party to any third party other than Representatives of the Receiving Party or of its Affiliates who reasonably need to know that Confidential Information in the performance of Isologic's responsibilities under this Agreement and who are obligated or directed to maintain the confidential Information. The Receiving Party will not use the Confidential Information for any purpose other than in connection with exercising its rights and fulfilling its obligations under this Agreement or to the extent required for financial reporting, legal or regulatory compliance purposes. The Receiving Party and its Representatives will use the same degree of care to prevent and restrain the unauthorized use or disclosure of the Confidential Information of the Disclosing Party as they currently use for their own confidential information of a like nature, but in no event less than a reasonable standard of care.

- (b) The foregoing confidentiality obligations will not apply to Confidential Information that is required to be disclosed by a court or tribunal, legal process, applicable Law or the rules of any applicable stock exchange, in which case the Receiving Party or its applicable Representatives will promptly notify (in advance of such disclosure, if legally permitted) the Disclosing Party, so that the Disclosing Party can seek (at its own cost) a protective order to protect the confidentiality of the Confidential Information to be disclosed. In any such case, the Receiving Party or its Representative will disclose only the minimum Confidential Information that its legal counsel advises it is legally required to disclose.
- (c) Each Party agrees that, should the foregoing confidentiality or use obligations be breached, money damages may be inadequate to remedy such a breach, and the other Party will be entitled to seek, and a court of competent jurisdiction may grant, specific performance and injunctive or other equitable relief as a remedy for any such breach or threatened breach. That remedy will be in addition to all other remedies, including money damages, available to a non-breaching Party at law or in equity.
- (d) The obligations under this Section 5.7 will survive *** (***) years following the Term.
- 5.8. <u>Publicity</u>. Neither Party will make any press release or other public disclosure regarding this Agreement or the transactions contemplated by this Agreement that mentions or identifies the other Party without the other Party's prior written consent, except as required by a Governmental Authority and applicable Law, in which case the Party required to make the press release or public disclosure will use commercially reasonable efforts to obtain the approval of the other Party as to the form, nature and extent of the press release or public disclosure prior to issuing the press release or making the public disclosure.

Page 17 of 27

5.9. <u>Obligation to Terminate, and Refrain From Enter Into, Conflicting Agreements</u>. Isologic has disclosed on Schedule 5.1 the existence and material terms of the only of its contracts and other arrangements that conflicts or is inconsistent with the purchase commitments and other obligations set forth in this Agreement ("<u>Conflicting Agreement</u>"). Isologic agrees ***. Isologic ****. For the avoidance of doubt, nothing in this Section 5.9 affects or otherwise denigrates any of Isologic's representations and warranties in this Agreement, including Section 5.1, or its obligations under this Agreement, including its Minimum Purchase Requirements (as modified by the last paragraph of Section 3.1).

ARTICLE 6 TERM AND TERMINATION

- 6.1. <u>Term</u>. Unless the Parties otherwise agree in writing, this Agreement will terminate on the fifth (5th) anniversary of the Effective Date or on any earlier date on which it is terminated pursuant to its terms (the "<u>Initial Term</u>"). Subject to Isologic's and LMIC's compliance in all material respects with all of their respective obligations under this Agreement during the Term (including those described in Article 3), this Agreement will automatically renew for additional consecutive one (1) year renewal terms, unless either Party provides a written notice of non-renewal to the other Party at least *** (***) days in advance of the expiration of the Initial Term or any then-current renewal term, as the case may be. The foregoing renewal terms, if any, and the Initial Term are referred to collectively in this Agreement as the "<u>Term</u>."
- 6.2. Termination. In addition to the other provisions expressly providing rights to terminate this Agreement, this Agreement may be terminated as follows:
 - (a) In the event that any Party materially breaches any of the provisions of this Agreement (including SIEA), the other Party will have the right to terminate this Agreement upon *** (***) days' (or, in the case of nonpayment, *** (***) days') prior written notice, unless that material breach is cured during that *** (***) day (or *** (***) day) period, in which event this Agreement will continue in full force and effect.
 - (b) If any Party institutes for its protection or is made a defendant in any proceeding under bankruptcy, insolvency, reorganization or receivership Law, or that Party is placed in receivership or makes an assignment for benefit of creditors, then the other Party may terminate this Agreement immediately by written notice to the first Party.
 - (c) In the event that LMIC or its Affiliates fails to comply with a final decision or order of an arbitrator or Governmental Authority of competent jurisdiction in respect of a breach by LMIC or its Affiliates of section 6.8 of the APA (as determined by an arbitrator or Governmental Authority of competent jurisdiction), Isologic will have the right to terminate this Agreement upon *** (***) days' prior written notice, unless that failure to comply is cured during that *** (***) day period, in which event this Agreement will continue in full force and effect.

Page 18 of 27

6.3. <u>Consequences of Termination or Expiration</u>. Termination or expiration of this Agreement will be without prejudice to any rights or remedies that have accrued to the benefit of any Party prior to that termination. Without limiting the foregoing, termination or expiration of this Agreement will not terminate Isologic's obligation to pay all invoices for Product shipped during the Term. Termination or expiration of this Agreement will not relieve any Party from its obligations that are expressly indicated to survive the termination of this Agreement. Sections 3.3, 4.3, 4.4, 4.5, 4.6, 4.7, 4.8, 5.3, 5.4, 5.6, 5.7 and 5.8 and Articles 1, 6, 7 (except for Section 7.5) and 8 will survive the termination or expiration of this Agreement for any reason.

ARTICLE 7 INDEMNIFICATION; LIMITATIONS ON LIABILITY

- 7.1. Indemnification by Isologic. Isologic will indemnify and hold harmless LMIC, its Affiliates and their respective directors, officers, employees and agents (collectively, "LMIC Related Persons") from and against any suits, claims, losses, demands, liabilities, damages, costs and expenses (including reasonable attorney's fees and reasonable investigative costs) (collectively, "Liabilities"), whether asserted by any LMIC Related Party or third party, to the extent arising out of or resulting from or in connection with (a) any breach of this Agreement by Isologic (including its Minimum Purchase Requirements) or (b) any negligence or willful misconduct by Isologic, its Affiliates, and their respective directors, officers, employees and agents (collectively, "Isologic Related Persons"), in each case, except to the extent that any of the foregoing arises out of or results from the breach of this Agreement by LMIC or the negligence or willful misconduct of any of the LMIC Related Persons.
- 7.2. Indemnification by LMIC. LMIC will indemnify and hold harmless Isologic Related Persons from and against all Liabilities, whether asserted by any Isologic Related Party or third party, to the extent arising out of or resulting from or in connection with (a) any breach of this Agreement by LMIC or (b) any negligence, or willful misconduct by any of the LMIC Related Persons, except to the extent that any of the foregoing arises out of or results from the breach of this Agreement by Isologic or the negligence or willful misconduct of any of the Isologic Related Persons.
- 7.3. <u>Indemnification Procedures</u>. All claims for indemnification under this Agreement will be asserted and resolved as follows:
 - (a) A party claiming indemnification under this Agreement (the "<u>Indemnified Party</u>") will promptly notify in writing the Party from whom indemnification is sought (the "<u>Indemnifying Party</u>") of any claim against the Indemnified Party that could give rise to a right of indemnification under this Agreement. For third party claims, the Indemnifying Party will have the right to defend, at its sole cost and expense, that third party claim, on its own behalf and on the behalf of the Indemnified Party, by all appropriate proceedings, which proceedings will be prosecuted diligently by the Indemnifying Party to a final conclusion or settled at the discretion of the Indemnifying Party; provided, however, that the Indemnifying Party may not enter into any compromise or settlement unless the

Page 19 of 27

Indemnified Party consents thereto, which consent will not be unreasonably withheld, delayed or conditioned. If requested by the Indemnifying Party, the Indemnified Party will, at the sole cost and expense of the Indemnifying Party (excluding the internal costs and expenses of the Indemnified Party), cooperate with the Indemnifying Party and its counsel in contesting any third party claim that the Indemnifying Party elects to contest, including, without limitation, the making of any related counterclaim against the Person asserting the third party claim or any cross-complaint against that person.

- (b) Notwithstanding the Indemnifying Party's election to assume the defense of any third party claim, the Indemnified Party will have the right to employ separate counsel and to participate in the defense of that third party claim; provided that the Indemnifying Party will bear the reasonable and documented costs and expenses of that separate counsel if (i) the use of counsel chosen by the Indemnifying Party to represent both the Indemnifying Party and the Indemnified Party would present that counsel with a conflict of interest, (ii) the actual or potential defendants in, or targets of, any such third party claim include both the Indemnifying Party and the Indemnified Party will have reasonably concluded that there may be a legal defense available to it which is different from or additional to the defenses available to the Indemnifying Party (in which case the Indemnifying Party will not have the right to assume the defense of that third party claim on behalf of the Indemnified Party), (iii) the Indemnifying Party will not have employed counsel reasonably satisfactory to the Indemnified party to represent the Indemnified Party within a reasonable time after notice of the institution of that third party claim or (iv) the Indemnifying Party authorizes the Indemnified Party to employ separate counsel at the Indemnifying Party's cost and expense.
- (c) If the Indemnifying Party fails to notify the Indemnified Party within ten (10) days after receipt of notice in accordance with Section 7.3(a) (or any shorter period necessary to respond to that claim) that the Indemnifying Party elects to defend the Indemnified Party pursuant to this Section 7.3, or if the Indemnifying Party elects to defend the Indemnified Party pursuant to this Section 7.3 but fails to defend the Indemnifying Party elects to defend the Indemnifying Party claim diligently and promptly, then the Indemnified Party will have the right to defend, at the sole cost and expense of the Indemnified Party with respect to a third party claim for which the Indemnified Party is entitled to indemnification under this Agreement.
- 7.4. <u>Limitations on Liability</u>. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY AND NOTWITHSTANDING ANYTHING PROVIDED FOR UNDER APPLICABLE LAW TO THE CONTRARY, IN NO EVENT WILL EITHER PARTY BE LIABLE TO THE OTHER PARTY OR TO ANY THIRD PARTY FOR ANY INDIRECT, SPECIAL, TREBLE, PUNITIVE, EXEMPLARY OR CONSEQUENTIAL DAMAGES, OR FOR ANY AMOUNTS REPRESENTING LOSS OF PROFITS OR LOSS OF BUSINESS, INCLUDING ANY SUCH DAMAGES

Page 20 of 27

RESULTING FROM DELAYS IN DELIVERY, OR FAILURE TO DELIVER, ANY PRODUCT, REGARDLESS OF WHETHER THAT FIRST PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF THOSE DAMAGES (IN EACH CASE, OTHER THAN ANY SHORTFALL PAYMENTS PAYABLE UNDER THIS AGREEMENT AND EXCLUDED CLAIMS).

NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY AND NOTWITHSTANDING ANYTHING PROVIDED FOR UNDER APPLICABLE LAW TO THE CONTRARY, LMIC'S AGGREGATE LIABILITY UNDER THIS AGREEMENT WILL BE LIMITED TO ***.

7.5. <u>Insurance</u>. Each Party will maintain, at all times during the Term, standard general commercial liability insurance and professional liability/errors and omissions insurance covering negligent acts, errors and omissions in the performance of services from a reputable insurance company. Coverage under each such policy will be in a reasonable amount and consistent with, or more comprehensive than, industry standards with respect to such Party's obligations under this Agreement. All policies of insurance required to be maintained by either Party under this Agreement will be primary and non-contributory with any other insurance and/or self-insurance carried by the other Party or its Affiliates. Upon request, each Party will provide evidence to the other Party of the coverage or self-insurance required of such Party under this Agreement. The minimum level of insurance or self-coverage set forth in this Agreement will not be construed to create a limit on such Party's liability under this Agreement.

ARTICLE 8 MISCELLANEOUS

- 8.1. <u>Governing Law</u>. This Agreement (and all claims, controversies or causes of action (whether in contract, tort or otherwise) that may be based upon, arise out of or in connection with, or relate to, this Agreement or its negotiation, execution or performance, including any claim, controversy or cause of action that may be based upon, arise out of or in connection with, or relate to, any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) will be governed by, and construed in accordance with, the laws of the State of New York in the United States, without giving effect to any principles of conflicts of laws that would require or permit the application of a different law.
- 8.2. <u>Entire Agreement</u>. This Agreement (together with its exhibits (including the SIEA), which are incorporated by reference in this Agreement) constitutes the sole and entire agreement of the Parties with respect to the subject matter of this Agreement, and supersedes and prevails over all prior and contemporaneous understandings, agreements, representations and warranties (whether written or oral), with respect to that subject matter. The Parties agree that neither Party has relied on (and neither Party will have any remedies or causes of action (whether in contract, tort or otherwise) for) any statements, communications, disclosures, failures to disclose, representations, warranties or agreements of the other Party (or of any other person acting on the other Party's behalf) that is not expressly set forth in this Agreement, including any representations, warranties or agreements arising from statute or otherwise in law.

Page 21 of 27

- 8.3. <u>Severability</u>. In the event that any provision of this Agreement is deemed to be invalid, illegal or unenforceable in any jurisdiction, then (i) the validity, legality or enforceability of the remaining provisions of this Agreement will not be affected or impaired in any way by that provision and (ii) the Parties will negotiate in good faith to replace that provision with a valid and enforceable one that effects the original intent of the Parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated by this Agreement be consummated as originally contemplated to the greatest extent possible.
- 8.4. <u>Relationship of the Parties</u>. In making and performing this Agreement, the Parties are acting, and intend to be treated, as independent entities and nothing contained in this Agreement will be construed or implied to create an agency, partnership, joint venture, franchise or employer-employee relationship between LMIC and/or any of its employees, independent contractors and/or agents, on the one hand, and Isologic and/or any of its employees, independent contractors and/or agents, on the other hand. Neither Party will have any express or implied right or authority to (i) make any representation, warranty or commitment, (ii) assume, create or impose any obligations, contracts, agreements or undertakings, (iii) incur any charges or expenses, in each case, for, on behalf of, or in the name of, the other Party, unless expressly so authorized in writing by the other Party.
- 8.5. Force Majeure. Each of the Parties will be excused from the performance of its obligations under this Agreement (except for any obligations to make payments to the other Party under this Agreement) for so long as, and neither Party will be liable to the other Party or to any third Person in the event that, its performance is prevented or delayed due to a Force Majeure Event. The Party suffering the occurrence of a Force Majeure Event will notify the other Party as soon as reasonably practicable, stating the period for which that Force Majeure Event is expected to continue, and any time for performance under this Agreement will be extended by the actual time of delay caused by that Force Majeure Event; provided that, (i) Isologic will be allowed to purchase from alternate suppliers those quantities of Products that LMIC is unable to supply under this Agreement on account of any Force Majeure Event that lasts for more than *** (***) days, then Isologic will have the option at any time during the continuation of LMIC's inability to supply that Product under this Agreement on account of the Force Majeure Event the term of this Agreement beyond the Term.
- 8.6. <u>Arbitration</u>. Any dispute, controversy or claim arising out of or relating to compliance with, or breach or alleged breach, interpretation or validity of, this Agreement or otherwise relating to the Parties (each a "<u>Dispute</u>") will be exclusively resolved by binding arbitration, which arbitration may be commenced by sending a written notice to the other Party demanding arbitration of that Dispute (the "<u>Demand</u>"). In that event, the Dispute will be finally resolved by arbitration in accordance with the United States

Page 22 of 27

Arbitration Act and the Commercial Arbitration Rules of the American Arbitration Association. The place of the arbitration will be New York, New York. The arbitration will be conducted in the English language before a panel of three arbitrators. Each Party will name one arbitrator, and the two so named will name the third arbitrator, who will act as chairman. If the two party arbitrators cannot agree on a third arbitrator within *** (***) days after the Demand, the third arbitrator will be selected by the American Arbitration Association. The arbitrators will promptly meet, fix the time, date and place of the hearing and notify the Parties. The arbitration will be conducted within *** (***) days after any Demand. The panel of arbitrators will promptly transmit an executed copy of its decision to the Parties. The decision of the arbitrators will be final, binding and conclusive upon the Parties. Judgment on the award rendered by the arbitrators may be entered in any court having jurisdiction thereof. Each Party retains the right to seek from a court any interim or provisional relief that may be necessary to protect the rights or property of that Party pending the establishment of the arbitrators' determination of the merits of the controversy, and any such action 8.6 are specifically enforceable and will survive any termination of this Agreement. All awards are subject to Section 7.4; provided that the arbitrators and legal counsel incurred in the arbitration proceedings, or the arbitrators may award the costs on a distributive basis that apportions costs on an issue-by-issue basis and based on the inverse proportion that any amount actually contested but not awarded to Isologic or LMIC bears to the aggregate amount actually contested by Isologic and LMIC, respectively.

8.7. <u>Notices</u>. All notices, requests, demands, consents, approvals and other communications to any Party required to be given or delivered under, or by reason of, this Agreement will be in writing, and will be deemed to have been given when actually received (or refused) by the addressee after being sent by personal delivery, certified or registered mail, reputable overnight express courier or facsimile (with hard copy to follow) to the address for the addressee set forth below:

Notices to LMIC:

Lantheus MI Canada, Inc. 1111 Dr. Frederik-Phillips Boulevard Suite 100 Montreal, Quebec, H4M 2X6 Canada Attn: Vice President, International

Page 23 of 27

with a copy to:

Lantheus Medical Imaging, Inc. 331 Treble Cove Road North Billerica, Massachusetts 01862 United States of America Attn: Vice President, General Counsel

Notices to Isologic:

Isologic Innovative Radiopharmaceuticals Ltd. 1855 32e Avenue, Montreal, Quebec, Canada H8T 3J1 Canada Attn: Andre Gagnon

with a copy to:

DW Healthcare Partners 66 Wellington St. W., Suite 4030 Toronto, Ontario, Canada M5K 1J5 Attn: Sameer Mathur

Either Party may change its notice address by giving a notice to the other Party pursuant to this Section 8.7. Notwithstanding the foregoing, all reporting for Adverse Events, Serious Adverse Events, Special Situations and Product Quality Complaints will be made pursuant to Section 5.6.

- 8.8. <u>Successors and Assigns: Assignment</u>. Neither this Agreement, nor any right, interest or obligation under this Agreement, may be assigned or otherwise transferred by either Party (whether by contract, operation of law or otherwise), in whole or in part, without the prior written consent of the other Party, which consent may be withheld in the sole discretion of the other Party; provided, however, that:
 - (a) either Party may assign or otherwise transfer any or all of its rights, or delegate any or all of its respective duties or obligations, under this Agreement without the consent of the other Party, to an Affiliate of such Party (but only for as long as such Person remains such Party's Affiliate, and provided that such Person has the capabilities and legal authorization necessary to perform the assigned duties and obligations), it being agreed that no such assignment to a Party's Affiliate will release the assigning Party from its obligations under this Agreement;
 - (b) each Party will assign or otherwise transfer all of its applicable rights, and delegate all of its applicable duties and obligations, under this Agreement without the consent of the other Party, to (i) an acquirer of, or successor to, (1) all or

Page 24 of 27

substantially all of the assets of that Party (or any of its parent companies) or (2) all or substantially all of any business line to which this Agreement relates or (ii) the surviving entity in any merger, consolidation, equity exchange or reorganization to which (1) that Party (or any of its parent companies) or (2) that business line is a party; and

(c) either Party may assign or otherwise transfer any or all of its rights, or delegate any or all of its duties or obligations, under this Agreement without the prior written consent of the other Party for the benefit of any lenders under any financing arrangement of the first-mentioned Party or its Affiliates;

provided, in each case, the assigning Party provides to the other Party written notice of such assignment or transfer as soon as reasonably possible and the assignee, acquirer, successor or surviving entity, as the case may be, agrees in writing to be bound (or by operation of law is bound) by all of the obligations of that Party under this Agreement. Any assignment or transfer in violation of this Agreement will be null and void and have no force or effect. This Agreement will be binding upon and inure to the benefit of the Parties, and its respective successors and assigns as permitted under this Agreement.

- 8.9. <u>Third Party Beneficiaries</u>. None of the provisions of this Agreement will be for the benefit of or enforceable by any third party (other than the LMIC Related Parties and the Isologic Related Parties with respect to Article 7) including any creditor of any Party. No third party will obtain any right under any provision of this Agreement or will by reason of any such provision make any claim in respect of any debt, liability or obligation (or otherwise) against any Party.
- 8.10. <u>Modifications</u>. Except as otherwise expressly provided, this Agreement may only be amended, modified or supplemented by an agreement in a writing signed by each of the Parties. The Parties agree that, in the event that there is a change in Law that affects (or may affect) the legality or enforceability of this Agreement or any of its provisions or that materially and adversely affects the ability of any Party to perform its obligations or receive the benefits intended under this Agreement, then, as soon as reasonably practical following written notice thereof, the Parties will negotiate reasonably and in good faith (provided that neither Party will have any obligation to enter into) an amendment or substitute agreement in order to best reflect the original intent of the Parties in a manner consistent with that change in applicable Law.
- 8.11. <u>Waivers</u>. No waiver of any provision of this Agreement will be effective unless it is explicitly set forth in writing and signed by the Party so waiving. No failure or delay of a Party in exercising any right, remedy, power or privilege arising from this Agreement will operate or be construed as a waiver thereof. No single or partial waiver in any one or more instances will be deemed to constitute a further or continuing waiver in other instances or a waiver of any provision not expressly identified by that written waiver, whether of a similar or different character, and whether occurring before or after that waiver.

Page 25 of 27

8.12. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which will be deemed to be an original, and those counterparts will together constitute one and the same instrument. A facsimile transmission of an executed counterpart signature page will be deemed an original.

[The remainder of this page is left blank intentionally.]

Page 26 of 27

IN WITNESS WHEREOF, the Parties, intending to be legally bound, have duly executed this Agreement as of the day and year first written above.

LANTHEUS MEDICAL IMAGING, INC.

By:

Name: Cyrille Villeneuve Title: Vice President, International

ISOLOGIC INNOVATIVE RADIOPHARMACEUTICALS LTD.

By:

Name: Sameer Mathur Title: Director

Page 27of 27

<u>Exhibit A</u>

Product Schedule

<u>Exhibit B</u>

Form of Written Report and Certification

Exhibit C

Foreign Corrupt Practices Act and Anti-Bribery Compliance Policy

<u>Exhibit D</u>

Safety Information Exchange Agreement

Schedule 5.1

Conflicts

<u>Exhibit D to</u> <u>Amended and Restated Purchase Agreement</u>

FORM OF TRANSITION SERVICES AGREEMENT

This **Transition Services Agreement** (this "**Agreement**"), dated as of January 13, 2016 (the "**Closing Date**"), is made by and between Lantheus MI Canada, Inc., a Canadian corporation ("**Seller**"), and Isologic Innovative Radiopharmaceuticals, Ltd., a Canadian corporation ("**Buyer**") (Seller and Buyer being also individually designated as a "**Party**" or collectively as the "**Parties**"). Capitalized terms are defined in Section 1.

WITNESSETH:

WHEREAS, pursuant to the terms of that certain Asset Purchase Agreement between Seller and Buyer, dated as of the date of this Agreement (as may be amended, modified and/or supplemented from time to time, the "**Purchase Agreement**"), Seller has sold, conveyed, assigned and transferred to Buyer certain Purchased Assets in connection with the operation of the Radiopharmacy Business in the Territory; and

WHEREAS, in connection with the Purchase Agreement, Buyer desires, and Seller is willing, to provide certain transition services during the Transition Period with respect to the Radiopharmacy Business on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the agreements and covenants contained herein, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

1. **DEFINITIONS**

All capitalized terms in this Agreement (whether used in the singular or the plural), unless otherwise defined herein, will have the meanings as set forth in the Purchase Agreement. In addition, the following terms as used in this Agreement, whether used in the singular or the plural, will have the meanings as set forth in this Section 1. References to "Sections" and "subsections" in this Agreement will be to Sections and subsections, respectively, of this Agreement unless otherwise specifically provided.

- 1.1 "Agreement" will have the meaning given to such term in the introductory paragraph to this Agreement.
- 1.2 "Authorized Purpose" will have the meaning given to such term in Section 12.2.
- 1.3 "Buyer" will have the meaning given to such term in the introductory paragraph to this Agreement.
- 1.4 "Buyer Related Parties" will have the meaning given to such term in Section 9.1.
- 1.5 **"Confidential Information"** will have the meaning given to such term in Section 12.1.
- 1.6 "Covered Product" will mean the drug GLUDEF® (the rights to which form part of the Purchased Assets).

- 1.7 "cGMP" will mean Health Canada's current Good Manufacturing Practice requirements as promulgated under the Food and Drugs Act (R.S.C. 1985, c. F-27) and Food and Drug Regulations (C.R.C., c. 870) at Division 2 of Part C, and as further defined by Health Canada guidance documents, as such may be amended from time to time, applicable to the operation of the Radiopharmacy Business.
- 1.8 "Closing Date" will have the meaning given to such term in the introductory paragraph to this Agreement.
- 1.9 "Demand" will have the meaning given to such term in Section 13.6.
- 1.10 "Disclosing Party" will have the meaning given to such term in Section 12.1.
- 1.11 "Dispute" will have the meaning given to such term in Section 13.6.
- 1.12 **"Force Majeure Event"** means, with respect to an affected Party, any circumstances or events that are beyond the reasonable control of any one or more of: that Party, its respective Affiliates, or any of their respective vendors, suppliers or shipping carriers (including, in each case to the extent such circumstances or events are beyond the reasonable control of the affected Party, any (a) act of God, (b) natural disaster or severe weather condition (e.g., lightning, earthquakes, hurricanes, floods, tornadoes, drought, blizzards, ice storms, volcanic eruption, epidemic, etc.), fire or explosion, (c) war, invasion, hostilities (whether war is declared or not), terrorist threat or act, riot, rebellion, mutiny, sabotage or other civil unrest, (d) act or decision of any Governmental Authorities or change in applicable Law, (e) sinking, crashing, embargo or blockade, (f) strikes, labor disturbances, stoppages or slowdowns or other industrial disturbances, (g) failure or delay of public utilities or common carriers, (h) batch failure, supply failure or outage, equipment failure or malfunction, shortages of fuel, power or raw materials or (i) any other circumstance or event that is not under the reasonable control of the affected Party).
- 1.13 "Indemnified Party" will have the meaning given to such term in Section 9.2.
- 1.14 "Indemnifying Party" will have the meaning given to such term in Section 9.2.
- 1.15 **"Losses"** will have the meaning given to such term in Section 9.1.
- 1.16 "Party" and "Parties" will have the meaning given to such term in the introductory paragraph to this Agreement.
- 1.17 "Personal Information" will have the meaning given to such term in Section 12.4.
- 1.18 "Purchase Agreement" will have the meaning given to such term in the recitals to this Agreement.
- 1.19 "Receiving Party" will have the meaning given to such term in Section 12.1.
- 1.20 "Sales Taxes" will have the meaning given to such term in Section 8.3.

- 2 -

- 1.21 "Seller" will have the meaning given to such term in the introductory paragraph to this Agreement.
- 1.22 "Seller Employees" will mean any employees used by Seller to render the Transition Services.
- 1.23 "Seller Related Parties" will have the meaning given to such term in Section 9.2.
- 1.24 **"Term"** will have the meaning given to such term in Section 10.
- 1.25 "Transition Service" and "Transition Services" will have the meanings given to such terms in Section 2.1.
- 1.26 **"Transition Period"** will mean, with respect to each Transition Service, the period commencing on the Closing Date and ending on the earlier of (a) the expiration date specified for that Transition Service on <u>Schedule A</u> (as such expiration date may be amended from time to time by mutual agreement of the Parties) and (b) the date on which this Agreement is terminated with respect to that Transition Service in accordance with the terms, and subject to the conditions, of this Agreement (including the proviso in Section 10.1).
- 1.27 "Transition Services" will have the meaning given to such term in Section 2.1.

2. TRANSITION SERVICES

- 2.1 **General.** Subject to the terms and conditions of this Agreement, during the applicable Transition Period, Buyer hereby retains Seller to provide, and Seller hereby agrees to provide (or cause an Affiliate to provide) to Buyer, the transition services relating to the operation of the Radiopharmacy Business in the Territory specifically set forth on <u>Schedule A</u> (each, a "**Transition Service**" and, collectively, the "**Transition Services**").
- 2.2 **Buyer Diligence Obligation**. Buyer will develop and implement a transition for the Radiopharmacy Business from the Transition Services to a standalone basis (including through engaging the services of third parties) as soon as reasonably practicable after Closing (and, in any event, no later than *** (***) days after the Closing Date (except that such *** (***) day period shall be extended with respect to any Transition Services that are required to be provided in order for Buyer to comply in all respects with all applicable Laws and Scheduled Approvals, until all Assigned Approvals have been transferred to Buyer and the Buyer has obtained all other Scheduled Approvals, provided that such extension shall apply only to the extent that the failure of the Buyer to have obtained the Scheduled Approvals within such *** (***) day period (and during the period of any such extension) is outside the reasonable control of the Buyer)) and in a manner that is consistent with the terms and conditions of this Agreement.

3. EMPLOYEES

3.1 The Transition Services will be performed by the individuals designated by Seller from time to time, at its sole discretion.

- 3 -

3.2 The Seller Employees will remain employees of Seller and nothing contained in this Agreement is intended to create nor will be construed as creating the relationship of employer and employee between Seller Employees and Buyer. During the Term, Seller will maintain, at Seller's expense, Workers Compensation and Employers' Liability insurance in compliance with the local Law requirements of the province/jurisdiction in which the Transition Services are performed under this Agreement.

4. OBLIGATIONS OF THE PARTIES

- 4.1 **Compliance with Laws.** Each Party will, and will cause its respective employees and officers, to comply with all Laws which may be applicable to the Transition Services.
- 4.2 **Standard of Performance.** Seller will provide the Transition Services (a)(i) in substantially the same manner as it performed those services for itself in conducting the Radiopharmacy Business during the twelve (12) months immediately prior to the Closing Date or (ii) in a manner that provides Buyer with substantially the same benefits, functionality or use as Seller had received in conducting the Radiopharmacy Business during the twelve (12) months immediately prior to the Closing Date (in the case of each of clauses (a)(i) and (a)(ii) so long as it does not unreasonably interfere (taking into account that Seller is undergoing a transition) with the operations of the Retained Business or the Seller or its Affiliates) and, (b) as applicable, in accordance with cGMP.
- 4.3 Governance Principles. Buyer acknowledges that Seller will have sole control over the manner in which the Transition Services are performed, except for the specific obligations expressly set forth in this Agreement. Such Transition Services performed by Seller for the benefit of Buyer are provided for the sole benefit of the Radiopharmacy Business purchased by Buyer from Seller under the Purchase Agreement. Seller will also be responsible for the proper management of, and control over, the provisions of the Transition Services (including, without limitation, the determination or designation, at any time, of the equipment, employees and other resources of Seller or Buyer to be used in connection with the provisions of the Transition Services, Seller, as it deems appropriate and acting reasonably, may, subject to Section 4.2, (i) use its own personnel, equipment, facilities, systems and other resources or, so long as it does not unreasonably interfere (taking into account that Buyer or its Affiliates, the Purchased Assets or equipment, facilities, systems and other resources of Buyer; (ii) employ the services of a subcontractor (subject to the approval of the Buyer (not to be unreasonably withheld, conditioned or delayed) and provided that Seller shall be responsible for the performance of any such subcontractor, and any subcontractor must agree in writing to be bound by confidentiality obligation at least as protective as the terms of this Agreement regarding confidentiality); and/or (iii) provide substitutes for said personnel, equipment and subcontractors, as well any other resources employed for the provide substitutes for said personnel, equipment and subcontractors, as well any other resources employed for the provision of the provision of the provide substitutes for said personnel, equipment and subcontractors, as well any other resources employed for the provision of the provide substitutes for said personnel, equipment and subcontractors, as well any other resources employed for the provision of t
- 4.4 **Consent.** Seller will obtain, prior to the Closing Date, all consents, waivers, extension of contracts and other permissions required to enable Seller to provide, and Buyer to

- 4 -

receive, the Transition Services; provided that Seller will not be obligated to incur any costs in obtaining any such consents. If Seller is unable to secure the required consent, waiver, extension or other permission, the Parties must negotiate in good faith and use their respective reasonable endeavours to agree upon the reasonable and lawful means by which Seller may make available to Buyer the benefit of the Transition Services to which that consent, waiver, extension or other permission relates; provided that, in the event that the Parties are unable to so agree, then Buyer will be excused of its obligations to provide such Transition Services under this Agreement.

- 4.5 **Collaboration from the Buyer.** Buyer acknowledges that the provision of some of the Transition Services may require its reasonable collaboration, instructions and information and other reasonable support and assistance, which Buyer will use commercially reasonable efforts to provide to Seller, at Buyer's cost and expense on a timely basis, failing which Seller will be excused from its obligations to provide such Transition Services to the extent of any impact resulting from such failure.
- 4.6 **Buyer Obligations.** In order to permit performance of the Transition Services by Seller, Buyer will (in each case, so long as it does not unreasonably interfere (taking into account that Seller is undergoing a transition) with the operations of the Radiopharmacy Business or the Buyer or its Affiliates):
 - (i) permit Seller Employees to have reasonable access to the Purchased Assets (including the radiopharmacies forming part of the Purchased Assets and all related equipment, premises, facilities, systems and other resources) as reasonably necessary for the provision of the Transition Services; provided that Buyer will not be required to grant access or furnish information to the Seller to the extent that such assets or information: (i) is subject to solicitor/client or solicitor work product privileges unless Seller agrees to enter into a common interest privilege agreement to protect such privilege, (ii) consists of personnel records of Buyer relating to individual performance or evaluation records, medical histories or other information that in Buyer's good faith opinion is sensitive (including commercially or competitively sensitive information) or the disclosure of which could subject Buyer to risk of liability, or (iii) is subject to an obligation of confidentiality or non-use in favour of a third party that does not entitle Buyer to disclose to Seller, or Seller to use, as a representative of Buyer thereunder; provided that if such access or information is denied for such reasons, then Seller will be excused from its obligations to provide the impacted Transition Services to the extent of any impact resulting from such refusal;
 - (ii) make available to Seller Employees the reasonable cooperation of the Transferred Employees or Buyer's other employees involved in integration and transition efforts, in each case, as reasonably necessary for the provision of the Transition Services; and
 - (iii) provide such other resources (including personnel, equipment, tools and supplies) and commercially reasonable efforts as reasonably necessary for the provision of the Transition Services.

- 5 -

5. RELATIONSHIP WITH CUSTOMERS OF COVERED PRODUCT

- 5.1 **Customers.** In order to inform customers of the change in ownership of the Radiopharmacy Business and to otherwise transfer the sales, distribution and customer support services for the Covered Product to Buyer, the Parties will (by a method and at a time mutually agreed to by the Parties (but not later than the end of the Term) provide notice to all contracted customers of the transfer of the sales, distribution and customer support services for the Covered Product to Buyer which allows adequate transition time for such contracted customers.
- 5.2 **Terms and conditions.** As of the Closing Date, all Covered Product will be sold under and pursuant to Buyer's standard terms and conditions as the same may be in effect from time to time. It will be Buyer's responsibility to inform customers of such terms and conditions at the time Buyer accepts orders for such Covered Product.
- 5.3 **Pricing of Covered Product.** As of the Closing Date, pricing of Covered Product will (subject to its contractual obligations) be solely within the control of Buyer.

6. OWNERSHIP OF WORK PRODUCT

- 6.1 **Ownership.** This Agreement and the performance of the Transition Services hereunder will not affect the ownership of any assets (including Purchased Assets) allocated in the Purchase Agreement. Subject to the foregoing, Seller's proprietary information, materials and other property and property rights existing prior to the execution of this Agreement or produced, developed, created or reduced to practice by Seller in providing the Transition Services (including such work product, data, tools, or other information, materials, and other property rights produced, developed, created, or reduced to practice by Seller in providing the Services), in each case, other than customer transaction data of Buyer, are the Seller's exclusive property and, as between the Parties, will remain exclusively owned by Seller.
- 6.2 **Grant of License.** Subject to any restrictions imposed in any applicable agreement between Buyer and a third-party licensor or service provider of which Buyer provides notice to Service Provider, Buyer hereby grants to Seller and its Affiliates a royalty-free, nonexclusive license to use, access, duplicate, modify, display, and distribute internally within Seller the materials that Buyer provides to the Seller to enable or facilitate its performance of Transition Services only to the extent, and only for so long as, reasonably required to provide the Transition Services.

7. ACCESS TO DATA AND RETURN OF PROPERTY

7.1 Access to Data. The Seller will have the right to reasonably access all of the Buyer's data in the possession of Seller as a result of this Agreement at any time during the Term only to the extent reasonably necessary for the provision of the Transition Services, except for data that (i) is subject to solicitor/client or solicitor work product privileges unless Seller agrees to enter into a common interest privilege agreement to protect such privilege, (ii) consists of personnel records of Buyer relating to individual performance or evaluation

- 6 -

records, medical histories or other information that in Buyer's good faith opinion is sensitive (including commercially or competitively sensitive information) or the disclosure of which could subject Buyer to risk of liability, or (iii) is subject to an obligation of confidentiality or non-use in favour of a third party that does not entitle Buyer to disclose to Seller, or Seller to use, as a representative of Buyer thereunder; provided that if such access or information is denied for such reasons, then Seller will be excused from its obligations to provide the impacted Transition Services to the extent of any impact resulting from such refusal.

- 7.2 **Isolation of Business Records**. Each of the Parties will ensure that it is able to isolate the other Party's data and records in the possession of one Party as a result of this Agreement from those of the other Party.
- 7.3 **Extraction of Business Records**. In addition to any regular data extraction provided as part of the Transition Services, Seller will make appropriate arrangements for the separation and delivery to Buyer of Buyer's data, records and items in process in the possession of Seller as a result of this Agreement. Such separation and delivery will take place as promptly as practicable after the end of the Term or the termination of the relevant Transition Services, as the case may be.
- 7.4 **Return of Property**. Seller agrees that on the expiration or termination of this Agreement, for any cause whatsoever, Seller will return to Buyer, as promptly as practicable, any of Buyer's property in Seller's possession or under Seller's control as a result of this Agreement. Buyer agrees to the same obligation in relation to Seller's property in Buyer's possession or under Buyer's control.

8. FEES AND PAYMENT

- 8.1 Fees for Transition Services. Buyer will pay Seller the fees set forth on <u>Schedule A</u> for each of the Transition Services performed during the Term.
- 8.2 **Expenses.** Buyer will also reimburse Seller for any documented out-of-pocket expenses (including fees paid to any independent contractors, service providers, vendors or licensors) set forth on Schedule A or that are otherwise approved in writing by the Buyer and are reasonably incurred in rendering the Transition Services or otherwise performing its obligations under this Agreement, including any third party licensing, service or maintenance fees payable with respect to any Transition Service. <u>Schedule A</u> sets forth a summary of the Seller's reasonable identification and estimate of any out-of-pocket expenses it expects to incur in rendering the Transition Services. Notwithstanding the foregoing, Buyer shall not reimburse Seller for any expenses incurred in fulfilling its obligations under the Purchase Agreement (except to the extent that the Purchase Agreement provides that Buyer shall reimburse such expenses).
- 8.3 **Invoicing and Payment.** Within *** (***) days after the end of each calendar month, Seller will provide Buyer with a written invoice for all fees and expenses related to the Transition Services rendered for the prior period. Each such invoice will be reasonably detailed. Seller will provide such additional supporting information and documentation as may be reasonably requested by Buyer. Such invoices will be paid by Buyer within *** (***) days of receipt of such invoice.

-7-

8.4 Taxes. The amounts payable by Buyer to Seller, as set out in this Agreement, do not include any goods and services, harmonized sales, value-added, sales, use, consumption, multi-staged, ad valorem, personal property, customs, excise, stamp, transfer, or similar taxes, duties, or charges, (collectively, the "Sales Taxes"). All Sales Taxes related to the Services are Buyer's responsibility and will be paid concurrent with the payment of any consideration payable pursuant to this Agreement, unless Buyer qualifies for an exemption from any such applicable Sales Taxes, in which case Buyer will, in lieu of payment of such applicable Sales Taxes, deliver to Seller such certificates, elections or other documentation required by Law or the administration thereof to substantiate and affect the exemption claimed by Recipient. All invoices will contain the information prescribed in the *Input Tax Credit (GST/HST) Regulations* or equivalent provincial legislation. Seller shall have the responsibility for remittance of all such Sales Taxes payable and actually received from Buyer to the appropriate taxing authority, and shall be liable for any penalties, interest and other charges and any other fees or costs arising from Seller's failure to remit any such amounts actually received to the authority. Buyer will not have the right to set off any amounts due and payable under this Agreement against any liabilities arising or amounts owed by Seller under any of the Purchase Agreement, except to the extent of a final resolution of such claims in its favor (whether such resolution is by mutually agreement or pursuant to a final and binding arbitration decision rendered in accordance with the terms of the Purchase Agreement).

9. INDEMNIFICATION

- 9.1 Scope of Indemnification by Buyer. Buyer will defend, indemnify and hold harmless Seller, its Affiliates and its and their directors, officers, shareholders, employees and agents and their respective successors and permitted assigns (collectively, "Buyer Related Parties") from any and all third party claims, actions, causes of action, liabilities, losses, costs, damages or expenses, including reasonable attorneys' fees (collectively, "Losses") (including, for greater certainty, Losses arising out of the bodily injury or death of any person (including the Seller Employees)), to the extent such Losses directly or indirectly arise out of or relate to (i) Seller's provision of Transition Services in accordance with this Agreement, (ii) Buyer's material breach of any of its obligations, warranties or representations hereunder, (iii) Buyer's negligence or willful misconduct, or (iv) the actions of the Buyer, after the Closing Date but prior to all Assigned Approvals having been transferred to Buyer and the Buyer having obtained all other Scheduled Approvals, with respect to the marketing and promotion of the Covered Product or the Buyer's non-compliance with applicable Laws applicable to marketing and promotion of the Covered Product or the extent such Losses directly or indirectly arise out of or relate to the negligence, willful misconduct or breach of this Agreement of any of the Seller Related Parties.
- 9.2 Scope of Indemnification by Seller. Seller will defend, indemnify and hold harmless Buyer, its Affiliates and its and their directors, officers, shareholders, employees and agents, and their respective successors and permitted assigns (collectively, "Seller

- 8 -

Related Parties") from any and all Losses to the extent such Losses directly or indirectly arise out of or relate to (i) Seller's material breach of its obligations, warranties or representations hereunder, (ii) Seller's negligence or willful misconduct of the Seller in the performance of its obligations under this Agreement, in each case, except to the extent such Losses directly or indirectly arise out of or relate to the negligence, willful misconduct or breach of this Agreement of any of the Buyer Related Parties.

- 9.3 **Claim.** A Party seeking indemnification pursuant to Section 9.1 or 9.2 (an "**Indemnified Party**") will give prompt written notice to the Party from whom such indemnification is sought (the "**Indemnifying Party**") of the assertion of any claim, or the commencement of any action or proceeding, in respect of which indemnity may be sought hereunder. The Indemnifying Party will have the right to, and will at the request of the Indemnified Party, assume the defense, with counsel reasonably satisfactory to the Indemnified Party, of any such suit, action or proceeding at its own expense. An Indemnifying Party will not be liable under Section 9.1 or 9.2 for any settlement effected without its consent of any claim, litigation or proceeding in respect of which indemnity may be sought hereunder, which consent will not be unreasonably withheld. Neither Party will be responsible to or bound by any settlement made by the other Party without its prior written consent.
- 9.4 Limit of Liability. EXCEPT IN THE EVENT OF A PARTY'S GROSS NEGLIGENCE OR WILLFUL MISCONDUCT, IN NO EVENT WILL EITHER PARTY OR ITS AFFILIATES OR THEIR RESPECTIVE DIRECTORS, OFFICERS, EMPLOYEES OR AGENTS BE LIABLE TO THE OTHER PARTY OR ITS AFFILIATES, OR THEIR RESPECTIVE DIRECTORS, OFFICERS, SHAREHOLDERS, EMPLOYEES OR AGENTS AND THEIR RESPECTIVE SUCCESSORS AND PERMITTED ASSIGNS, FOR ANY SPECIAL, INDIRECT, PUNITIVE, TREBLE, EXEMPLARY, CONSEQUENTIAL OR INCIDENTAL DAMAGES OR LOSSES OF ANY KIND, NATURE OR DESCRIPTION WHATSOEVER (INCLUDING BUT NOT LIMITED TO LOST PROFITS, LOST REVENUES AND/OR LOST SAVINGS) SUFFERED OR INCURRED BY SUCH PARTY FOR ANY CAUSE WHATSOEVER, REGARDLESS OF WHETHER ARISING FROM BREACH OF CONTRACT, WARRANTY, TORT, STRICT LIABILITY OR OTHERWISE, EVEN IF SUCH PARTY IS ADVISED OF THE POSSIBILITY OF SUCH LOSS OR DAMAGE OR IF SUCH LOSS OR DAMAGE COULD HAVE BEEN REASONABLY FORESEEN.

NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED HEREIN OR AT LAW OR IN EQUITY, EXCEPT IN THE CASE OF WILFUL MISCONDUCT OR FRAUD, IN NO EVENT WILL SELLER BE LIABLE UNDER THIS AGREEMENT FOR ANY LOSSES OF ANY KIND OR NATURE IN EXCESS OF ***.

9.5 **No Representations and Warranties.** THIS IS A SERVICES AGREEMENT. EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT OR IN THE PURCHASE AGREEMENT, INCLUDING ALL SCHEDULES AND OTHER ATTACHMENTS ATTACHED HERETO: (i) THE PARTIES MAKE NO REPRESENTATIONS, WARRANTIES OR CONDITIONS WITH RESPECT TO THE SUBJECT MATTER OF THIS AGREEMENT; AND (ii) THE PARTIES HEREBY

- 9 -

DISCLAIM ALL REPRESENTATIONS, WARRANTIES AND CONDITIONS, EXPRESS, IMPLIED OR OTHERWISE, WITH REGARD TO THIS AGREEMENT (INCLUDING, ALL SCHEDULES HERETO) AND THE TRANSITION SERVICES. TO THE EXTENT THAT A PARTY MAY NOT AS A MATTER OF APPLICABLE LAW DISCLAIM ANY IMPLIED OR OTHER REPRESENTATION, WARRANTY OR CONDITION, THE SCOPE AND DURATION OF SUCH REPRESENTATION, WARRANTY OR CONDITION WILL BE THE MINIMUM REQUIRED UNDER SUCH APPLICABLE LAW.

10. TERM AND TERMINATION

- 10.1 This Agreement will be effective as of the Closing Date and will continue in effect until the date that is *** (***) days after the Closing Date (except that such *** (***) day period shall be extended with respect to any Transition Services that are required to be provided in order for Buyer to comply in all respects with all applicable Laws and Scheduled Approvals, until all Assigned Approvals have been transferred to Buyer and the Buyer has obtained all other Scheduled Approvals, provided that such extension shall apply only to the extent that the failure of the Buyer to have obtained the Scheduled Approvals within such *** (***) day period (and during the period of any such extension) is outside the reasonable control of the Buyer), unless earlier terminated as provided in this Section 10; provided that, notwithstanding anything to the contrary in this Agreement, under no circumstances without Seller's prior written consent will this Agreement be terminated by Buyer with respect to any Transition Services that, in Seller's sole judgment, are required to be provided in order for Seller (as the holder of any Scheduled Approvals required to operate the Radiopharmacy Business) to comply in all respects with all applicable Laws and Scheduled Approvals, unless and until all Assigned Approvals have been transferred to Buyer and the Buyer has obtained all other Scheduled Approvals. Such effective period will be referred to herein as the "Term."
- 10.2 Subject to the proviso in Section 10.1, this Agreement may be terminated for all or any of the services included in the Transition Services, as follows:
 - (a) By any Party for all Transition Services, if the other Party has breached any material obligation hereunder, and has not cured such breach within *** days after receipt of a notice from the non-breaching Party requesting the correction of such breach; such termination will be effective upon failure of the breaching Party to cure such breach within the specified time period;
 - (b) By any Party for all Transition Services, upon the filing or institution of any bankruptcy, reorganization, liquidation or receivership proceedings by the other Party, or upon the failure by the other Party for more than *** (***) days to discharge any such actions against it;
 - (c) By Buyer for all or some of the Transition Services, as applicable, if a Force Majeure Event prevents Seller from rendering such Transition Services for more than *** (***) consecutive days;

- 10 -

- (d) By Buyer, with respect to some or all of the Transition Services, by providing Seller with a *** (***) days' prior written notice;
- (e) Upon the mutual agreement of the Parties set out in writing with respect to some or all of the Transition Services.
- 10.3 The termination of this Agreement for all or some Transition Services will not affect the rights and obligations of the Parties arising prior to such termination or, as applicable, arising after a partial termination of this Agreement with respect to any Transition Services performed after such partial termination. Notwithstanding any provision of the Purchase Agreement to the contrary, termination of this Agreement will not affect the rights and obligations of the Parties under the Purchase Agreement.
- 10.4 Buyer expressly waives the benefit of Articles 2125 and 2129 of the Civil Code of Québec in the event of the termination of the present Agreement so that any termination be governed by the terms of this Agreement.

11. REGULATORY COMPLIANCE OF BUYER

11.1 **Regulatory Compliance.** After the Closing Date and thereafter during the remainder of the applicable Transition Period, Buyer will fulfill all responsibilities for regulatory compliance that it may have with respect to the Covered Product as the holder of the Assigned Approvals for the Covered Product under applicable Law, subject to and in accordance with Section 6.2 of the Purchase Agreement. Seller will fulfill all responsibilities for regulatory compliance that it may have as provider of the Transition Services and in accordance with Section 6.2 of the Purchase Agreement.

12. CONFIDENTIALITY

12.1 **Confidential Information**. In this Agreement, "**Confidential Information**" means any and all confidential or proprietary information (whether in written, visual, oral or any other form) disclosed by or on behalf of any Party (the "**Disclosing Party**") to the other Party (the "**Receiving Party**"), either directly or indirectly, or otherwise relating to the Disclosing Party, of a business, financial or technical nature, including, without limitation, information or documents relating to the business, affairs, operations, results of operations, clients, liabilities, prospects, financial conditions or assets of the Disclosing Party or relating to technologies, inventions, concepts, ideas, developments, methods, processes, techniques, know-how, trade secrets, creative works, systems, specifications, formulations, applications, models, data, prototypes, documentation, reports, records, manuals, business plans, as well as any other business and financial information and other tangible or intangible valuable confidential information. Confidential Information will also include summaries, reports, analyses, compilations, memoranda, notes, extracts, studies or other writings or documents prepared by, for or on behalf of the Receiving Party to the extent they contain or reflect the Confidential Information of the Disclosing Party.

Notwithstanding the foregoing, for the purposes of this Agreement Confidential Information of the Disclosing Party will not include (i) any information which was publicly known and made generally available in the public domain prior to the time of disclosure by the Disclosing Party, (ii) any information which becomes publicly known and made generally available after disclosure by the Disclosing Party to the Receiving Party through no action or inaction of the Receiving Party, (iii) any information which is already in the possession of the Receiving Party at the time of disclosure by the Disclosing Party as shown by the Receiving Party's written files and records as they existed prior to the time of disclosure (excluding information in the possession of Seller as a result of its ownership of the Purchased Assets), (iv) any information which is independently developed by the Receiving Party without as of or reference to the Disclosing Party's Confidential Information, as proven by the Receiving Party's written files and records (excluding information in the possession of Seller as a result of its ownership of its ownership of the Purchased Assets), or (vi) information to the extent that it relates primarily to the Receiving Party or its Affiliates or their respective businesses.

- 12.2 **Confidentiality and Non-Use.** The Seller agrees to use any Confidential Information of the Buyer solely for the purpose of providing the Transition Services which Seller has agreed to provide under this Agreement (the "**Authorized Purpose**"), and not for any other purpose, including for any third party's benefit. The Receiving Party agrees that it will keep strictly confidential and will not disclose or disseminate any of the Confidential Information provided by the Disclosing Party to anyone without the express written consent of the Disclosing Party, except that the Seller will have the limited right to disclose the Confidential Information to its employees, advisors, subcontractors and service providers who are required to have the information in order to provide the Transition Services to the Buyer (which persons or entities will be bound by confidentiality obligations substantially similar to those set forth herein).
- 12.3 Legally Compelled Disclosure. In the event the Receiving Party is required to disclose the Disclosing Party's Confidential Information pursuant to a valid order by a court or other governmental body or as otherwise required by applicable Laws, prior to any such compelled disclosure, the Receiving Party will (i) forthwith notify the Disclosing Party of the legal process, and allow the Disclosing Party to assert the privileged and confidential nature of the Confidential Information against the third party seeking disclosure, and (ii) cooperate fully with the Disclosing Party in protecting against any such disclosure and/or obtaining a protective order narrowing the scope of such disclosure and/or use of the Confidential Information. In the event that such protection against disclosure is not obtained, the Receiving Party will be entitled to disclose the Confidential Information, but only as and to the extent necessary to legally comply with such compelled disclosure.
- 12.4 **Personal Information**. In the course of performing its duties under this Agreement, each of the Parties may collect or receive from or through the other Party, personal information that can be linked to an identifiable individual ("**Personal Information**"). Each of the Parties will comply with and take all necessary measures to ensure that their activities comply with the applicable provisions of the *Personal Information Protection*

- 12 -

and Electronic Documents Act (Canada) and any similar provincial privacy legislation (as amended or enacted from time to time) including without restriction, obligations relating to the collection, use, retention, disclosure and protection of Personal Information. The Parties hereby undertake to immediately notify the other Parties if there is reason to believe that there has been an unauthorized access to or disclosure of Personal Information or if there is reason to believe that there has been a contravention, or there is likely to be a contravention, of any provision of this Agreement or applicable Laws in connection with the collection, use, disclosure, retention or access to Personal Information, and under no circumstances will either Party discuss such incident with any other person.

13. MISCELLANEOUS

- 13.1 **Governing Law**. This Agreement (and all claims, controversies or causes of action (whether in contract, tort or otherwise) that may be based upon, arise out of or in connection with, or relate to, this Agreement or its negotiation, execution or performance, including any claim, controversy or cause of action that may be based upon, arise out of or in connection with, or relate to, any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) will be governed by, and construed in accordance with, the Laws of the State of New York in the United States, without giving effect to any principles of conflicts of laws that would require or permit the application of a different Law.
- 13.2 Entire Agreement. This Agreement (together with the Purchase Agreement and their respective exhibits and schedules) constitutes the sole and entire agreement of the Parties with respect to the subject matter of this Agreement, and supersedes and prevails over all prior and contemporaneous understandings, agreements, representations and warranties (whether written or oral), with respect to that subject matter. The Parties agree that neither Party has relied on (and neither Party will have any remedies or causes of action (whether in contract, tort or otherwise) for) any statements, communications, disclosures, failures to disclose, representations, warranties or agreements of the other Party (or of any other person acting on the other Party's behalf) that is not expressly set forth in this Agreement, including any representations, warranties or agreements arising from statute or otherwise in law.
- 13.3 Severability. In the event that any provision of this Agreement is deemed to be invalid, illegal or unenforceable in any jurisdiction, then (i) the validity, legality or enforceability of the remaining provisions of this Agreement will not be affected or impaired in any way by that provision and (ii) the Parties will negotiate in good faith to replace that provision with a valid and enforceable one that effects the original intent of the Parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated by this Agreement be consummated as originally contemplated to the greatest extent possible.
- 13.4 **Relationship of the Parties**. In making and performing this Agreement, the Parties are acting, and intend to be treated, as independent entities and nothing contained in this Agreement will be construed or implied to create an agency, partnership, joint venture,

- 13 -

franchise or employer-employee relationship between Seller and/or any of its employees, independent contractors and/or agents, on the one hand, and Buyer and/or any of its employees, independent contractors and/or agents, on the other hand. Neither Party will have any express or implied right or authority to (i) make any representation, warranty or commitment, (ii) assume, create or impose any obligations, contracts, agreements or undertakings, (iii) incur any charges or expenses, in each case, for, on behalf of, or in the name of, the other Party, unless expressly so authorized in writing by the other Party.

- 13.5 Force Majeure. Each of the Parties will be excused from the performance of its obligations under this Agreement (except for any obligations to make payments to the other Party under this Agreement) for so long as, and neither Party will be liable to the other Party or to any third Person in the event that, its performance is prevented or delayed due to a Force Majeure Event. The Party suffering the occurrence of a Force Majeure Event will notify the other Party as soon as reasonably practicable, stating the period for which that Force Majeure Event is expected to continue, and any time for performance under this Agreement, the applicable Transition Period and, if applicable, the Term of this Agreement will be extended by the actual time of delay caused by that Force Majeure Event. The Party affected by a Force Majeure Event shall exert commercially reasonable efforts to eliminate, cure or overcome such event and to resume performance of its obligations.
- Arbitration. Any dispute, controversy or claim arising out of or relating to compliance with, or breach or alleged breach, interpretation or validity of, 13.6 this Agreement or otherwise relating to the Parties (each a "Dispute") will be exclusively resolved by binding arbitration, which arbitration may be commenced by sending a written notice to the other Party demanding arbitration of that Dispute (the "Demand"). In that event, the Dispute will be finally resolved by arbitration in accordance with the United States Arbitration Act and the Commercial Arbitration Rules of the American Arbitration Association. The place of the arbitration will be New York, New York. The arbitration will be conducted in the English language before a panel of three arbitrators. Each Party will name one arbitrator, and the two so named will name the third arbitrator, who will act as chairman. If the two party arbitrators cannot agree on a third arbitrator within *** (***) days after the Demand, the third arbitrator will be selected by the American Arbitration Association. The arbitrators will promptly meet, fix the time, date and place of the hearing and notify the Parties. The arbitration will be conducted within *** (***) days after any Demand. All awards are subject to Section 9.4. The panel of arbitrators will promptly transmit an executed copy of its decision to the Parties. The decision of the arbitrators will be final, binding and conclusive upon the Parties. Judgment on the award rendered by the arbitrators may be entered in any court having jurisdiction thereof. Each Party retains the right to seek from a court any interim or provisional relief that may be necessary to protect the rights or property of that Party pending the establishment of the arbitrators' determination of the merits of the controversy, and any such action will not be deemed incompatible with this Agreement to arbitrate or a waiver of the right to arbitration. The obligations of the Parties under this Section 13.6 are specifically enforceable and will survive any termination of this Agreement. All awards are subject to Section 9.4; provided that the arbitrators may award to the party prevailing in the arbitration its reasonable out-of-pocket costs, including the reasonable fees and expenses

- 14 -

of the arbitrators and legal counsel incurred in the arbitration proceedings, or the arbitrators may award the costs on a distributive basis that apportions costs on an issue-by-issue basis and based on the inverse proportion that any amount actually contested but not awarded to Buyer or Seller bears to the aggregate amount actually contested by Buyer and Seller, respectively.

13.7 Notices. All notices, requests, demands, consents, approvals and other communications to any Party required to be given or delivered under, or by reason of, this Agreement will be in writing, and will be deemed to have been given when actually received (or refused) by the addressee after being sent by personal delivery, certified or registered mail, reputable overnight express courier or facsimile (with hard copy to follow) to the address for the addressee set forth below:

Notices to Seller:

Lantheus MI Canada, Inc. 1111 Dr. Frederik-Phillips Boulevard Suite 100 Montreal, Quebec, H4M 2X6 Canada Attn: Vice President, International

with a copy to:

Lantheus Medical Imaging, Inc. 331 Treble Cove Road North Billerica, Massachusetts 01862 United States of America Attn: Vice President, General Counsel

Notices to Buyer:

Isologic Innovative Radiopharmaceuticals Ltd. 1855 32e Avenue, Montreal, Quebec, Canada H8T 3J1 Canada Attn: Andre Gagnon

with a copy to:

DW Healthcare Partners 66 Wellington St. W., Suite 4030 Toronto, Ontario, Canada M5K 1J5 Attn: Sameer Mathur

- 15 -

Either Party may change its notice address by giving a notice to the other Party pursuant to this Section 13.7.

- 13.8 **Successors and Assigns; Assignment**. Neither this Agreement, nor any right, interest or obligation under this Agreement, may be assigned or otherwise transferred by either Party (whether by contract, operation of law or otherwise), in whole or in part, without the prior written consent of the other Party, which consent may be withheld in the sole discretion of the other Party; provided, however, that:
 - (a) either Party may assign or otherwise transfer any or all of its rights, or delegate any or all of its respective duties or obligations, under this Agreement without the consent of the other Party, to an Affiliate of such Party (but only for as long as such Person remains such Party's Affiliate, and provided that such Person has the capabilities and legal authorization necessary to perform the assigned duties and obligations), it being agreed that no such assignment to a Party's Affiliate will release the assigning Party from its obligations under this Agreement;
 - (b) each Party will assign or otherwise transfer all of its applicable rights, and delegate all of its applicable duties and obligations, under this Agreement without the consent of the other Party, to (i) an acquirer of, or successor to, (1) all or substantially all of the assets of that Party (or any of its parent companies) or (2) all or substantially all of any business line to which this Agreement relates or (ii) the surviving entity in any merger, consolidation, equity exchange or reorganization to which (1) that Party (or any of its parent companies) or (2) that business line is a party; and
 - (c) either Party may assign or otherwise transfer any or all of its rights, or delegate any or all of its duties or obligations, under this Agreement without the prior written consent of the other Party for the benefit of any lenders under any financing arrangement of the first-mentioned Party or its Affiliates;

provided, in each case, the assigning Party provides to the other Party written notice of such assignment or transfer as soon as reasonably possible and the assignee, acquirer, successor or surviving entity, as the case may be, agrees in writing to be bound (or by operation of law is bound) by all of the obligations of that Party under this Agreement. Any assignment or transfer in violation of this Agreement will be null and void and have no force or effect. This Agreement will be binding upon and inure to the benefit of the Parties, and its respective successors and assigns as permitted under this Agreement.

13.9 **Third Party Beneficiaries**. None of the provisions of this Agreement will be for the benefit of or enforceable by any third party (other than the Seller Related Parties and the Buyer Related Parties with respect to Section 10) including any creditor of any Party. No third party will obtain any right under any provision of this Agreement or will by reason of any such provision make any claim in respect of any debt, liability or obligation (or otherwise) against any Party.

- 16 -

- 13.10 **Modifications.** Except as otherwise expressly provided, this Agreement may only be amended, modified or supplemented by an agreement in a writing signed by each of the Parties. The Parties agree that, in the event that there is a change in Law that affects (or may affect) the legality or enforceability of this Agreement or any of its provisions or that materially and adversely affects the ability of any Party to perform its obligations or receive the benefits intended under this Agreement, then, as soon as reasonably practical following written notice thereof, the Parties will negotiate reasonably and in good faith (provided that neither Party will have any obligation to enter into) an amendment or substitute agreement in order to best reflect the original intent of the Parties in a manner consistent with that change in applicable Law.
- 13.11 Waivers. No waiver of any provision of this Agreement will be effective unless it is explicitly set forth in writing and signed by the Party so waiving. No failure or delay of a Party in exercising any right, remedy, power or privilege arising from this Agreement will operate or be construed as a waiver thereof. No single or partial waiver in any one or more instances will be deemed to constitute a further or continuing waiver in other instances or a waiver of any provision not expressly identified by that written waiver, whether of a similar or different character, and whether occurring before or after that waiver.
- 13.12 **Counterparts**. This Agreement may be executed in two or more counterparts, each of which will be deemed to be an original, and those counterparts will together constitute one and the same instrument. A facsimile transmission of an executed counterpart signature page will be deemed an original.
- 13.13 **Purchase Agreement.** The Parties expressly acknowledge and agree that certain provisions of the Purchase Agreement are incorporated by reference herein, or by their terms otherwise apply hereto, and further agree that such provisions will be given full effect in interpreting and enforcing this Agreement. In the event of any inconsistency between this Agreement and the Purchase Agreement, the Purchase Agreement will control.
- 13.14 **Further Assurances.** Each of Buyer and Seller will, at the request of the other Party and at such other Party's expense, promptly execute and deliver to such other Party all such further instruments, assignments, filings and other documents, and do all such things, as such other Party may reasonably request in connection with the carrying out and effectuating the terms of this Agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

-17-

IN WITNESS WHEREOF, the undersigned have executed this Transition Services Agreement as of the Closing Date.

<u>SELLER</u>: LANTHEUS MI CANADA, INC.

By:

Name: Cyrille Villeneuve Title: Vice President, International

<u>BUYER:</u> ISOLOGIC INNOVATIVE RADIOPHARMACEUTICALS LTD.

By:

Name: Sameer Mathur Title: Director

- 18 -

SCHEDULE A

TRANSITION SERVICES

FORM OF SECRETARY'S CERTIFICATE OF

LANTHEUS MI CANADA, INC.

Dated: January 13, 2016

Reference is made to that certain Asset Purchase Agreement, amended and restated as of January 13, 2016, by and among Isologic Innovative Radiopharmaceuticals Ltd. (the "*Buyer*") and Lantheus MI Canada, Inc. (the "*Seller*") (the "*Asset Purchase Agreement*"). Unless otherwise defined herein, capitalized terms used shall have the meanings given to them in the Asset Purchase Agreement. The undersigned duly authorized Secretary of the Seller does hereby certify to the Seller, in his or her capacity as Secretary of the Seller and not in his or her personal capacity whatsoever, as follows:

- A. Attached hereto as Exhibit A is a true, correct and complete copy of the articles of incorporation of the Seller (the "Articles") and any amendments thereto, and such Articles are in full force and effect at the date hereof. No proceeding is pending to amend or surrender or, to my actual knowledge, cancel such Articles.
- B. Attached hereto as Exhibit B is a true and correct copy of the bylaws of the Seller, together with any amendments or modifications thereto (the "**Bylaws**"), and such Bylaws are in full force and effect on and as of the date hereof and have not been amended, repealed, modified or restated except as otherwise reflected therein.
- C. Attached hereto as <u>Exhibit C</u> are true and correct copies of the resolutions duly adopted by the board of directors of the Seller authorizing the execution, delivery and performance of the Asset Purchase Agreement and the Transaction Documents by the Seller and the consummation of the transactions contemplated thereby, which resolutions are in full force and effect as of the date hereof without modification in any respect and are the only resolutions dealing with the subject matter thereof.
- D. The Seller is not bound by nor is it a party to any unanimous shareholder agreement or declaration which restricts, in whole in part, the powers of the directors of the Seller to manage or supervise the business and affairs of the Seller.

[Signatures provided on following page.]

IN WITNESS WHEREOF, the undersigned has executed this Secretary's Certificate as of the date first written above.

LANTHEUS MI CANADA, INC.

By: Name:

Title:

[Signature Page - Secretary's Certificate (Buyer)]

<u>Exhibit A</u>

Articles

<u>Exhibit B</u>

Bylaws

<u>Exhibit C</u>

Resolutions

FORM OF SECRETARY'S CERTIFICATE OF

ISOLOGIC INNOVATIVE RADIOPHARMACEUTICALS LTD.

Dated: January 13, 2016

Reference is made to that certain Asset Purchase Agreement, amended and restated as of January 13, 2016, by and among Isologic Innovative Radiopharmaceuticals Ltd. (the "*Buyer*") and Lantheus MI Canada, Inc. (the "*Seller*") (the "*Asset Purchase Agreement*"). Unless otherwise defined herein, capitalized terms used shall have the meanings given to them in the Asset Purchase Agreement. The undersigned duly authorized Secretary of the Buyer does hereby certify to the Seller, in his capacity as Secretary of the Buyer and not in his personal capacity whatsoever, as follows:

- A. Attached hereto as Exhibit A is a true, correct and complete copy of the articles of incorporation of the Buyer (the "Articles") and any amendments thereto, and such Articles are in full force and effect at the date hereof. No proceeding is pending to amend or surrender or, to my actual knowledge, cancel such Articles.
- B. Attached hereto as <u>Exhibit B</u> is a true and correct copy of the bylaws of the Buyer, together with any amendments or modifications thereto (the "**Bylaws**"), and such Bylaws are in full force and effect on and as of the date hereof and have not been amended, repealed, modified or restated except as otherwise reflected therein.
- C. Attached hereto as <u>Exhibit C</u> are true and correct copies of the resolutions duly adopted by the board of directors of the Buyer authorizing the execution, delivery and performance of the Asset Purchase Agreement and the Transaction Documents by the Buyer and the consummation of the transactions contemplated thereby, which resolutions are in full force and effect as of the date hereof without modification in any respect and are the only resolutions dealing with the subject matter thereof.
- D. The Buyer is not bound by nor is it a party to any unanimous shareholder agreement or declaration which restricts, in whole in part, the powers of the directors of the Buyer to manage or supervise the business and affairs of the Buyer.

[Signatures provided on following page.]

IN WITNESS WHEREOF, the undersigned has executed this Secretary's Certificate as of the date first written above.

ISOLOGIC INNOVATIVE RADIOPHARMACEUTICALS LTD.

By:

Name: Sameer Mathur

Title: Secretary

[Signature Page - Secretary's Certificate (Buyer)]

<u>Exhibit A</u>

Articles

<u>Exhibit B</u>

Bylaws

<u>Exhibit C</u>

Resolutions

EMPLOYMENT AGREEMENT

EMPLOYMENT AGREEMENT (the "**Agreement**") dated as of August corporation (the "**Company**") and **John Crowley** ("**Executive**").

, 2013 by and between Lantheus Medical Imaging, Inc., a Delaware

The Company desires to employ Executive and to enter into an agreement embodying the terms of such employment;

Executive desires to accept such employment and enter into such an Agreement.

In consideration of the premises and mutual covenants herein and for other good and valuable consideration, the parties agree as follows:

1. <u>At-Will Employment.</u> Executive's employment with the Company commenced as of September 27, 2010. This agreement was subsequently put in place as of August 12, 2013 (the "**Effective Date**"). Such employment shall be "at-will" employment. Subject to the terms of this Agreement, the Company may terminate Executive's employment and this Agreement for any reason at any time, with or without prior notice and with or without Cause (as defined herein), but subject to certain terms set forth in Section 8 below. Similarly, subject to the terms of this Agreement, Executive may terminate his employment at any time, subject to Section 8 below.

2. Position.

a. Commencing as of the Effective Date, Executive shall serve as the Company's Vice President of Finance and shall report to the Chief Financial Officer of the Company (the "**CFO**"), or the Chief Executive Officer (CEO), or such CEO's designee. Executive shall have such duties and responsibilities as are consistent with such title and position and/or such other duties and responsibilities as may be assigned from time to time by the CFO, CEO or the Board of Directors of Lantheus MI Holdings, Inc. (the "**Board**"). If requested, Executive shall serve as an officer or a member of the Board of Directors of any of the Company's subsidiaries or affiliates without additional compensation.

b. Executive will devote Executive's full business time and best efforts to the performance of Executive's duties hereunder and will not engage in any other business, profession or occupation for compensation or otherwise which would conflict or interfere with the rendition of such services either directly or indirectly, without the prior written consent of the Board; <u>provided</u> that nothing herein shall preclude Executive, subject to the prior approval of the Board, from accepting appointment to or continuing to serve on any board of directors or trustees of any business corporation or any charitable organization; <u>provided</u> in each case, and in the aggregate, that such activities do not conflict or interfere with the performance of Executive's duties hereunder or conflict with Section 9.

3. <u>Base Salary</u>. During Executive's employment hereunder, the Company shall pay Executive a base salary at the annualized rate of \$245,000, payable in regular installments in accordance with the Company's payment practices from time to time. Executive shall be entitled to annual performance and salary review, and any increase in base salary shall be in the sole discretion of the Compensation Committee of the Board. Executive's annual base salary, as in effect from time to time, is hereinafter referred to as the "**Base Salary**".

4. <u>Annual Bonus</u>. With respect to each full fiscal year ending during Executive's employment hereunder, Executive shall be eligible to earn an annual bonus award of thirty percent (30%) of Executive's Base Salary (the "**Target**") based upon achievement of annual EBITDA and/or other performance targets established by the Compensation Committee of the Board within the first three months of each fiscal year (the "**Annual Bonus**"). The Annual Bonus, if any, shall be paid to Executive at the same time as an annual bonus is paid to other similarly situated executives; <u>provided</u>, that Executive is an active employee in good standing with the Company on such date of payment.

5. Equity. Executive shall be eligible to receive future equity awards from time to time pursuant to the Lantheus MI Holdings, Inc. 2013 Equity Incentive Plan, commensurate with Executive's level of responsibilities and the level of awards for similarly situated executives, as determined by the Compensation Committee of the Board in its sole discretion. The terms and conditions of any such equity awards shall be set forth in a separate award agreement.

6. <u>Employee Benefits</u>. During Executive's employment hereunder, Executive shall be entitled to participate in the Company's health, life and disability insurance, and retirement and fringe employee benefit plans as in effect from time to time (collectively "**Employee Benefits**"), on the same basis as those benefits are generally made available to other similarly situated executives of the Company.

7. <u>Business Expenses</u>. During Executive's employment hereunder, reasonable business expenses incurred by Executive in the performance of Executive's duties hereunder shall be reimbursed by the Company in accordance with Company policies.

8. Termination of Employment.

(a) Termination By the Company Without Cause. If Executive's employment is terminated by the Company without Cause, executive shall receive the following, subject to Section 8(g):

(i) an amount equal to one half of the Executive's Base Salary on the date of termination, less taxes and withholdings, payable in substantially equal installments over a period of 6 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;

(ii) a pro rata portion of the Target Annual Bonus amount that Executive would have been eligible to receive pursuant to Section 4 hereof in such year of termination, based upon the percentage of the fiscal year that shall have elapsed through the date of Executive's termination of employment, less taxes and withholdings, payable in substantially equal installments over a period of 6 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would be otherwise due prior thereto;

(iii) provided that Executive elects to purchase continued healthcare coverage under COBRA, an amount equal to the Company's portion of the premium for medical and dental benefits under the Company's group medical and dental plans that the Company was paying on Executive's behalf on the date of termination (which subsidy will be treated as imputed income) for a period of 6 months, with the first payment commencing on the Company's first payroll date after the 60th day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;

(iv) a lump sum amount equal to any earned, but unpaid, Annual Cash Bonus, if any, for the year prior to the year of termination, less taxes and withholdings, which shall be payable on the 60th day following Executive's termination of employment;

(v) a lump sum amount equal to any earned, but unpaid, Base Salary, if any, through the date of Executive's termination of employment, less taxes and withholdings, which shall be payable with the Company's first payroll after Executive's termination of employment; and

(vi) a lump sum amount equal to any unreimbursed business expenses, if any, pursuant to and in accordance with Section 7, incurred through the date of Executive's termination of employment.

(b) Termination Without Cause or For Good Reason following a Change of Control. If, within 12 months following the occurrence of a Change of Control (as defined in the Shareholders Agreement) of Holdings, Executive terminates his employment for Good Reason or the Company terminates Executive's employment with the Company without Cause, Executive shall receive the following, subject to Section 8(g):

(i) an amount equal to one half of the Executive's Base Salary on the date of termination, less taxes and withholdings, payable in substantially equal installments over a period of 6 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;

(ii) an amount equal to the full Target Bonus for the year of termination, less taxes and withholdings, payable in substantially equal installments over a period of 6 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;

(iii) provided that Executive elects to purchase continued healthcare coverage under COBRA, an amount equal to the Company's portion of the premium for medical and dental benefits under the Company's group medical and dental plans that the Company was paying on Executive's behalf on the date of termination (which subsidy will be treated as imputed income) for a period of 6 months, with the first payment commencing on the Company's first payroll date after the 60th day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;

(iv) a lump sum amount equal to any earned, but unpaid, Annual Cash Bonus, if any, for the year prior to the year of termination, less taxes and withholdings, which shall be payable on the 60th day following Executive's termination of employment;

(v) a lump sum amount equal to any earned, but unpaid, Base Salary, if any, through the date of Executive's termination of employment, less taxes and withholdings, which shall be payable on the first payroll date after Executive's termination of employment; and

(vi) a lump sum amount equal to any unreimbursed business expenses, if any, pursuant to and in accordance with Section 7, incurred through the date of Executive's termination of employment. Executive acknowledges and agrees that, in connection with any Change of Control transaction, except as otherwise provided in a separate agreement, Executive shall not be entitled to receive, and shall not be paid, any transaction, success, sale or similar bonus or payment.

(c) Termination Due to Death or Permanent Disability. Executive's employment with the Company shall terminate automatically on Executive's death. In the event of Executive's Permanent Disability, the Company shall be entitled to terminate his employment.

For purposes of this Agreement, the "**Permanent Disability**" of Executive shall mean Executive's inability, because of mental or physical illness or incapacity, whether total or partial, to perform one or more of the material functions of Executive's position with or without reasonable accommodation, for a period of: (i) 90 consecutive calendar days or (ii) an aggregate of 120 days out of any consecutive 12 month period, and which entitles Executive to receive benefits under a disability plan provided by the Company.

In the event of a termination of employment under this section, Executive shall be entitled to following, subject to Section 8(g):

- (i) a lump sum amount equal to any earned, but unpaid, Annual Cash Bonus, if any, for the year prior to the year of termination, less taxes and withholdings, payable on the sixtieth (60th) day following Executive's termination of employment;
- (ii) a lump sum amount equal to any earned, but unpaid, Base Salary, if any, through the date of Executive's termination of employment, less taxes and withholdings, which shall be payable on the first payroll date after Executive's termination of employment;
- (iii) a lump sum amount equal to any unreimbursed business expenses, if any, pursuant to and in accordance with Section 7, incurred through the date of Executive's termination of employment; and
- (iv) a pro rata portion of any Annual Cash Bonus, to the extent earned based on actual performance by the Company, that Executive would have been eligible to receive hereunder in the year of termination, based on the percentage of the fiscal year that shall have elapsed through the date of Executive's termination of employment, payable at such time as any such Annual Cash Bonuses are paid to active senior executives of the Company.

(d) Other Terminations. Executive shall not be entitled to the post-termination benefits set forth in Section 8(a), Section 8(b) or Section 8(c) above if his employment with the Company ceases for any reason other than his termination by the Company without Cause, his resignation for Good Reason or his termination as a result of his death or Permanent Disability; it being understood that if Executive's employment with the Company ceases or terminates for any other reason, he will not be entitled to any severance or post-termination benefits or payments, whether

hereunder or pursuant to any policy of the Company, other than a lump sum amount equal to any earned, but unpaid, Base Salary, if any, through the date of Executive's termination of employment, less taxes and withholdings (payable on the first payroll date after Executive's termination of employment), and a lump sum amount equal to any unreimbursed business expenses, if any, pursuant to and in accordance with Section 3(e), incurred through the date of Executive's termination of employment; provided, that this paragraph shall not alter Executive's rights or obligations he may have or be subject to in connection with respect to his equity interests in Holdings, and Executive's indemnification rights shall continue to be governed in accordance with any Directors and Officers Liability Insurance Policy that the Company may maintain and/or with the Company's certificate of incorporation or by-bylaws or similar governing document, and otherwise in accordance with Section 7.

(e) Cause Definition. For purposes of this Agreement, "**Cause**" means (i) material failure by Executive to perform Executive's employment duties (other than as a consequence of any illness, accident or disability), (ii) continued, willful failure of Executive to carry out any reasonable lawful direction of the Company, (iii) material failure of Executive to comply with any of the applicable rules of the Company contained in its Employee Handbook or any other Company policy, (iv) fraud, willful malfeasance, gross negligence or recklessness of Executive in the performance of employment duties, (v) willful failure of Executive to comply with any of the material terms of this Agreement, (vi) other serious, willful misconduct of Executive which causes material injury to the Company or its reputation, including, but not limited to, willful or gross misconduct toward any of the Company's other employees, and (vii) conviction of a crime (or a pleading of guilty or nolo contendere), other than one which in the opinion of the Board does not affect Executive's position as an employee of the Company.

(f) Good Reason Definition. For purposes of this Agreement, "Good Reason" shall mean, without the Executive's Consent, (A) the failure of the Company to pay, or cause to be paid, Executive's Base Salary or Bonus, as the case may be, when due, (B) a permanent decrease in the Executive's Base Salary, or a failure by the Company to pay material compensation or provide material benefits due and payable to the Executive under his Employment Agreement, (C) the Company requiring the Executive to be based at any office or location that is more than 50 miles from the Company's current headquarters in Billerica, Massachusetts, or (D) the failure of the Company to cause the transferee or successor to all or substantially all of the assets of the Company to assume by operation of law or contractually the Company's obligations hereunder, and provided further that any of the events described in clauses (A) or (D) of this section shall constitute Good Reason only if the Company fails to cure such event within 30 days after receipt from Executive of written notice of the event which constitutes Good Reason, and provided further, that Good Reason shall cease to exist for an event on the 30th day following the later of its occurrence or Executive's reporting relationships, including but not limited to a change in the number of direct or indirect reports to Executive, shall not constitute a material and adverse reduction in Executive's responsibilities, and (y) commensurate with Executive performing his duties Executive will be expected to work at the Company's headquarters in North Billerica, Massachusetts, or as reasonably requested by the Company's headquarters in North Billerica, Massachusetts, as necessitated by business demands or as reasonably requested by the Company.

(g) Separation Agreement and General Release. The payments and benefits set forth in Sections 8(a), 8(b) and 8(c) above shall be expressly conditioned upon Executive's (or his estate or legal representatives, in the case of Section 4(c)) execution and delivery to the Company of a Separation Agreement and General Release in a form that is acceptable to the Company (the "**Separation Agreement**") and such Separation Agreement becoming irrevocable within sixty (60) days following Executive's termination of employment; provided, that any payments or benefits otherwise due prior to such sixtieth (60th) day shall be paid on such sixtieth (60th) day. For the avoidance of doubt, the payments and benefits set forth in Sections 8(a), 8(b) and 8(c) above shall be forfeited if such Separation Agreement has not been executed, delivered and become irrevocable within such sixty (60) day period. Such Separation Agreement shall contain release language substantially similar to the language set forth in Exhibit A attached hereto.

e. <u>Board/Committee Resignation</u>. Upon termination of Executive's employment for any reason, Executive agrees to resign, as of the date of such termination and to the extent applicable, from the Board (and any committees thereof) and the Board of Directors (and any committees thereof) of any of the Company's subsidiaries or affiliates.

9. Non-Competition.

a. Executive acknowledges and recognizes the highly competitive nature of the businesses of the Company and its affiliates and accordingly agrees as follows:

(1) During Executive's employment with the Company and, for a period of one year following the date Executive ceases to be employed by the Company (the **"Restricted Period**"), Executive will not, whether on Executive's own behalf of or in conjunction with any person, firm, partnership, joint venture, association, corporation or other business organization, entity or enterprise whatsoever ("**Person**"), directly or indirectly solicit or assist in soliciting in competition with the Company, the business of any client or prospective client:

(i) with whom Executive had personal contact or dealings on behalf of the Company during the one-year period preceding Executive's termination of employment;

(ii) with whom employees reporting to Executive had personal contact or dealings on behalf of the Company during the one year immediately preceding the Executive's termination of employment; or

(iii) for whom Executive had direct or indirect responsibility during the one year immediately preceding Executive's termination of employment.

(2) During the Restricted Period, Executive will not directly or indirectly:

(i) engage in any business that competes with the business or businesses of the Company or any of its affiliates, namely in the testing, development and manufacturing services for the development, manufacture, distribution, marketing or sale of radiopharmaceutical products, contrast imaging agents and/or radioactive generators for the global medical imaging and pharmaceutical industries, and including, without limitation, businesses which the Company or its affiliates have specific plans to conduct in the future and as to which Executive is aware of such planning (a "Competitive Business");

(ii) enter the employ of, or render any services to, any Person (or any division or controlled or controlling affiliate of any Person) who or which engages in a Competitive Business;

(iii) acquire a financial interest in, or otherwise become actively involved with, any Competitive Business, directly or indirectly, as an individual, partner, shareholder, officer, director, principal, agent, trustee or consultant; or

(iv) interfere with, or attempt to interfere with, business relationships (whether formed before, on or after the date of this Agreement) between the Company or any of its affiliates and customers, clients, suppliers, partners, members or investors of the Company or its affiliates.

(3) Notwithstanding anything to the contrary in this Agreement, Executive may, directly or indirectly, own, solely as an investment, securities of any Person engaged in the business of the Company or its affiliates which are publicly traded on a national or regional stock exchange or on the over-the-counter market if Executive (i) is not a controlling person of, or a member of a group which controls, such Person and (ii) does not, directly or indirectly, own 5% or more of any class of securities of such Person.

(4) During the Restricted Period, Executive will not, whether on Executive's own behalf of or in conjunction with any Person, directly or indirectly:

i. solicit or encourage any employee or consultant of the Company or its affiliates to leave the employment of, or cease providing services to, the Company or its affiliates; or

ii. hire any such employee or consultant who was employed by or providing services to the Company or its affiliates as of the date of Executive's termination of employment with the Company or who left the employment of or ceased providing services to the Company or its affiliates coincident with, or within one year prior to or after, the termination of Executive's employment with the Company.

iii. It is expressly understood and agreed that although Executive and the Company consider the restrictions contained in this Section 9 to be reasonable, if a final judicial determination is made by a court of competent jurisdiction that the time or territory or any other restriction contained in this Agreement is an unenforceable restriction against Executive, the provisions of this Agreement shall not be rendered void but shall be deemed

amended to apply as to such maximum time and territory and to such maximum extent as such court may judicially determine or indicate to be enforceable. Alternatively, if any court of competent jurisdiction finds that any restriction contained in this Agreement is unenforceable, and such restriction cannot be amended so as to make it enforceable, such finding shall not affect the enforceability of any of the other restrictions contained herein.

The provisions of this Section 9 shall survive the termination of this Agreement and Executive's employment for any reason.

10. <u>Non-Disparagement.</u> The Executive shall not at any time (whether during or after Executive's employment with the Company) make, or cause to be made, any statement or communicate any information (whether oral or written) that disparages or reflects negatively on the Company or any of its affiliates, except for truthful statements that may be made pursuant to legal process, including without limitation in litigation, arbitration or similar dispute resolution proceedings. This Section 10 shall survive the termination of this Agreement and Executive's employment for any reason.

- 11. Confidentiality; Intellectual Property.
 - a. Confidentiality.

(i) Executive will not at any time (whether during or after Executive's employment with the Company) (x) retain or use for the benefit, purposes or account of Executive or any other Person; or (y) disclose, divulge, reveal, communicate, share, transfer or provide access to any Person outside the Company (other than its professional advisers who are bound by confidentiality obligations), any non-public, proprietary or confidential information - including, without limitation, trade secrets, know-how, research and development, software, databases, inventions, processes, formulae, technology, designs and other intellectual property, information concerning finances, investments, profits, pricing, costs, products, services, vendors, customers, clients, partners, investors, personnel, compensation, recruiting, training, advertising, sales, marketing, promotions, government and regulatory activities and approvals - concerning the past, current or future business, activities and operations of the Company, its subsidiaries or affiliates and/or any third party that has disclosed or provided any of same to the Company on a confidential basis ("Confidential Information") without the prior written authorization of the Board.

(ii) Confidential Information shall not include any information that is (A) generally known to the industry or the public other than as a result of Executive's breach of this covenant or any breach of other confidentiality obligations by third parties; (B) made legitimately available to Executive by a third party without breach of any confidentiality obligation; or (C) required by law to be disclosed; provided that Executive shall give prompt written notice to the Company of such requirement, disclose no more information than is so required, and cooperate with any attempts by the Company to obtain a protective order or similar treatment.

(iii) Except as required by law, Executive will not disclose to anyone, other than Executive's immediate family and legal or financial advisors, the existence or contents of this Agreement; provided that Executive may disclose to any prospective future employer the provisions of Sections 9, 10 and 11 of this Agreement provided they agree to maintain the confidentiality of such terms.

(iv) Upon termination of Executive's employment with the Company for any reason, Executive shall (x) cease and not thereafter commence use of any Confidential Information or intellectual property (including without limitation, any patent, invention, copyright, trade secret, trademark, trade name, logo, domain name or other source indicator) owned or used by the Company, its subsidiaries or affiliates; (y) immediately return to the Company all Company property and destroy, delete, or return to the Company, at the Company's option, all originals and copies in any form or medium (including memoranda, books, papers, plans, computer files, letters and other data) in Executive's possession or control (including any of the foregoing stored or located in Executive's office, home, laptop or other computer, whether or not Company property) that contain Confidential Information or otherwise relate to the business of the Company, its affiliates and subsidiaries, except that Executive may retain only those portions of any personal notes, notebooks and diaries that do not contain any Confidential Information; and (z) notify and fully cooperate with the Company regarding the delivery or destruction of any other Confidential Information of which Executive is or becomes aware and promptly return any other Company property in Executive's possession.

b. Intellectual Property.

(i) If Executive has created, invented, designed, developed, contributed to or improved any works of authorship, inventions, intellectual property, materials, documents or other work product (including without limitation, research, reports, software, databases, systems, applications, presentations, textual works, content, or audiovisual materials) ("**Works**"), either alone or with third parties, prior to Executive's employment by the Company, that are relevant to or implicated by such employment ("**Prior Works**"), Executive hereby grants the Company a perpetual, nonexclusive, royalty-free, worldwide, assignable, sublicensable license under all rights and intellectual property rights (including rights under patent, industrial property, copyright, trademark, trade secret, unfair competition and related laws) therein for all purposes in connection with the Company's current and future business. A list of all such material Works as of the date hereof is attached hereto as <u>Exhibit B</u>.

(ii) If Executive creates, invents, designs, develops, contributes to or improves any Works, either alone or with third parties, at any time during Executive's employment by the Company and within the scope of such employment and/or with the use of any Company resources ("**Company Works**"), Executive shall promptly and fully disclose such works to the Company and hereby irrevocably assigns, transfers and conveys, to the maximum extent permitted by applicable law, all rights and intellectual property rights therein (including rights under patent, industrial property, copyright, trademark, trade secret, unfair competition and related laws) to the Company to the extent ownership of any such rights does not vest originally in the Company.

(iii) Executive agrees to keep and maintain adequate and current written records (in the form of notes, sketches, drawings, and any other form or media requested by the Company) of all Company Works. The records will be available to and remain the sole property and intellectual property of the Company at all times.

(iv) Executive shall take all requested actions and execute all requested documents (including any licenses or assignments required by a government contract) at the Company's expense (but without further remuneration) to assist the Company in validating, maintaining, protecting, enforcing, perfecting, recording, patenting or registering any of the Company's rights in the Prior Works and Company Works. If the Company is unable for any other reason to secure Executive's signature on any document for this purpose, then Executive hereby irrevocably designates and appoints the Company and its duly authorized officers and agents as Executive's agent and attorney-in-fact, to act for and on Executive's behalf to execute any documents and to do all other lawfully permitted acts in connection with the foregoing.

(v) Executive shall not improperly use for the benefit of, bring to any premises of, divulge, disclose, communicate, reveal, transfer or provide access to, or share with the Company any confidential, proprietary or non-public information or intellectual property relating to a former employer or other third party without the prior written permission of such third party. Executive hereby indemnifies, holds harmless and agrees to defend the Company and its officers, directors, partners, employees, agents and representatives from any breach of the foregoing covenant. Executive shall comply with all relevant policies and guidelines of the Company, including regarding the protection of confidential information and intellectual property and potential conflicts of interest. Executive acknowledges that the Company may amend any such policies and guidelines from time to time, and that Executive remains at all times bound by their most current version.

c. The provisions of this Section 11 shall survive the termination of this Agreement and Executive's employment for any reason.

12. Specific Performance. Executive acknowledges and agrees that the Company's remedies at law for a breach or threatened breach of any of the provisions of Section 9, Section 10 or Section 11 would be inadequate and the Company would suffer irreparable damages as a result of such breach or threatened breach. In recognition of this fact, Executive agrees that, in the event of such a breach or threatened breach, in addition to any remedies at law, the Company, without posting any bond, shall be entitled to cease making any payments or providing any benefit otherwise required by this Agreement and obtain equitable relief in the form of specific performance, temporary restraining order, temporary or permanent injunction or any other equitable remedy which may then be available.

13. Miscellaneous.

a. <u>Governing Law.</u> This Agreement shall be governed by, construed and interpreted in all respects, in accordance with the laws of the State of New York, without regard to conflicts of laws principles thereof.

b. Entire Agreement/Amendments. This Agreement contains the entire understanding of the parties with respect to the employment of Executive by the Company and supersedes all prior agreements, understandings, discussions, negotiations and undertakings, whether written or oral between the Executive and the Company or any of its affiliates with respect to the Executive's employment. There are no restrictions, agreements, promises, warranties, covenants or undertakings between the parties with respect to the subject matter herein other than those expressly set forth herein. This Agreement may not be altered, modified, or amended except by written instrument signed by the parties hereto.

c. <u>No Waiver</u>. The failure of a party to insist upon strict adherence to any term of this Agreement on any occasion shall not be considered a waiver of such party's rights or deprive such party of the right thereafter to insist upon strict adherence to that term or any other term of this Agreement.

d. <u>Severability</u>. In the event that any one or more of the provisions of this Agreement shall be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions of this Agreement shall not be affected thereby.

e. <u>Assignment.</u> This Agreement, and all of Executive's rights and duties hereunder, shall not be assignable or delegable by Executive. Any purported assignment or delegation by Executive in violation of the foregoing shall be null and void ab initio and of no force and effect. This Agreement may be assigned by the Company to a person or entity which is an affiliate or a successor in interest to substantially all of the business operations of the Company. Upon such assignment, the rights and obligations of the Company hereunder shall become the rights and obligations of such affiliate or successor person or entity.

f. <u>Set Off.</u> The Company's obligation to pay Executive the amounts provided and to make the arrangements provided hereunder shall be subject to set-off, counterclaim or recoupment of amounts owed by Executive to the Company or its affiliates.

g. <u>Dispute Resolution</u>. Except with respect to Sections 9, 10, 11 and 12 hereof, any controversy or claim arising out of or related to any provision of this Agreement that cannot be mutually resolved by the parties hereto shall be settled by final, binding and nonappealable arbitration in New York, NY by a single mutually-acceptable arbitrator. Subject to the following provisions, the arbitration shall be conducted in accordance with

the applicable rules of American Arbitration Association then in effect. Any award entered by the arbitrator shall be final, binding and nonappealable and judgment may be entered thereon by either party in accordance with applicable law in any court of competent jurisdiction. This arbitration provision shall be specifically enforceable. The arbitrator shall have no authority to modify any provision of this Agreement or to award a remedy for a dispute involving this Agreement other than a benefit specifically provided under or by virtue of the Agreement. Each party shall be responsible for its own expenses relating to the conduct of the arbitration or litigation (including attorney's fees and expenses) and shall share the fees of the American Arbitration Association and the arbitrator equally.

h. <u>Compliance with Section 409A of the Code</u>. The parties acknowledge and agree that the interpretation of Section 409 A of the Code and its application to the terms of this Agreement is uncertain and may be subject to change as additional guidance and interpretations become available. Anything to the contrary herein notwithstanding, all benefits or payments provided by the Company to the Executive that would be deemed to constitute "nonqualified deferred compensation" within the meaning of Section 409A of the Code are intended to comply with Section 409A of the Code. If, however, any such benefit or payment is deemed to not comply with Section 409A of the Code, the Company and the Executive agree to renegotiate in good faith any such benefit or payment (including, without limitation, as to the timing of any severance payments payable hereunder), if possible, so that either (i) Section 409A of the Code will not apply or (ii) compliance with Section 13(h); provided that neither the Company nor any of its employees or representatives shall have any liability to Executive with respect to thereto.

i. <u>Successors; Binding Agreement.</u> This Agreement shall inure to the benefit of and be binding upon personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees of the parties hereto.

j. <u>Notice</u>. For the purpose of this Agreement, notices and all other communications provided for in the Agreement shall be in writing and shall be deemed to have been duly given when delivered by hand or overnight courier or three days after it has been mailed by United States registered mail, return receipt requested, postage prepaid, addressed to the respective addresses set forth below in this Agreement, or to such other address as either party may have furnished to the other in writing in accordance herewith, except that notice of change of address shall be effective only upon receipt.

If to the Company:

Lantheus Medical Imaging, Inc. 331 Treble Cove Rd. Bldg. 600-2 N. Billerica, MA 01862 Attention: Michael Duffy, Vice President and General Counsel Email: <u>Michael.Duffy@lantheus.com</u>

If to Executive: To Executive's address on file with the Company

k. <u>Executive Representation</u>. Executive hereby represents to the Company that (i) Executive has been provided with sufficient opportunity to review this Agreement and has been advised by the Company to conduct such review with an attorney of his choice, and (ii) the execution and delivery of this Agreement by Executive and the Company and the performance by Executive's duties hereunder shall not constitute a breach of, or otherwise contravene, the terms of any employment agreement or other agreement or policy to which Executive is a party or otherwise bound.

1. <u>Cooperation</u>. Executive shall provide Executive's reasonable cooperation in connection with any action or proceeding (or any appeal from any action or proceeding) which relates to events occurring during Executive's employment hereunder. This provision shall survive any termination of this Agreement or Executive's employment.

m. <u>Withholding Taxes</u>. The Company may withhold from any amounts payable under this Agreement such Federal, state and local taxes as may be required to be withheld pursuant to any applicable law or regulation.

n. <u>Counterparts</u>. This Agreement may be signed in counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

[Signatures on following page]

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the day and year first above written.

Lantheus Medical Imaging, Inc.

/s/ Jeffrey Bailey

By: Jeffrey Bailey Title: President and Chief Executive Officer /s/ John Crowley John Crowley

EXHIBIT A RELEASE

This RELEASE ("**Release**") dated as of ,20 between Lantheus Medical Imaging, Inc., a Delaware corporation (the "**Company**"), and (the "**Executive**").

WHEREAS, the Company and the Executive previously entered into an employment agreement dated March , 2008 (the "**Employment**"); and

WHEREAS, the Executive's employment with the Company has terminated effective , 20

NOW, THEREFORE, in consideration of the premises and mutual agreements contained herein and in the Employment Agreement, the Company and the Executive agree as follows:

;

1. Executive agrees to and does waive any claims he may have for employment by the Company and agrees not to seek such employment or reemployment by the Company in the future. The Executive, on his own behalf and on behalf of his heirs, estate and beneficiaries, further does hereby release the Company, and in such capacities, any of its subsidiaries or affiliates, and each of their respective past, present and future officers, directors, agents, employees, shareholders, investors, employee benefit plans and their administrators or fiduciaries, insurers of any such entities, and its and their successors and assigns and others related to such entities from any and all claims made, to be made, or which might have been made of whatever nature, whether known or unknown, from the beginning of time, including those that arose as a consequence of his employment with the Company, or arising out of the separation from the Company, the severance of such employment relationship, or any act committed or omitted during or after the existence of such employment relationship, all up through and including the date on which this Release is executed, including, but not limited to, those which were, could have been or could be the subject of an administrative or judicial proceeding filed by the Executive or on his behalf under federal, state or local law, whether by statute, regulation, in contract or tort, and including, but not limited to, every claim for front pay, back pay, wages, bonus, fringe benefit, any form of discrimination, wrongful termination, tort, emotional distress, pain and suffering, breach of contract, fraud, defamation, compensatory or punitive damages, interest, attorney's fees, reinstatement or reemployment, and any rights or claims under the Civil Rights Act of 1866, the Age Discrimination in Employment Act of 1967, as amended, 29 U.S.C. sec. 621, et seq., the Americans with Disabilities Act, the Family and Medical Leave Act, the Civil Rights Act of 1964, Title VII, as amended, the Civil Rights Act of 1991, the Employee Retirement Income Security Act of 1974, as amended, the Equal Pay Act, the Worker Adjustment and Retraining Notification Act, the New York State Human Rights Law, the New York City Human Rights Law, the Massachusetts Civil Rights Act, the Massachusetts Equal Pay and Maternity Benefits Law, the Massachusetts Equal Rights for Elderly and Disabled Law, the Massachusetts Small Necessities Leave Act, the Massachusetts Age Discrimination Law, or any other federal, state or local law relating to employment, discrimination in employment, termination of employment, wages, benefits or otherwise. The Executive acknowledges and agrees that even though claims and facts in addition to those now known or believed by him to exist may subsequently be discovered, it is his intention to fully settle and release all claims he may have against the Company and the persons and entities described above, whether known, unknown or suspected. Employee does not waive his right to have

a charge filed with the Equal Employment Opportunity Commission ("**EEOC**") or any state civil rights agency or to participate in an investigation conducted by the EEOC or any state civil rights agency; however, Employee expressly waives his right to recover any monetary relief should any administrative agency, including but not limited to the EEOC, pursue any claim on Employee's behalf.

2. The Company and the Executive acknowledge and agree that the release contained in Paragraph 1 does not, and shall not be construed to, release or limit the scope of any existing obligation of the Company and/or any of its subsidiaries or affiliates (i) to indemnify the Executive for his acts as an officer or director of the Company and/or its subsidiaries or affiliates in accordance with their respective charters or bylaws or under an indemnification agreement to which the Executive and the Company or any of its subsidiaries are parties or under any applicable Directors and Officers insurance policies or under any applicable law or (ii) to the Executive and his eligible, participating dependents or beneficiaries under any existing group welfare (excluding severance), equity, or retirement plan of the Company in which the Executive and/or such dependents are participants.

3. The Executive acknowledges that before entering into this Release, he has had the opportunity to consult with any attorney or other advisor of the Executive's choice, and the Executive is hereby advised to consult with an attorney. The Executive further acknowledges that by signing this Release, he does so of his own free will and act, that it is his intention to be legally bound by its terms, and that no promises or representations have been made to the Executive by any person to induce the Executive to enter into this Release other than the express terms set forth herein. The Executive further acknowledges that he has carefully read this Release, knows and understands its contents and its binding legal effect, including the waiver and release of claims set forth in Paragraph 1 above.

4. The Executive acknowledges that he has been provided at least 21 days to review the Release. In the event the Executive elects to sign this Release prior to this 21 day period, he agrees that it is a knowing and voluntary waiver of his right to wait the full 21 days. The Executive further understand that he has 7 days after the signing hereof to revoke this Release by so notifying the Company, Lantheus Medical Imaging, Inc., 331 Treble Cove Rd., Bldg. 600-2, N. Billerica, MA 01862, Attention: Michael Duffy in writing, such notice to be received by the Company within the 7 day period. This Release shall not become effective or enforceable, and no payments or benefits under Sections 8(c)(ii)(B),(C) and (D) of the Employment Agreement, as applicable, shall be made or provided, until this seven (7) day revocation period expires without the Executive having revoked this Release.

IN WITNESS WHEREOF, the parties have executed this Release on the date first above written.

Lantbeus Medical Imaging, Inc.

By:

Name: Title:

Employee Name

<u>EXHIBIT B</u> PRIOR WORKS

[None]

Exhibit 10.2

Execution Version

AMENDMENT TO EMPLOYMENT AGREEMENT

This Amendment to Employment Agreement ("<u>Amendment</u>") is entered into as of June , 2015, by and between **John Crowley**, an individual ("<u>Employee</u>"), and Lantheus Medical Imagining, Inc., a Delaware corporation (the "<u>Company</u>").

WHEREAS, the Company and the Employee are party to that certain Employment Agreement entered onto on August 12, 2013 (the "Employment Agreement");

WHEREAS, the first underwritten public offering and sale of shares of common stock of Lantheus Holdings, Inc. ("Holdings"), the Company's parent, for cash pursuant to an effective registration statement on Form S-1 under the Securities Act of 1933, as amended (the "Initial Public Offering") shall occur in the near future;

WHEREAS, in anticipation of Holdings' Initial Public Offering, the Company and Employee desire to amend the Employment Agreement to reflect the changes set forth herein; provided, that such amendments and this Amendment shall be effective upon the consummation of such Initial Public Offering; and

WHEREAS, capitalized terms that are not defined herein shall have the same meaning as set forth in the Employment Agreement.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties agree as follows:

1. Amendment to Employment Agreement.

(a) The first sentence of Section 3 of the Employment Agreement is amended and restated in its entirety to read as follows:

"During Executive's employment hereunder, the Company shall pay Executive a base salary at the annualized rate of \$280,000, payable in regular installments in accordance with the Company's payment practices from time to time."

(b) Section 4 of the Employment Agreement is amended and restated in its entirety to read as follows:

"4. <u>Annual Bonus</u>. With respect to each full fiscal year ending during Executive's employment hereunder, Executive shall be eligible to earn an annual bonus award of thirty-five percent (35%) of Executive's Base Salary (the "<u>Target</u>"), and a maximum bonus opportunity of sixty-three percent (63%) of Executive's Base Salary for such fiscal year, in each case, based upon achievement of annual EBITDA and/or other performance targets established by the Compensation Committee of the Board within the first three months of each fiscal year (the "<u>Annual Bonus</u>"). Annual Bonuses, if any, are generally paid in March of the year following the year to which such Annual Bonus relates, by the 15th of that month; provided, that Executive is an active employee in good standing with the Company on such date of payment."

Page 1 of 6

(c) Section 8(a)(i) of the Employment Agreement is amended and restated in its entirety to read as follows:

"(i) an amount equal to the sum of (x) a pro rata portion of an amount equal to 35% of Executive's Base Salary on the date of termination, based upon the percentage of the fiscal year that shall have elapsed through the date of Executive's termination of employment, plus (y) Executive's Base Salary on the date of termination, less taxes and withholdings, payable in substantially equal installments over a period of 12 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;"

(d) Section 8(a)(ii) of the Employment Agreement is deleted in its entirety and replaced with the following:

"(ii) [Intentionally Deleted];"

(e) Section 8(b)(i) of the Employment Agreement is amended and restated in its entirety to read as follows:

"(i) an amount equal to the sum of (x) an amount equal to 35% of Executive's Base Salary on the date of termination, plus (y) Executive's Base Salary on the date of termination, less taxes and withholdings, payable in substantially equal installments over a period of 12 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;"

(f) Section 8(b)(ii) of the Employment Agreement is deleted in its entirety and replaced with the following:

"(ii) [Intentionally Deleted];"

(g) Section 13(h) of the Employment Agreement is amended and restated in its entirety to read as follows:

"(h) Section 409A.

(i) The intent of the parties is that the payments and benefits under this Agreement comply with or be exempt from Section 409A of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>") and, accordingly, to the maximum extent permitted, this Agreement shall be interpreted to be in compliance therewith. To the extent that any provision hereof is modified in order to comply with or be exempt from Section 409A, such modification shall be made in good faith and shall, to the maximum extent reasonably possible, maintain the original intent and economic benefit to Executive and the Company of the applicable provision without violating the provisions of Code Section 409A ("<u>Section 409A</u>").

Page 2 of 6

(ii) If any payment, compensation or other benefit provided to Executive under this Agreement in connection with Executive's "separation from service" (within the meaning of Section 409A) is determined, in whole or in part, to constitute "nonqualified deferred compensation" (within the meaning of Section 409A) and Executive is a specified employee (as defined in Code Section 409A(a)(2)(B)(i)) at the time of separation from service, no part of such payments shall be paid before the day that is six months plus one day after the date of separation or, if earlier, ten business days following Executive's death (the "New Payment Date"). The aggregate of any payments and benefits that otherwise would have been paid and/or provided to Executive during the period between the date of separation of service and the New Payment Date shall be paid to Executive in a lump sum on such New Payment Date. Thereafter, any payments and/or benefits that remain outstanding as of or following the New Payment Date shall be paid without delay over the time period originally scheduled, in accordance with the terms of this Agreement.

(iii) A termination of employment shall not be deemed to have occurred for purposes of any provision of this Agreement providing for the payment of any amounts or benefits subject to Section 409A upon or following a termination of employment unless such termination is also a "separation from service" (within the meaning of Section 409A), and for purposes of any such provision of this Agreement, references to a "resignation," "termination," "terminate," "termination of employment" or like terms shall mean separation from service (within the meaning of Section 409A).

(iv) All expenses or other reimbursements as provided herein shall be payable in accordance with the Company's policies in effect from time to time, but in any event shall be made on or prior to the last day of the taxable year following the taxable year in which such expenses were incurred by Executive. With regard to any provision herein that provides for reimbursement of costs and expenses or in-kind benefits, except as permitted by Section 409A: (i) the right to reimbursement or in-kind benefits shall not be subject to liquidation or exchange for another benefit; and (ii) the amount of expenses eligible for reimbursements or in-kind benefits provided during any taxable year shall not affect the expenses eligible for reimbursement or in-kind benefits to be provided in any other taxable year.

(v) For purposes of Section 409A, Executive's right to receive any installment payments pursuant to this Agreement shall be treated as a right to receive a series of separate and distinct payments. Whenever a payment under this Agreement specifies a payment period with reference to a number of days (e.g., payment shall be made within 30 days following the date of termination), the actual date of payment within the specified period shall be within the sole discretion of the Company."

(h) A new Section 14 is hereby added to the Employment Agreement as follows:

"(a) Anything in this Agreement to the contrary notwithstanding, in the event that the receipt of all payments or distributions by the Company in the nature of compensation to or for the Executive's benefit, whether paid or payable pursuant to this Agreement or otherwise

Page 3 of 6

(a "<u>Payment</u>"), would subject the Executive to the excise tax under Section 4999 of the Code, the accounting firm which audited the Company prior to the corporate transaction which results in the application of such excise tax (the "<u>Accounting Firm</u>") shall determine whether to reduce any of the Payments to the Reduced Amount (as defined below). The Payments shall be reduced to the Reduced Amount only if the Accounting Firm determines that the Executive would have a greater Net After-Tax Receipt (as defined below) of aggregate Payments if the Executive's Payments were reduced to the Reduced Amount. If such a determination is not made by the Accounting Firm, the Executive shall receive all Payments to which the Executive is entitled.

(b) If the Accounting Firm determines that aggregate Payments should be reduced to the Reduced Amount, the Company shall promptly give the Executive notice to that effect and a copy of the detailed calculation thereof. All determinations made by the Accounting Firm under this Section 14 shall be made as soon as reasonably practicable and in no event later than sixty (60) days following the date of termination or such earlier date as requested by the Company. For purposes of reducing the Payments to the Reduced Amount, such reduction shall be implemented by determining the Parachute Payment Ratio (as defined below) for each Payment and then reducing the Payments in order beginning with the Payment of such Payments, with amounts having later payment dates being reduced first. For Payments with the same Parachute Payments shall be reduced based on the time of payment, such Payments shall be reduced on a pro rata basis (but not below zero) prior to reducing Payments with a lower Parachute Payment Ratio. In all cases, the reduction of Payments shall be implemented in a manner that complies with Section 409A of the Code. All other provisions of any agreement embodying the Payments shall remain in full force and effect. All fees and expenses of the Accounting Firm shall be bome solely by the Company.

(c) As a result of the uncertainty in the application of Section 4999 of the Code at the time of the initial determination by the Accounting Firm hereunder, it is possible that amounts will have been paid or distributed by the Company to or for the benefit of the Executive pursuant to this Agreement or otherwise which should not have been so paid or distributed (the "<u>Overpayment</u>") or that additional amounts which will have not been paid or distributed by the Company to or for the benefit of the Executive pursuant to this Agreement or otherwise could have been so paid or distributed (the "<u>Underpayment</u>"), in each case, consistent with the calculation of the Reduced Amount hereunder. In the event that the Accounting Firm, based upon the assertion of a deficiency by the Internal Revenue Service against either the Company or the Executive which the Accounting Firm believes has a high probability of success, determines that an Overpayment has been made, the Executive shall pay any such Overpayment to the Company together with interest at the applicable federal rate provided for in Section 7872(f)(2) of the Code; *provided, however*, that no amount shall be payable by the Executive to the Company if and to the extent such payment would not either reduce the amount on which the Executive is subject to tax under Section 1 and Section 4999 of the Code or generate a refund of such taxes. In the event that the Accounting Firm, based upon controlling precedent or substantial authority, determines that an Underpayment is determined) by the Company to or for the benefit of the Executive to gether with interest at the applicable for in Section 7872(f)(2) of the Code; for the benefit of the Executive to gether with interest at the applicable federal rate provided for in Section 7872(f)(2) of the Code; provided, however, that no amount shall be payable by the Executive to the Company if and to the extent such payment would not either reduce the amount on which the Executive is subject to tax under Section 1 and Section 14999

Page 4 of 6

(d) For purposes hereof, the following terms have the meanings set forth below: (i) "<u>Reduced Amount</u>" shall mean the greatest amount of Payments that can be paid that would not result in the imposition of the excise tax under Section 4999 of the Code if the Accounting Firm determines to reduce Payments pursuant to this Section 14, (ii) "<u>Net After-Tax Receipt</u>" shall mean the present value (as determined in accordance with Sections 280G(b)(2) (A)(ii) and 280G(d)(4) of the Code) of a Payment net of all taxes imposed on the Executive with respect thereto under Sections 1 and 4999 of the Code and under applicable state and local laws, determined by applying the highest marginal rate under Section 1 of the Code and under state and local laws which applied to the Executive's taxable income for the immediately preceding taxable year, or such other rate(s) as the Executive certifies, in the Executive's sole discretion, as likely to apply to the Executive in the relevant tax year(s), and (iii) "<u>Parachute Payment Ratio</u>" shall mean a fraction the numerator of which is the present value (as determined in accordance with Sections 280G(b)(2)(A)(ii) and 280G(d)(4) of the Code) of the applicable Payment for purposes of Section 280G and the denominator of which is the intrinsic value of such Payment."

2. <u>References</u>. All references in the Employment Agreement to "this Agreement" and any other references of similar import shall hereinafter refer to the Employment Agreement as amended by this Amendment.

3. <u>Remaining Provisions</u>. Except as expressly modified by this Amendment, the Employment Agreement shall remain in full force and effect. This Amendment embodies the entire agreement and understanding of the parties hereto with respect to the subject matter hereof, and supersedes all prior and contemporaneous agreements and understandings, oral or written, relative thereto.

4. Governing Law. This Amendment shall be governed by, construed and interpreted in all respects, in accordance with the laws of the State of New York, without regard to conflicts of laws principles thereof.

5. <u>Amendment Effective Date</u>. This Amendment shall be effective as of immediately prior to the consummation of Holdings' Initial Public Offering, and to the extent such Initial Public Offering does not occur prior to December 31, 2015, this amendment shall be void ab inito.

6. <u>Counterparts</u>. This Amendment may be executed by either of the parties hereto in counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same instrument.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

Page 5 of 6

IN WITNESS WHEREOF, the parties have executed this Amendment as of the date first set forth above.

LANTHEUS MEDICAL IMAGING, INC.

By Michael P. Duffy

Name: Michael P. Duffy Title: Vice President, General Counsel and Secretary

ACCEPTED AND AGREED:

/s/ John Crowley

Name: John Crowley Title: Vice President and Chief Accounting Officer Date: 6/22/2015

Page 6 of 6

AMENDMENT TO EMPLOYMENT AGREEMENT

This Amendment to Employment Agreement ("<u>Amendment</u>") is entered into as of March 25, 2016, by and between **John Crowley**, an individual ("<u>Employee</u>"), and Lantheus Medical Imagining, Inc., a Delaware corporation (the "<u>Company</u>").

WHEREAS, the Company and the Employee are party to that certain Employment Agreement entered onto on August 12, 2013 (the "Employment Agreement") which was subsequently modified effective June 22, 2015;

WHEREAS, the Board of Directors has appointed Mr. Crowley to the role of Chief Financial Officer; and

WHEREAS, capitalized terms that are not defined herein shall have the same meaning as set forth in the Employment Agreement.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties agree as follows:

1. Amendment to Employment Agreement.

(a) Section 2a of the Employment Agreement is amended and restated in its entirety to read as follows:

"Commencing as of the Effective Date, Executive shall serve as the Company's Chief Financial Officer (CFO) and shall report to the Chief Executive Officer (CEO), or such CEO's designee. Executive shall have such duties and responsibilities as are consistent with such title and position and/or such other duties and responsibilities as may be assigned from time to time by the CEO or the Board of Directors of Lantheus MI Holdings, Inc. (the "**Board**"). If requested, Executive shall serve as an officer or a member of the Board of Directors of any of the Company's subsidiaries or affiliates without additional compensation."

(a) The first sentence of Section 3 of the Employment Agreement is amended and restated in its entirety to read as follows:

"During Executive's employment hereunder, the Company shall pay Executive a base salary at the annualized rate of \$360,000, payable in regular installments in accordance with the Company's payment practices from time to time."

(b) Section 4 of the Employment Agreement is amended and restated in its entirety to read as follows:

"4. <u>Annual Bonus</u>. With respect to each full fiscal year ending during Executive's employment hereunder, Executive shall be eligible to earn an annual bonus award of fifty percent (50%) of Executive's Base Salary (the "<u>Target</u>"), and a maximum bonus opportunity of

Page 1 of 3

ninty percent (90%) of Executive's Base Salary for such fiscal year, in each case, based upon achievement of annual EBITDA and/or other performance targets established by the Compensation Committee of the Board within the first three months of each fiscal year (the "<u>Annual Bonus</u>"). Annual Bonuses, if any, are generally paid in March of the year following the year to which such Annual Bonus relates, by the 15th of that month; provided, that Executive is an active employee in good standing with the Company on such date of payment."

(c) Section 8(a)(i) of the Employment Agreement is amended and restated in its entirety to read as follows:

"(i) an amount equal to the sum of (x) a pro rata portion of an amount equal to 50% of Executive's Base Salary on the date of termination, based upon the percentage of the fiscal year that shall have elapsed through the date of Executive's termination of employment, plus (y) Executive's Base Salary on the date of termination, less taxes and withholdings, payable in substantially equal installments over a period of 12 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;"

(e) Section 8(b)(i) of the Employment Agreement is amended and restated in its entirety to read as follows:

"(i) an amount equal to the sum of (x) an amount equal to 50% of Executive's Base Salary on the date of termination, plus (y) Executive's Base Salary on the date of termination, less taxes and withholdings, payable in substantially equal installments over a period of 12 months in accordance with the Company's normal payroll practices, with payments commencing with the Company's first payroll after the sixtieth (60th) day following Executive's termination of employment, and such first payment shall include any such amounts that would otherwise be due prior thereto;"

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

Page 2 of 3

IN WITNESS WHEREOF, the parties have executed this Amendment as of the date first set forth above.

LANTHEUS MEDICAL IMAGING, INC.

By /s/ Michael Duffy

Name: Michael Duffy Title: General Counsel, Secretary & SVP, Strategy & Business Development

ACCEPTED AND AGREED:

/s/ John Crowley

Name: John Crowley Title: Chief Financial Officer Date: 3/25/2016

Page 3 of 3

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECURITIES EXCHANGE ACT RULES 13a-14(a) AND 15d-14(a), AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Mary Anne Heino, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Lantheus Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: May 3, 2016

/s/ MARY ANNE HEINO Name: Mary Anne Heino Title: President and Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO SECURITIES EXCHANGE ACT RULES 13a-14(a) AND 15d-14(a), AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, John Crowley, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Lantheus Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: May 3, 2016

/s/ JOHN CROWLEY Name: John Crowley Title: *Chief Financial Officer*

CERTIFICATION OF THE CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED BY SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

Each of the undersigned hereby certifies, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350 that:

The Quarterly Report on Form 10-Q of Lantheus Holdings, Inc. (the "Company") for the fiscal quarter ended March 31, 2016 (the "Report") filed with the Securities and Exchange Commission on the date hereof fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m); and

The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: May 3, 2016

/s/ MARY ANNE HEINO Name: Mary Anne Heino Title: President and Chief Executive Officer

Dated: May 3, 2016

/s/ JOHN CROWLEY Name: John Crowley Title: *Chief Financial Officer*

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.